

Optimization with set relations

Dissertation

zur Erlangung des akademischen Grades doctor rerum naturalium (Dr. rer. nat.)

vorgelegt der

Mathematisch-Naturwissenschaftlich-Technischen Fakultät der Martin-Luther-Universität Halle-Wittenberg

von Herrn Andreas Löhne geboren am 15. Oktober 1972 in Querfurt

Gutachterin/Gutachter:

 Prof. Dr. Christiane Tammer, MLU Halle–Wittenberg
 Prof. Dr. Johannes Jahn, Universität Erlangen–Nürnberg eingereicht am: 18.01.2005
 Tag der Verteidigung: 27.05.2005

urn:nbn:de:gbv:3-000008490

[http://nbn-resolving.de/urn/resolver.pl?urn=nbn%3Ade%3Agbv%3A3-000008490]

Acknowledgments

First of all I would like to thank Prof. Dr. Christiane Tammer and Dr. Andreas Hamel for being great supervisors.

I'm greatly indebted to Prof. Dr. C. Zălinescu from the university "Al. I. Cuza" Iasi (Romania) for the many helpful suggestions, which influenced this work very positively.

Moreover, I would like to thank all my colleagues (and former colleagues) from the Institute of Optimization and Stochastics for the many inspiring discussions and for the good working atmosphere at the institute.

I gratefully acknowledge the "Land Sachsen–Anhalt" and the Martin–Luther–University Halle– Wittenberg for the financial support by a scolarship over almost three years.

Last but not least I would like to express my deepest gratitude to my parents, to my grand parents, to my son Pascal and, most of all, to my girl–friend Jana Lippke for their great support all the time.

Danksagung

Mein besonderer Dank gilt Frau Prof. Dr. Christiane Tammer und Herrn Dr. Andreas Hamel für die sehr gute Betreuung dieser Arbeit.

Für die vielen sehr hilfreichen Hinweise, die diese Arbeit sehr positiv beeinflusst haben, bin ich Herrn Prof. Dr. C. Zălinescu von der Universität "Al. I. Cuza" Iasi (Rumänien) äußerst dankbar.

Weiterhin gilt mein Dank den Angehörigen (und ehemals Angehörigen) des Instituts für Optimierung und Stochastik für die vielen anregenden Diskussionen und das sehr angenehme Arbeitsklima im Institut.

Dem Land Sachsen–Anhalt und der Martin–Luther–Universität Halle–Wittenberg möchte ich für die Gewährung eines Graduiertenstipendiums über knapp drei Jahre danken.

Nicht zuletzt möchte ich meinen Eltern, meinen Großeltern, meinem Sohn Pascal und vor allem meiner Lebensgefährtin Jana Lippke für ihre Unterstützung ganz herzlich danken.

Contents

In	Introduction					
N	Notation					
1	The	e closed convex sets in \mathbb{R}^p	13			
	1.1	The space $\widehat{\mathcal{F}}$ of closed subsets of \mathbb{R}^p	13			
	1.2	The space $\widehat{\mathcal{C}}$ of closed convex subsets of \mathbb{R}^p	14			
	1.3	Embedding subsets of $\widehat{\mathcal{C}}$ into a linear space $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	18			
2	Convergence of closed convex sets					
	2.1	Painlevé–Kuratowski convergence	23			
	2.2	\mathcal{C} -Convergence	25			
	2.3	Scalar convergence vs. \mathcal{C} -convergence	34			
3	Fun	ctions with values in $\widehat{\mathcal{C}}$	39			
	3.1	Semi–continuity of $\widehat{\mathcal{F}}$ –valued functions	40			
	3.2	Semi-continuity	43			
	3.3	Locally bounded functions	50			
	3.4	Convex functions	53			

Contents

3.5	Conjugates	58
3.6	Biconjugation theorem	60

4 Duality

Duality					
4.1	Fenchel duality	65			
4.2	Lagrange duality	71			
4.3	Some special cases	75			

5	5 Relationship to vector optimization		
	5.1	Basic concepts	77
	5.2	Point relation vs. set relation approach	80
	5.3	On the structure of vector optimization problems $\ldots \ldots \ldots \ldots \ldots \ldots$	83

Appendix

А	Some calculus rules of sets	87
В	Partially ordered sets	91
С	Ordered conlinear spaces	92

Index of notation

Bibliography

97

 $\mathbf{95}$

Introduction

Since the middle of the 80ies, *set-valued optimization problems*, i.e., optimization problems with a set-valued objective map, have been investigated. This field of research is called *set-valued optimization* or *set optimization*. The most results in the literature are based on the following (or a similar) solution concept:

One considers a set-valued objective map $F: X \rightrightarrows Y$ and a set of feasible points $S \subseteq X$, where X and Y are linear spaces and Y is partially ordered by a convex pointed cone. A pair (\bar{x}, \bar{y}) with $\bar{x} \in S$ and $\bar{y} \in F(\bar{x})$ is called a *minimizer* of the set-valued optimization problem if \bar{y} is a minimal element of the set $F(S) := \bigcup_{x \in S} F(x)$, where the minimality notion is understood with respect to the partial ordering in the space Y. Therefore, (in this work) set optimization based on this solution concept is called set optimization with point relations. In this field, there are papers on optimality conditions, e.g. Corley [21], Luc [58], Jahn and Rauh [43], Götz and Jahn [32], Chen and Jahn [16], Crespi, Ginchev and Rocca [22], as well as papers on duality assertions, e.g. Tanino and Sawaragi [79], Corley [19], where only the dual problems are set-valued, and Kawasaki [44], Postolică [66], Corley [20], Luc and Jahn [59], Song [73], where both the primal and the dual problem are set-valued. A good survey as well as further references can be found in the book of Jahn [42]. Set–valued optimization problems based on this solution concept naturally occur in vector optimization, for instance, as dual problems (see e.g. Tanino and Sawaragi [79], Corley [19]). Of course, vector optimization problems provide a very important special case of set optimization with numerous applications. Moreover, it turned out that the answer to certain problems in vector optimization can be found, if the vector optimization problem is considered in a set-valued framework. For instance, by Hamel, Heyde, Löhne, Tammer and Winkler [34] a set–valued approach is used to solve the problem of the duality gap in linear vector optimization in the case that the right hand side of the inequality constraints is zero.

Optimization with set relations (actually "set optimization with set relations") provides quite a different approach to set optimization. The basic idea is to understand the set-valued objective map as a function $f: X \to \mathcal{P}(Y)$ into the space $\mathcal{P}(Y)$ of all subsets of Y. This space is provided with an appropriate ordering relation. Such ordering relations have been investigated, for instance, by Young [83], Nishnianidze [64], Brink [13], Kuroiwa [49] and Hamel [33]. In the special case that Y is a linear space, $K \subseteq Y$ a convex pointed cone and $A, B \subseteq Y$, these relations can be expressed by

$$A \preccurlyeq_K B :\Leftrightarrow B \subseteq A + K$$
 and $A \preccurlyeq_K B :\Leftrightarrow A \subseteq B - K$.

In a sequence of very similar papers of Kuroiwa, e.g. [49], [50], [51], [52], [53], corresponding optimization problems are investigated and some first steps towards a duality theory are taken. These optimization problems are based on the following solution concept: A point $x \in S$ is called a \preccurlyeq_K -minimal solution of the set-valued optimization problem if

$$(f(x) \preccurlyeq_K f(\bar{x}), x \in S) \Rightarrow f(\bar{x}) \preccurlyeq_K f(x).$$

Some other results in this field are generalizations of Ekeland's variational principle (Truong

[80]) as well as Phelps type minimal point theorems (Hamel and Löhne [35]) to the framework of these ordering relations.

This work is concerned with optimization with set relations, however, our investigations are not based on the solution concept introduced above. Instead we use the lattice structure of the image space $\mathcal{P}(Y)$ and ask for the infimum and supremum of the objective function over the set S. Our theory is modeled on scalar optimization rather than vector optimization. Nevertheless we show that this approach is beneficial for a better understanding of the structures in vector optimization and set optimization with point relations, because problems of this kind are hidden in every vector optimization and set optimization problem. Let us proceed with a more detailed discussion.

In this work we investigate optimization problems based on set relations where we restrict the image space of the objective function f to the family of closed convex subsets of \mathbb{R}^p , in the following denoted by $\hat{\mathcal{C}}$. The space $\hat{\mathcal{C}}$ is equipped with an addition and a multiplication by nonnegative reals, more precisely, it is a conlinear space (see Appendix C). The reason for supposing convex sets is that many assertions, even in the classical theory, require convexity of the objective function. The convexity notion in our framework, however, implies that the values are convex sets. The convexity of the sets also ensures that the "second distributive law" is valid in our image space $\hat{\mathcal{C}}$, which is beneficial in some cases. The reason for the closedness is that we only use convergences that do not distinguish between a set and its closure. The space $\hat{\mathcal{C}}$ is ordered by set inclusion. In fact, $(\hat{\mathcal{C}}, \supseteq)$ is an ordered conlinear space (see Appendix C). In contrast to that, $\hat{\mathcal{C}}$ being equipped with the orderings \preccurlyeq_K or \preccurlyeq_K is just a quasi-ordered conlinear space, because \preccurlyeq_K and \preccurlyeq_K are not antisymmetric. Of course, one can switch over to equivalence classes in order to obtain antisymmetry. Then, however, \preccurlyeq_K and \preccurlyeq_K reduce to the usual set inclusion \supseteq and \subseteq , respectively. We next observe that our image space $(\hat{\mathcal{C}}, \supseteq)$ is order complete (a complete lattice). Infimum and supremum of a nonempty subset $\mathcal{A} \subseteq \hat{\mathcal{C}}$ can be expressed by

$$\inf \mathcal{A} = \operatorname{cl}\operatorname{conv} \bigcup_{A \in \mathcal{A}} A \qquad \text{and} \qquad \sup \mathcal{A} = \bigcap_{A \in \mathcal{A}} A.$$

This allows us to proceed analogously to the classical scalar optimization theory. We consider a $\hat{\mathcal{C}}$ -valued objective function and a set $S \subseteq X$ of feasible points. Our goal is to determine the sets $\inf_{x \in S} f(x)$ and $\sup_{x \in S} f(x)$ or, equivalently, the sets $\operatorname{cl} \operatorname{conv} \bigcup_{x \in S} f(x)$ and $\bigcap_{x \in S} f(x)$. It is also interesting to ask for a corresponding solution concept (in the space X), however, this is beyond the scope of this work.

Optimization based on set inclusion subsumes the classical scalar optimization theory. This can be seen by the following simple reformulation. Consider an extended real-valued objective function $f: X \to \mathbb{R} \cup \{-\infty, +\infty\}$. Then the function \bar{f} , defined by $\bar{f}(\cdot) := \{f(\cdot)\} + \mathbb{R}_+$ (where we set $\{+\infty\} + \mathbb{R}_+ = \emptyset$ and $\{-\infty\} + \mathbb{R}_+ = \mathbb{R}$), is a \hat{C} -valued function and it holds

$$\left\{\inf_{x\in S} f(x)\right\} + \mathbb{R}_{+} = \inf_{x\in S} \bar{f}(x) \quad \text{and} \quad \left\{\sup_{x\in S} f(x)\right\} + \mathbb{R}_{+} = \sup_{x\in S} \bar{f}(x).$$

On the other hand, in case we know $\inf_{x \in S} \bar{f}(x) \in \hat{\mathcal{C}}$ and $\sup_{x \in S} \bar{f}(x) \in \hat{\mathcal{C}}$ we obtain the real

Introduction

counterparts by

$$\inf_{x \in S} f(x) = \inf \left\{ \inf_{x \in S} \bar{f}(x) \right\} \quad \text{ and } \quad \sup_{x \in S} f(x) = \inf \left\{ \sup_{x \in S} \bar{f}(x) \right\}.$$

We observe that a scalar optimization problem can be decomposed into two components, namely,

- 1. an optimization problem based on set inclusion, where the involved sets have the special structure of $A_r := \{s \in \mathbb{R} | s \ge r\}, r \in \mathbb{R} \cup \{-\infty, +\infty\}.$
- 2. the (trivial) problem of determining the infimum of such sets.

In this manner it is interesting to compare the construction of real numbers by Dedekind cuts.

Optimization problems based on set inclusion are also hidden in other problems such as vector optimization problems and set optimization problems based on point relations. Similar to the scalar case, set optimization problems can be decomposed into two components, namely,

1. an optimization problem based on set inclusion, in fact, the problem to determine the set

$$F(S) := \bigcup_{x \in S} F(x),$$

2. the problem of determining the set $\inf F(S)$ of infimal or the set $\min F(S)$ of minimal (efficient) points with respect to the partial ordering in Y.

We show in this work that a separate investigation of both parts leads to deeper insights into the structure of set optimization problems based on point relations. Many assertions that are well-known for scalar problems can be generalized to optimization problems based on set inclusion. For instance, we will generalize a biconjugation theorem and duality assertions of Fenchel as well as Lagrange type to this framework. In doing so, we essentially maintain the structures of the scalar case. Putting both components together, certain parts of these structures are lost. This is due to the fact that the second component, which is "nontrivial" in contrast to the scalar case, destroys the lattice structure in the sense that the set of minimal points cannot be described with an infimum (in the sense of lattice theory) in the space Y.

The investigations on optimization problems with set relations leads to some results, which could be of independent interest. Embedding of convex sets into a linear space has been investigated by many authors, beginning with the paper of Rådström [67], but only for spaces of compact or bounded sets. We extend some results to nonbounded sets. Secondly, we obtain some results on convergence in the space of closed convex subsets of \mathbb{R}^p . We introduce a convergence class and investigate the relationsship to well-known convergences.

This work is organized as follows. Chapter 1 is devoted to the study of algebraic and order theoretic properties of the ordered conlinear space $(\hat{\mathcal{C}}, \oplus, \supseteq)$ of closed convex subsets of \mathbb{R}^p

(where \oplus is the closure of the Minkowski addition). This involves an investigation of the dual description of a closed convex set by its support function, which leads to an assertion about the possibility of embedding certain subsets of $\hat{\mathcal{C}}$ into a linear space. Some results in this context are extensively used in the subsequent chapters.

The second chapter deals with topological properties of \hat{C} . We introduce a convergence class in \hat{C} , which seems to be new. We define this convergence by appropriate concepts of upper and lower limits. A characterization by (a certain type of) convergence of support functions is indicated. We compare our convergence with well–established concepts such as Painlevé– Kuratowski convergence and scalar convergence.

In Chapter 3, we investigate functions with values in \hat{C} . Based on the results of the second chapter, we develop semi-continuity concepts. Moreover, we introduce the notion of a conjugate of a \hat{C} -valued function. The main result of this chapter is a biconjugation theorem for \hat{C} -valued functions.

Chapter 4 is devoted to optimization problems with $\hat{\mathcal{C}}$ -valued objective function and with respect to the set relation approach. In particular, we draw our attention to duality theory. We prove weak as well as strong duality assertions based on Fenchel as well as Lagrange approach. The biconjugation theorem, established in Chapter 3, provides a main tool for the proof of the strong duality assertions.

The last chapter contains a comparison of the duality results for optimization with set inclusion and duality assertions in vector optimization (and set optimization with point relations). The closest relationship between both types of problems can be shown for vector optimization problems based on weakly minimal points. In this context, a strong duality result for vector optimization problems is easily obtained from a duality result for a problem based on set inclusion and vise versa. Moreover, we continue to point out that optimization problems based on set relations are hidden in every vector optimization problem.

In the appendix we summarize some well–known facts, which could be useful for reading this work. Part A is a collection of calculus rules for sets. In Part B we recall some fundamental notions with respect to partially ordered sets and Part C gives the definition of an *ordered conlinear space*, which is the underlying structure of the image space of an optimization problem based on set inclusion.

Notation

Notation

Up to a few exceptions, we frequently use the notation of Rockafellar's "Convex Analysis" [1]. The following notations are not in accordance with this book. We write \mathbb{R} (instead of R) for the set of real numbers and \mathbb{N} for the set of positive integers. We set $\mathbb{R}_{+}^{p} :=$ $\{y \in \mathbb{R}^{p} | \forall i \in \{1, ..., p\} : y_{i} \geq 0\}$ and $\mathbb{R}_{+} := \mathbb{R}_{+}^{1}$. The symbol \oplus does not mean the direct sum, instead it will be used for the "closed Minkowski addition", i.e., for $A, B \in \mathbb{R}^{p}$ we set $A \oplus B := \operatorname{cl}(A+B)$. The Euclidian norm in \mathbb{R}^{p} is denoted by $\|\cdot\|$ (instead of $|\cdot|$). Furthermore we write $\operatorname{rg} T := \bigcup_{x \in X} T(x)$ for the range of a function $T : X \to Y$. By rb A we denote the relative boundary of a set $A \subseteq \mathbb{R}^{p}$, i.e., rb $A := \operatorname{cl} A \setminus \operatorname{ri} A$. A cone $C \subseteq \mathbb{R}^{p}$ is said to be *pointed* if its lineality space is zero, i.e., $C \cap -C = \{0\}$. The Euclidian (closed) unit ball is denoted by \mathbb{B} (instead of B). If M is a matrix, M^{T} denotes the transposed matrix. Further notations are defined when they first occur. A short explanation can be found at page 95 f.

Notation

Chapter 1

The closed convex sets in \mathbb{R}^p

In this chapter, we investigate algebraic and order theoretic properties of the space of closed convex subsets of \mathbb{R}^p . In the following, this space is denoted by $\hat{\mathcal{C}}$. Equipped with the partial ordering "set inclusion" this space is an example for an ordered conlinear space, see Appendix C. In particular, $\hat{\mathcal{C}}$ is not a linear space. However, the structure of a conlinear space is rich enough for a meaningful definition of a cone as well as a convex set.

This chapter is organized as follows. In Section 1.1, we briefly summarize some properties of the space of closed subsets of \mathbb{R}^p . Section 1.2 is devoted to corresponding properties of the space of closed convex subsets of \mathbb{R}^p . In contrast to the closed subsets, the second distributive law is satisfied. Another very important aspect is that closed convex sets can be equivalently described by its support functions. Rockafellar [1] introduced the concept of *oriented* closed convex sets. This concept plays a crucial role in this work. In the third section of this chapter, we investigate the possibility of embedding subsets of \hat{C} into a linear space. We divide this space into classes that can be embedded. These classes depend on the recession cone of the sets. With the aid of the orientation we can re-interpret the inverse elements of the embedded elements as sets (with opposite orientation).

1.1 The space $\hat{\mathcal{F}}$ of closed subsets of \mathbb{R}^p

In this section we summarize some simple facts about the space of closed subsets of \mathbb{R}^p . Note that all the assertions of this section remain valid if \mathbb{R}^p is replaced by an arbitrary linear topological space. The space of all *nonempty closed subsets* of \mathbb{R}^p is denoted by $\mathcal{F}(\mathbb{R}^p)$ and the space of all *closed subsets* of \mathbb{R}^p is denoted by $\hat{\mathcal{F}}(\mathbb{R}^p)$. For simplicity of notation, we just write \mathcal{F} and $\hat{\mathcal{F}}$, respectively.

In $\hat{\mathcal{F}}$ we introduce an addition \oplus : $\hat{\mathcal{F}} \times \hat{\mathcal{F}} \to \hat{\mathcal{F}}$ and a multiplication by nonnegative real numbers \cdot : $\mathbb{R}_+ \times \hat{\mathcal{F}} \to \hat{\mathcal{F}}$, defined by

$$\forall A, B \in \hat{\mathcal{F}}: \qquad A \oplus B := \operatorname{cl} \left(A + B\right) = \operatorname{cl} \left\{y \in \mathbb{R}^p | \exists a \in A, \exists b \in B : y = a + b\right\}$$

Chapter 1. The closed convex sets in \mathbb{R}^p

$$\forall A \in \hat{\mathcal{F}}, \ \alpha \ge 0: \qquad \alpha \cdot A := \begin{cases} \left\{ \begin{array}{ll} \{y \in \mathbb{R}^p | \ \exists a \in A : y = \alpha \, a \} & \text{if} \quad \alpha > 0 \\ \{0\} & \text{if} \quad \alpha = 0 \end{cases} \end{cases}$$

The latter distinction of cases ensures that $0 \cdot \emptyset = \{0\}$. As usual, we sometimes write αA instead of $\alpha \cdot A$. For all $A, B, C \in \hat{\mathcal{F}}$ (including the empty set) and all $\alpha, \beta \in \mathbb{R}_+$ the following calculus rules hold true:

$$\begin{array}{ll} (C1) \ (A \oplus B) \oplus C = A \oplus (B \oplus C), \\ (C3) \ A \oplus B = B \oplus A, \\ (C5) \ 1 \cdot A = A, \\ (C7) \ 0 \cdot A = \{0\}. \end{array}$$

$$\begin{array}{ll} (C1) \ (A \oplus B) \oplus C = A \oplus (B \oplus C), \\ (C2) \ \{0\} \oplus A = A, \\ (C4) \ \alpha \ (\beta \ A) = (\alpha \beta) \ A, \\ (C6) \ \alpha \ (A \oplus B) = \alpha \ A \oplus \alpha \ B, \end{array}$$

This means that $(\hat{\mathcal{F}}, \oplus, \cdot)$ is a conlinear space, see Appendix C. For simplicity, we say $\hat{\mathcal{F}}$ is a conlinear space. The rule (C1) (in particular, if \mathbb{R}^p is replaced by an arbitrary linear topological space) can be shown by Proposition A.4 (vii). The other rules are obvious.

The second distributive law is not valid in $\hat{\mathcal{F}}$. This is clear because the second distributive law would imply that each member of $\hat{\mathcal{F}}$ is a convex subset of \mathbb{R}^p .

The set inclusion provides a partial ordering in $\hat{\mathcal{F}}$ such that $(\hat{\mathcal{F}}, \subseteq)$ and $(\hat{\mathcal{F}}, \supseteq)$ are ordered conlinear spaces. Both spaces are order complete, see Appendix B. For instance, for a nonempty subset $\mathcal{A} \subseteq (\hat{\mathcal{F}}, \subseteq)$, the infimum and supremum can be expressed by

$$\operatorname{INF} \mathcal{A} = \bigcap_{A \in \mathcal{A}} A, \qquad \operatorname{SUP} \mathcal{A} = \operatorname{cl} \bigcup_{A \in \mathcal{A}} A.$$

1.2 The space \hat{C} of closed convex subsets of \mathbb{R}^p

This section deals with the space of closed convex subsets of \mathbb{R}^p , which plays an important role in the whole work. Many assertions in this section are also valid in a more general setting, i.e., \mathbb{R}^p can be replaced by a more general space.

The space of all *nonempty closed convex subsets* of \mathbb{R}^p is denoted by $\mathcal{C}(\mathbb{R}^p)$ and the space of all *closed convex subsets* of \mathbb{R}^p is denoted by $\hat{\mathcal{C}}(\mathbb{R}^p)$. For abbreviation, we continue to write \mathcal{C} and $\hat{\mathcal{C}}$ instead of $\mathcal{C}(\mathbb{R}^p)$ and $\hat{\mathcal{C}}(\mathbb{R}^p)$, respectively.

In $\hat{\mathcal{C}}$ we introduce the same addition and the same multiplication by nonnegative real numbers as in $\hat{\mathcal{F}}$. Of course, these operations are closed in $\hat{\mathcal{C}}$, i.e., $\hat{\mathcal{C}}$ is a conlinear subspace of $\hat{\mathcal{F}}$. Hence the rules (C1) – (C7) are valid. Additionally, the second distributive law is satisfied, i.e.,

(C8)
$$\forall A \in \hat{\mathcal{C}}, \forall \alpha, \beta \ge 0: \qquad \alpha A \oplus \beta A = (\alpha + \beta) A.$$

Conversely, $\hat{\mathcal{C}}$ is the set of all members of $\hat{\mathcal{F}}$ satisfying (C8). It is easy to see that neither $\hat{\mathcal{C}}$ nor \mathcal{C} is a linear space, since the axiom of existence of an inverse element is violated. Moreover, it is *not possible to embed* \mathcal{C} *into a linear space*. Indeed, assuming there is an

injective homomorphism j (an embedding) from C into a linear space L. For $K := \mathbb{R}^p_+ \in C$ we have $K = K \oplus K$. Hence j(K) = j(K) + j(K) and $0_L = j(\{0\}) \neq j(K)$. Then there must be an inverse element $l \in L$ of j(K), i.e., $j(K) + l = 0_L$. It follows $0_L = j(K) + l = j(K) + l + j(K) = j(K)$, a contradiction.

Let $K \subseteq \mathbb{R}^p$ be a nonempty closed convex cone. The set $\mathcal{C}_K \subseteq \hat{\mathcal{C}}$ is defined to be the set of all elements $A \in \mathcal{C}$ with $0^+A = K$. The following assertions tells us that $\mathcal{C}_K \subseteq \hat{\mathcal{C}}$ has the structure of a convex cone, even though $\hat{\mathcal{C}}$ is not a linear space. The concept of a cone and a convex set can be analogously defined as in the framework of linear spaces, for the details see Appendix C.

Proposition 1.2.1 C_K is a convex cone in \hat{C} .

Proof. Let $A, B \in C_K$. It remains to show $0^+(A \oplus B) = K$. This is a consequence of [1, Corollary 9.1.1] if we can verify the following condition: If $z_1 \in 0^+A$ and $z_2 \in 0^+B$ such that $z_1 + z_2 = 0$, then z_1 belongs to the lineality space of A and z_2 belongs to the lineality space of B. Indeed, we have $0^+A = 0^+B = K$ and the lineality spaces of A and B are equal, namely $0^+A \cap (-0^+A) = 0^+B \cap (-0^+B) = K \cap (-K)$. Hence the mentioned condition is satisfied. \Box

Note that [1, Corollary 9.1.1] also implies that the addition \oplus in $\mathcal{C}_K \subseteq \mathcal{C}$ reduces to the usual Minkowski addition +, i.e., the closure operation is superfluous.

Rockafellar [1, Section 39] introduced the concept of orientation of convex sets in \mathbb{R}^p . A convex set $A \subseteq \mathbb{R}^p$ that is identified with its convex indicator function $\delta(\cdot|A)$ is said to be supremum oriented and a convex set A that is identified with the concave function $-\delta(\cdot|A)$ is called infimum oriented. This concept plays a crucial role in our theory. Thus we introduce the following notation: The space $\hat{\mathcal{C}}^*$ is defined to be the space $\hat{\mathcal{C}}$ having supremum oriented members. By $\hat{\mathcal{C}}^\diamond$ we denote the same space, but with infimum oriented members. Analogously, we define \mathcal{C}^* , \mathcal{C}^\diamond , \mathcal{C}^*_K and \mathcal{C}^\diamond_K . If not stated otherwise, the orientation is not changed while manipulating sets. For instance, this means that $\hat{\mathcal{C}}^*$ and $\hat{\mathcal{C}}^\diamond$ are conlinear spaces, the recession cone of a supremum (infimum) oriented set is supremum (infimum) oriented, and so on.

The space $\hat{\mathcal{C}}$ is now equipped with one of the reflexive, transitive and antisymmetric relations \supseteq and \subseteq . We establish standard relations in dependence on the orientation of the members of the space. In fact, let the standard relation be \supseteq in the space $\hat{\mathcal{C}}^*$ and \subseteq in $\hat{\mathcal{C}}^\diamond$. Both these standard relations have the meaning of "less or equal". Since sets with opposite orientation will be used in order to define a substitute for inverse elements in the framework of the conlinear space $\hat{\mathcal{C}}$, this identification is useful to obtain the well-known formulas of the context of linear spaces. Further motivation is given later on.

Convention 1.2.2 Throughout this work we use the following conventions.

- (i) For $A_1, A_2 \in \mathcal{A} \subseteq \hat{\mathcal{C}}^*$ we write $A_1 \leq A_2$ instead of $A_1 \supseteq A_2$.
- (ii) For $A_1, A_2 \in \mathcal{A} \subseteq \hat{\mathcal{C}}^\diamond$ we write $A_1 \leq A_2$ instead of $A_1 \subseteq A_2$.
- (iii) We write $\mathcal{A} \subseteq \hat{\mathcal{C}}$ if the corresponding assertion is valid for both $\mathcal{A} \subseteq \hat{\mathcal{C}}^*$ or $\mathcal{A} \subseteq \hat{\mathcal{C}}^\diamond$.

For the notation $\mathcal{A} \subseteq \mathcal{C}$ and $\mathcal{A} \subseteq \mathcal{C}_K$ we use a convention analogous to (iii).

This convention means that certain assertions for subsets of $\hat{\mathcal{C}}$ have to be interpreted differently, where the interpretation depends on the orientation, and they are valid for both interpretations. For instance, for $A_1, A_2 \in \mathcal{A} \subseteq \hat{\mathcal{C}}$ we write

$$A_1 \leq A_2 \quad \text{if} \quad \begin{cases} A_1 \supseteq A_2 \text{ in case that } \mathcal{A} \subseteq \hat{\mathcal{C}}^* \\ A_1 \subseteq A_2 \text{ in case that } \mathcal{A} \subseteq \hat{\mathcal{C}}^\diamond. \end{cases}$$

Of course, $\hat{\mathcal{C}}^{\star}$ and $\hat{\mathcal{C}}^{\diamond}$ (equipped with its standard relations) are ordered conlinear spaces.

Proposition 1.2.3 The spaces \hat{C}^* and \hat{C}^\diamond are order complete and the infimum and supremum of nonempty sets $\mathcal{A} \subseteq \hat{C}^*$ and $\mathcal{B} \subseteq \hat{C}^\diamond$ can be expressed as follows:

$$\sup \mathcal{A} = \bigcap_{A \in \mathcal{A}} A, \qquad \inf \mathcal{A} = \operatorname{cl} \operatorname{conv} \bigcup_{A \in \mathcal{A}} A,$$
$$\sup \mathcal{B} = \operatorname{cl} \operatorname{conv} \bigcup_{B \in \mathcal{B}} B, \qquad \inf \mathcal{B} = \bigcap_{B \in \mathcal{B}} B.$$

Proof. Follows from the definition.

In every order complete ordered conlinear space $(\mathcal{Y}, \oplus, \preceq)$ it is evident that

$$\inf \mathcal{A} \oplus \inf \mathcal{B} \preceq \inf (\mathcal{A} + \mathcal{B}) \qquad \text{and} \qquad \sup (\mathcal{A} + \mathcal{B}) \preceq \sup \mathcal{A} \oplus \sup \mathcal{B}, \tag{1.1}$$

where $\mathcal{A} + \mathcal{B} := \{A \oplus B | A \in \mathcal{A}, B \in \mathcal{B}\}$. In particular, (1.1) is valid in $\hat{\mathcal{C}}^{\star}$ and $\hat{\mathcal{C}}^{\diamond}$. In general, (1.1) does not hold with equality, see Example 1.2.5 below. We next show that, in dependence on the orientation, one inequality in (1.1) is even satisfied with equality. This equality is essential in duality theory, see Remark 3.5.4 below.

Proposition 1.2.4 For nonempty sets $\mathcal{A}, \mathcal{B} \subseteq \hat{\mathcal{C}}^*$ and $\bar{\mathcal{A}}, \bar{\mathcal{B}} \subseteq \hat{\mathcal{C}}^\diamond$ it holds

 $\inf(\mathcal{A} + \mathcal{B}) = \inf \mathcal{A} \oplus \inf \mathcal{B}$ and $\sup(\bar{\mathcal{A}} + \bar{\mathcal{B}}) = \sup \bar{\mathcal{A}} \oplus \sup \bar{\mathcal{B}}.$

Proof. It holds

$$\inf(\mathcal{A} + \mathcal{B}) \stackrel{\Pr. 1.2.3}{=} \operatorname{cl}\operatorname{conv} \bigcup_{C \in \mathcal{A} + \mathcal{B}} C = \operatorname{cl}\operatorname{conv} \bigcup_{A \in \mathcal{A}, B \in \mathcal{B}} (A \oplus B)$$

$$\leq \operatorname{cl}\operatorname{conv} \bigcup_{A \in \mathcal{A}, B \in \mathcal{B}} (A + B) = \operatorname{cl}\operatorname{conv} \left(\bigcup_{A \in \mathcal{A}} A + \bigcup_{B \in \mathcal{B}} B\right)$$

$$\stackrel{\Pr. A.4 (ix)}{=} \operatorname{cl} \left(\operatorname{cl}\operatorname{conv} \bigcup_{A \in \mathcal{A}} A + \operatorname{cl}\operatorname{conv} \bigcup_{B \in \mathcal{B}} B\right) \stackrel{\Pr. 1.2.3}{=} \inf \mathcal{A} \oplus \inf \mathcal{B}.$$

By (1.1) (or directly) we deduce equality. The second part is completely analogous.

In general, the latter assertion is not true for the supremum in \hat{C}^* and the infimum in \hat{C}^{\diamond} , as the following example shows.

Example 1.2.5 Let $\mathcal{A}, \mathcal{B} \subseteq \mathcal{C}^{\star}(\mathbb{R}^2), \ \mathcal{A} := \{A_1, A_2\}, \text{ where } A_1 := \{(0, 1)^T\} + \mathbb{R}^2_+, \ A_2 := \{(1, 0)^T\} + \mathbb{R}^2_+, \text{ and } \mathcal{B} := \{\mathbb{B}\}.$ Then, $\sup \mathcal{A} \oplus \sup \mathcal{B} = (A_1 \cap A_2) \oplus \mathbb{B} = \{(1, 1)^T\} + \mathbb{B} + \mathbb{R}^2_+.$ But, $\sup(\mathcal{A} + \mathcal{B}) = \sup\{A_1 \oplus \mathbb{B}, A_2 \oplus \mathbb{B}\} = \mathbb{R}^2_+, \text{ i.e., } \sup(\mathcal{A} + \mathcal{B}) \neq \sup \mathcal{A} \oplus \sup \mathcal{B}.$



Up to now, all the operations used did not influence the orientation of a set. We express the change of the orientation of a set as follows: Given an oriented set A we denote by $\boxplus A$ the same set, but with opposite orientation. As usual, the negative of a convex set A is defined by

$$-A := \left\{ y \in \mathbb{R}^p | -y \in A \right\}.$$

By convention, if A is an oriented set, this operation does not manipulate the orientation of A. In contrast to this, we introduce a second concept of a negative of a convex set which does so. Given an oriented set A, we define $\Box A$ being the set -A, but with the opposite orientation. Instead of two signs, we now have four signs, namely $+, -, \boxplus, \Box$. Obviously, the following assertions hold true:

$$A = \boxplus \boxplus A = \boxminus \square A, \qquad -A = \boxplus \square A = \boxminus \blacksquare A, \boxplus A = + \boxplus A = - \square A, \qquad \square A = + \square A = - \boxplus A.$$

Clearly, an expression is independent of the order of the signs. Note that \boxplus and \boxminus are signs but not operations. This means, the addition of contrarily oriented elements is not defined. Nevertheless, we write $A \boxplus B := A + (\boxplus B)$ and $A \boxminus B := A + (\boxplus B)$, if these expressions are defined, i.e., A and B are contrarily oriented. For a given set $\mathcal{A} \subseteq \hat{\mathcal{C}}$ we set $\boxminus \mathcal{A} := \{ \boxminus A \mid A \in \mathcal{A} \}$. It can be easily shown (Proposition 1.2.3) that

$$\exists \inf \mathcal{A} = \sup \exists \mathcal{A} \quad \text{and} \quad \exists \sup \mathcal{A} = \inf \exists \mathcal{A}.$$
(1.2)

Further motivation for the usage of the sign \boxminus is given later on. For instance, we define convex and concave functions with values in $\hat{\mathcal{C}}$ and we obtain the well-known convexityconcavity dualism for such functions, i.e., a function $f : X \to \hat{\mathcal{C}}$ is convex if and only if $\boxminus f : X \to \hat{\mathcal{C}}$ is concave. In the next section we embed certain subsets of \mathcal{C}^* and \mathcal{C}° in a (the same) linear space. There, $\boxminus A$ gets the meaning of an "inverse element" of A with respect to this embedding.

1.3 Embedding subsets of \hat{C} into a linear space

Embedding of spaces of convex sets into linear spaces was already investigated by Rådström [67], and as remarked there, the idea seems to go back to investigations of Brunn [15] in 1889. A nice overview over the literature in this field can be found in [3]. In the literature, compactness or boundedness assumptions to the sets are usually supposed. In this section, we investigate spaces of unbounded convex sets, too.

The aim of this section is to embed the convex cone $C_K \subseteq C$ into a partially ordered linear space. In dependence on the orientation of the members of C_K we use different embedding maps. This procedure allows us to re–interpret the inverse element of the embedding map's image of a member of C_K as an element of C_{-K} having opposite orientation.

The following lemma is a refinement of [1, Theorem 13.1]. It is shown that only the set $\operatorname{ri}(0^+A)^\circ := \operatorname{ri}((0^+A)^\circ)$ (instead of the whole space \mathbb{R}^p) is essential for the description of a nonempty closed convex set by its support function.

Lemma 1.3.1 Let $A, B \subseteq \mathbb{R}^p$ be a nonempty closed and convex. Then,

- (i) ri $(0^+A)^\circ \subseteq \operatorname{dom} \delta^*(\cdot | A) \subseteq (0^+A)^\circ$,
- (ii) $A = \bigcap_{y^* \in \mathrm{ri}\,(0^+A)^\circ} \{ y \in \mathbb{R}^p | \langle y^*, y \rangle \le \delta^* \left(y^* | A \right) \},$

(iii)
$$A \subseteq B \quad \Leftrightarrow \quad \forall y^* \in \operatorname{ri}(0^+B)^\circ: \ \delta^*(y^*|A) \le \delta^*(y^*|B),$$

(iv) If 0^+B is pointed, then

$$A \subseteq B \quad \Leftrightarrow \quad \forall y^* \in \mathbb{R}^p \setminus \operatorname{rb} \left(0^+ A \right)^\circ \colon \, \delta^*(y^*|A) \le \delta^*(y^*|B).$$

Proof. (i) As a consequence of [1, Theorem 14.2] we have $\operatorname{cl} \operatorname{dom} \delta^*(\cdot | A) = (0^+ A)^\circ$, compare [36, Theorem 2.2.4], too. Together with [1, Theorem 6.3] this yields $\operatorname{ri}(0^+ A)^\circ = \operatorname{ri} \operatorname{cl} \operatorname{dom} \delta^*(\cdot | A) \subseteq \operatorname{dom} \delta^*(\cdot | A) \subseteq (0^+ A)^\circ$.

(ii) From [1, Theorem 13.1] and (i) we obtain

$$A = \bigcap_{y^* \in \mathbb{R}^p} \left\{ y \in \mathbb{R}^p | \langle y^*, y \rangle \le \delta^* \left(y^* | A \right) \right\} = \bigcap_{y^* \in (0^+ A)^\circ} \left\{ y \in \mathbb{R}^p | \langle y^*, y \rangle \le \delta^* \left(y^* | A \right) \right\}.$$

It remains to show

$$Y_1 := \bigcap_{y^* \in (0^+A)^{\circ}} \left\{ y \in \mathbb{R}^p | \langle y^*, y \rangle \le \delta^* \left(y^* | A \right) \right\} = \bigcap_{y^* \in \mathrm{ri} \, (0^+A)^{\circ}} \left\{ y \in \mathbb{R}^p | \langle y^*, y \rangle \le \delta^* \left(y^* | A \right) \right\} =: Y_2.$$

The inclusion $Y_1 \subseteq Y_2$ is obvious. In order to show $Y_2 \subseteq Y_1$ let $y \in Y_2$ be arbitrarily chosen. It holds $\langle y^*, y \rangle \leq \delta^* (y^* | A)$ for all $y^* \in \operatorname{ri} (0^+ A)^\circ$. Let $\bar{y}^* \in (0^+ A)^\circ$ and $y^* \in \operatorname{ri} (0^+ A)^\circ$, then $\lambda \bar{y}^* + (1 - \lambda)y^* \in \operatorname{ri} (0^+ A)^\circ$ for all $\lambda \in [0, 1)$ (compare [1, Theorem 6.1]). It follows

$$\left\langle \lambda \bar{y}^* + (1-\lambda)y^*, y \right\rangle \le \delta^* \left(\lambda \bar{y}^* + (1-\lambda)y^* | A\right) \le \lambda \, \delta^* \left(\bar{y}^* | A \right) + (1-\lambda)\delta^* \left(y^* | A \right) + (1$$

By virtue of (i) we deduce that $\delta^*(y^*|A) < +\infty$. Letting $\lambda \to 1$ we obtain $\langle \bar{y}^*, y \rangle \leq \delta^*(\bar{y}^*|A)$, i.e., $y \in Y_1$.

(iii) It remains to show " \Leftarrow ". Let $y \in A$ be given. Then we have $\langle y^*, y \rangle \leq \delta^*(y^*|A)$ for all $y^* \in \mathbb{R}^p$ and so $\langle y^*, y \rangle \leq \delta^*(y^*|B)$ for all $y^* \in \operatorname{ri}(0^+B)^\circ$. From (ii) we obtain $y \in B$.

(iv) Again, it remains to show " \Leftarrow ". Let $y^* \in \mathbb{R}^p \setminus (0^+ A)^\circ$. By (i) we obtain $+\infty = \delta^*(y^*|A) \le \delta^*(y^*|B)$ and hence $y^* \notin \operatorname{ri}(0^+ B)^\circ$. This means we have $\operatorname{ri}(0^+ B)^\circ \subseteq (0^+ A)^\circ$. Since $0^+ B$ is pointed, we deduce that $\operatorname{ri}(0^+ B)^\circ = \operatorname{int}(0^+ B)^\circ \subseteq \operatorname{int}(0^+ A)^\circ = \operatorname{ri}(0^+ A)^\circ \subseteq \mathbb{R}^p \setminus \operatorname{rb}(0^+ A)^\circ$. From (iii) we deduce $A \subseteq B$.

The following example shows that the pointedness assumption in statement (iv) of the preceding lemma cannot be omitted.

Example 1.3.2 Let $A = \mathbb{R}^2_+$ and $B = \{y \in \mathbb{R}^2 | y_1 \ge 1\}$. Then we have $(0^+B)^\circ \subseteq \operatorname{rb}(0^+A)^\circ$ and, by (i), the right-hand side of (iv) is satisfied. But $A \not\subseteq B$.



The preceeding lemma is an essential tool for the proof of duality assertions for $\hat{\mathcal{C}}$ -valued functions. Furthermore, in the next chapter it is used to characterize a new type of convergence of convex sets by convergence of support functions. As a byproduct, we obtain the statement that \mathcal{C}_K^* and \mathcal{C}_K^\diamond can be embedded into a linear space. We first give an equivalent characterization of the ordered conlinear spaces \mathcal{C}_K^* and \mathcal{C}_K^\diamond (where K is the neutral element, i.e., they are cones but not conlinear subspaces of \mathcal{C}^* and \mathcal{C}^\diamond) by certain spaces of real-valued functions having the same domain (which depends on K). Then, the possibility of embedding into a linear space is obvious.

Let Γ_K^* be the space of all positively homogeneous concave functions from ri K° into \mathbb{R} and let Γ_K^\diamond be the space of all positively homogeneous convex functions from ri K° into \mathbb{R} . The spaces Γ_K^* and Γ_K^\diamond are conlinear spaces with respect to the addition and multiplication by nonnegative real numbers, being defined pointwise using the corresponding operation in \mathbb{R} . Moreover, Γ_K^* and Γ_K^\diamond equipped with the ordering relation \leq , which is defined pointwise using the usual \leq relation in \mathbb{R} , are ordered conlinear spaces.

Theorem 1.3.3 Let $K \subseteq \mathbb{R}^p$ be a nonempty closed convex cone. Then,

(i) There exists a bijective map $j^* : \mathcal{C}_K^* \to \Gamma_K^*$ such that for all $A, B \in \mathcal{C}_K^*$ and all positive real numbers $\alpha > 0$ it holds

(a)
$$j^{\star}(A+B) = j^{\star}(A) + j^{\star}(B)$$
, (b) $j^{\star}(\alpha A) = \alpha j^{\star}(A)$,

(c)
$$j^{\star}(K) = 0_{\Gamma_{K}^{\star}},$$
 (d) $A \supseteq B \Leftrightarrow j^{\star}(A) \le j^{\star}(B).$

(ii) There exists a bijective map $j^{\diamond} : \mathcal{C}_{-K}^{\diamond} \to \Gamma_{K}^{\diamond}$ such that for all $A, B \in \mathcal{C}_{-K}^{\diamond}$ and all positive real numbers $\alpha > 0$ it is true that

(a)
$$j^{\diamond}(A+B) = j^{\diamond}(A) + j^{\diamond}(B)$$
, (b) $j^{\diamond}(\alpha A) = \alpha j^{\diamond}(A)$,

(c)
$$j^{\diamond}(-K) = 0_{\Gamma_K^{\diamond}}$$
, (d) $A \subseteq B \Leftrightarrow j^{\diamond}(A) \le j^{\diamond}(B)$.

Proof. (i) Consider the map j^* , assigning to every $A \in \mathcal{C}_K^*$ the negative support function of the set $A \subseteq \mathbb{R}^p$, restricted to the set ri $K^\circ \subseteq \mathbb{R}^p$. More precisely, $\gamma_A = j^*(A)$ is defined by $\gamma_A : \operatorname{ri} K^\circ \to \mathbb{R} \cup \{-\infty, +\infty\}, \gamma_A(y^*) := -\delta^*(y^*|A)$. The map j^* is a function from \mathcal{C}_K^* into Γ_K^* . Indeed, let $A \in \mathcal{C}_K^*$. Since A is nonempty, we have $\delta^*(y^*|A) > -\infty$ for all $y^* \in \mathbb{R}^p$. With the aid of Lemma 1.3.1 (i) we obtain $\delta^*(y^*|A) < +\infty$ for all $y^* \in \operatorname{ri} K^\circ$. Hence $\gamma_A = j^*(A)$ only attains values in \mathbb{R} . Since support functions are sublinear and ri K° is a convex cone, $\gamma_A = j^*(A)$ is positively homogeneous and concave on ri K° .

Lemma 1.3.1 (ii) yields that $j^* : \mathcal{C}_K^* \to \Gamma_K^*$ is injective.

We next show that $j^* : \mathcal{C}_K^* \to \Gamma_K^*$ is surjective. Given some $\gamma \in \Gamma_K^*$ we define a function $d : \mathbb{R}^p \to \mathbb{R} \cup \{+\infty\}$ by

$$d(y^*) := \begin{cases} -\gamma(y^*) & \text{if} \quad y^* \in \operatorname{ri} K^c \\ +\infty & \text{else.} \end{cases}$$

It is easy to see that d is convex, positively homogeneous, not identically $+\infty$ and never $-\infty$. With the aid of [1, Corollary 13.2.1] we conclude that $\operatorname{cl} d$ is the support function of the nonempty closed convex set

$$A_{\gamma} := \bigcap_{y^* \in \mathrm{ri}\,K^{\circ}} \left\{ y \in \mathbb{R}^p | \langle y^*, y \rangle \le d(y^*) \right\} = \bigcap_{y^* \in \mathrm{ri}\,K^{\circ}} \left\{ y \in \mathbb{R}^p | \langle y^*, y \rangle \le -\gamma(y^*) \right\}.$$

Applying [1, Corollary 8.3.3], taking into account the considerations in [1, page 62] and applying Lemma 1.3.1 (ii) we obtain

$$0^+ A_{\gamma} = \bigcap_{y^* \in \mathrm{ri}\, K^{\circ}} 0^+ \left\{ y \in \mathbb{R}^p | \langle y^*, y \rangle \le d(y^*) \right\} = \bigcap_{y^* \in \mathrm{ri}\, K^{\circ}} \left\{ y \in \mathbb{R}^p | \langle y^*, y \rangle \le 0 \right\} = K.$$

By definition, we have $j^*(A_{\gamma})(y^*) = -\operatorname{cl} d(y^*)$ for all $y^* \in \operatorname{ri} K^\circ$. With the aid of [1, Theorem 7.4] we have $\operatorname{cl} d(y^*) = d(y^*)$ for all $y^* \in \operatorname{ri} K^\circ$. Hence $j^*(A_{\gamma}) = \gamma$.

(i)(a) and (i)(b) follow from elementary properties of the supremum in \mathbb{R} , compare [1, page 113], too. (i)(c) follows from the definition of the polar cone and of the support function. (i)(d) is a consequence of [1, Corollary 13.1.1].

(ii) Define
$$j^{\diamond}(A) := -j^{\star}(\Box A)$$
 and use (i). \Box

Let $\hat{\mathcal{C}}_K^\star := \mathcal{C}_K^\star \cup \{\emptyset\} \subseteq \hat{\mathcal{C}}^\star$ and let $\hat{\Gamma}_K^\star := \Gamma_K^\star \cup \{+\infty_K\}$, where $+\infty_K$ is an abbreviation for the function defined on ri K° and being identically $+\infty$. Likewise, we define $\hat{\mathcal{C}}_K^\diamond := \mathcal{C}_K^\diamond \cup \{\emptyset\} \subseteq \hat{\mathcal{C}}^\diamond$ and $\hat{\Gamma}_K^\diamond := \Gamma_K^\diamond \cup \{-\infty_K\}$.

Corollary 1.3.4 Theorem 1.3.3 remains valid if \mathcal{C}_{K}^{\star} , $\mathcal{C}_{-K}^{\diamond}$, Γ_{K}^{\star} and Γ_{K}^{\diamond} are replaced by $\hat{\mathcal{C}}_{K}^{\star}$, $\hat{\mathcal{C}}_{-K}^{\diamond}$, $\hat{\Gamma}_{K}^{\star}$ and $\hat{\Gamma}_{K}^{\diamond}$, respectively.

Proof. Extend the isomorphisms j^* and j^\diamond of Theorem 1.3.3 using the conventions $j^*(\emptyset) = +\infty_K$ and $j^\diamond(\emptyset) = -\infty_K$, respectively.

Let Γ_K be the space of all positively homogeneous (and not necessarily convex or concave) functions $\gamma : \operatorname{ri} K^\circ \to \mathbb{R}$. Let Γ_K be equipped with an addition and a scalar multiplication, defined pointwise using the corresponding operation in \mathbb{R} , and with an ordering relation \leq , defined pointwise using the usual \leq relation in \mathbb{R} . Then, the space Γ_K is a partially ordered linear space. Theorem 1.3.3 yields that $(\mathcal{C}_K^*, \supseteq)$ and $(\mathcal{C}_{-K}^\circ, \subseteq)$ are isomorphic to convex cones in the partially ordered linear space (Γ_K, \leq) . Let $j^* : \mathcal{C}_K^* \to \Gamma_K$ be the injective homomorphism which embeds \mathcal{C}_K^* into Γ_K and let $j^\diamond : \mathcal{C}_{-K}^\diamond \to \Gamma_K$ be analogously defined. Then it easily follows that

$$\forall A \in \mathcal{C}_K^\star : \ j^\star(A) + j^\diamond(\Box A) = 0, \qquad \forall A \in \mathcal{C}_{-K}^\diamond : \ j^\diamond(A) + j^\star(\Box A) = 0. \tag{1.3}$$

In this sense, $\boxminus A$ can be regarded as the "inverse element" of a nonempty closed convex set A. However, this does not imply that $\mathcal{C}_{K}^{\star} \cup \mathcal{C}_{-K}^{\diamond}$ is a linear space, because it is not a conlinear space.

Of course, $\hat{\mathcal{C}}_{K}^{\star}$ and $\hat{\mathcal{C}}_{K}^{\diamond}$ can also be embedded into a smaller linear space than Γ_{K} , for instance in the space of real-valued positively homogeneous DC-functions (e.g. [81]) being defined on ri K° . However, the main advantage of Γ_{K} is that infimum and supremum in Γ_{K} can be described pointwise by the infimum and supremum in \mathbb{R} . This means that the space (Γ_{K}, \leq) is Dedekind complete and for nonempty subsets $\mathcal{A} \subseteq \Gamma_{K}$ it holds

$$\mathcal{A} \text{ bounded above } \Rightarrow \forall y^* \in \operatorname{ri} K^\circ : \left(\sup_{\gamma \in \mathcal{A}} \gamma \right) (y^*) = \sup_{\gamma \in \mathcal{A}} \left(\gamma(y^*) \right),$$

$$\mathcal{A} \text{ bounded below } \Rightarrow \forall y^* \in \operatorname{ri} K^\circ : \left(\inf_{\gamma \in \mathcal{A}} \gamma \right) (y^*) = \inf_{\gamma \in \mathcal{A}} \left(\gamma(y^*) \right).$$

Let \mathcal{A}, \mathcal{B} be nonempty subsets of \mathcal{C}_K such that \mathcal{A} is bounded below and \mathcal{B} bounded above (see Convention 1.2.2). Further, let j be the map which embeds \mathcal{C}_K into the linear space Γ_K (where j stands for j^* or j° as defined above). Then, from Proposition B.1 we conclude that

$$j(\inf \mathcal{A}) \le \inf_{A \in \mathcal{A}} j(A)$$
 and $j(\sup \mathcal{B}) \ge \sup_{B \in \mathcal{B}} j(B).$ (1.4)

The following proposition shows that, in dependence of the orientation, one assertion in (1.4) even holds with equality.

Proposition 1.3.5 Let $\mathcal{A} \subseteq \mathcal{C}_K^*$ be nonempty and bounded below and let $\mathcal{B} \subseteq \mathcal{C}_K^\diamond$ be nonempty and bounded above. Then,

$$j^{\star}(\inf \mathcal{A}) = \inf_{A \in \mathcal{A}} j^{\star}(A)$$
 and $j^{\diamond}(\sup \mathcal{B}) = \sup_{B \in \mathcal{B}} j^{\diamond}(B).$

Proof. Taking into account Proposition B.1, it remains to show that $\inf_{A \in \mathcal{A}} j^*(A) \in \Gamma_K^*$ and $\sup_{B \in \mathcal{B}} j^{\diamond}(B) \in \Gamma_K^{\diamond}$. This is true because the pointwise infimum (supremum) over a set of concave (convex) functions is concave (convex), too.

The following example shows that, in general, (1.4) is not satisfied with equality.

Example 1.3.6 Let $\mathcal{A} = \{A_1, A_2\} \subseteq \hat{\mathcal{C}}^*(\mathbb{R}^2)$ as in Example 1.2.5. Then, $j^*(\sup \mathcal{A}), j^*(A_1)$ and $j^*(A_2)$ are functions from $-\operatorname{int} \mathbb{R}^2_+$ into \mathbb{R} , given by

$$j^{*}(\sup \mathcal{A})(y^{*}) = -\delta^{*}(y^{*}|A_{1} \cap A_{2}) = -(y_{1}^{*} + y_{2}^{*}),$$
$$j^{*}(A_{1})(y^{*}) = -\delta^{*}(y^{*}|A_{1}) = -y_{2}^{*}, \qquad j^{*}(A_{2})(y^{*}) = -\delta^{*}(y^{*}|A_{2}) = -y_{1}^{*}$$

Hence, $\sup_{A \in \mathcal{A}} j^{\star}(A) \neq j^{\star}(\sup \mathcal{A}).$

The boundedness assumptions in (1.4) are very restrictive, because the infimum with respect to $\hat{\mathcal{C}}^{\star}$ over a subset $\mathcal{A} \subseteq \hat{\mathcal{C}}_{K}^{\star} \subseteq \hat{\mathcal{C}}^{\star}$ often has a recession cone larger than K. The following result is valid without boundedness assumptions. It is used to prove duality assertions for $\hat{\mathcal{C}}$ -valued functions.

Proposition 1.3.7 Let $\mathcal{A} := \{A_i \in \hat{\mathcal{C}}^* | i \in I\} \subseteq \hat{\mathcal{C}}^*$, where I is an arbitrary index set. Then it holds

$$\forall y^* \in \mathbb{R}^p : -\delta^* (y^* | \inf_{i \in I} A_i) = \inf_{i \in I} \{ -\delta^* (y^* | A_i) \}, \\ \forall y^* \in \mathbb{R}^p : -\delta^* (y^* | \sup_{i \in I} A_i) \ge \sup_{i \in I} \{ -\delta^* (y^* | A_i) \}.$$

Proof. Without loss of generality we can assume $\mathcal{A} \subseteq \mathcal{C}^*$. We have $\inf_{i \in I} A_i = \operatorname{cl} \operatorname{conv} \bigcup_{i \in I} A_i$. Hence, the first assertion follows from the first part of [1, Corollary 16.5.1].

Since $\sup_{i \in I} A_i = \bigcap_{i \in I} A_i$, the second part of [1, Corollary 16.5.1] yields $\delta^*(\cdot |\sup_{i \in I} A_i) = \operatorname{cl\,conv} \left\{ \delta^*(\cdot |A_i) | i \in I \right\}$, where the convex hull of a collection of functions is defined as the convex hull of the pointwise infimum of the collection, i.e., $\operatorname{cl\,conv} \left\{ \delta^*(\cdot |A_i) | i \in I \right\} = \operatorname{cl\,conv} \inf_{i \in I} \delta^*(\cdot |A_i)$, compare [1, page 37]. It follows that $\delta^*(\cdot |\sup_{i \in I} A_i) \leq \inf_{i \in I} \delta^*(\cdot |A_i)$ which proves the second assertion.

Chapter 2

Convergence of closed convex sets

In this chapter, we introduce a convergence concept for closed convex subsets of \mathbb{R}^p , which seems to be new. This convergence is called \mathcal{C} -convergence. It is defined by appropriate notions of upper and lower limits. We compare this convergence with the well-known Painlevé-Kuratowski convergence [47], [4], [68] as well as with scalar convergence (i.e., the pointwise convergence of support functions) of convex sets [82], [69], [70], [74]. In Chapter 3, \mathcal{C} -convergence is used to define a meaningful concept of "lower semi-continuous hull" of a $\hat{\mathcal{C}}$ -valued functions. Although \mathcal{C} -convergence does not coincide with scalar convergence, it can be equivalently described by convergence of support functions. In fact, we show that a sequence $\{A_n\}_{n\in\mathbb{N}}$ \mathcal{C} -converges to \bar{A} if and only if the corresponding support functions converge pointwise, except at relative boundary points of the domain the support function of \bar{A} , to the support function of \bar{A} . This characterization of \mathcal{C} -convergence is used in the next chapter in order to prove a biconjugation theorem for $\hat{\mathcal{C}}$ -valued functions.

This chapter is organized as follows. In Section 2.1 we recall some basic properties of the well-known Painlevé-Kuratowski convergence. As mentioned in [68, page 111], the natural setting for the study of Painlevé-Kuratowski convergence is the space $\hat{\mathcal{F}}$ of closed subsets of \mathbb{R}^p . Section 2.2 is devoted to \mathcal{C} -convergence. We proceed analogously to the previous section by introducing upper and lower limits. However, in contrast to Painlevé-Kuratowski convergence, \mathcal{C} -convergence is adapted to the space $\hat{\mathcal{C}}$ of closed convex subsets of \mathbb{R}^p . We show that, under certain assumptions, our new concepts of upper and lower limits coincide with the well-known concepts related to Painlevé-Kuratowski convergence. In Section 2.3 we investigate the relationship between \mathcal{C} -convergence and scalar convergence.

2.1 Painlevé–Kuratowski convergence

In this section, we summarize some results about Painlevé–Kuratowski convergence in the space $\hat{\mathcal{F}}$ of closed subsets of \mathbb{R}^p . Our main reference is the book of Rockafellar/Wets [68]. We

frequently use the following notation of [68]:

$$\mathcal{N}_{\infty} := \{ N \subseteq \mathbb{N} | \mathbb{N} \setminus N \text{ finite} \} \text{ and } \mathcal{N}_{\infty}^{\#} := \{ N \subseteq \mathbb{N} | N \text{ infinite} \}.$$

Let $\{y_n\}_{n\in\mathbb{N}}\subseteq\mathbb{R}^p$ be a sequence. Sometimes we write $\bar{y} = \lim_{n\in\mathbb{N}} y_n$ or $y_n \to \bar{y}$ instead of $\bar{y} = \lim_{n\to\infty} y_n$. Moreover, we write $\bar{y} = \lim_{n\in\mathbb{N}} y_n$ or $y_n \xrightarrow{N} \bar{y}$ in the case of convergence of a subsequence designated by an index set $N \in \mathcal{N}_{\infty}^{\#}$ or $N \in \mathcal{N}_{\infty}$. It is clear that every subsequence of $\{y_n\}_{n\in\mathbb{N}}$ can be expressed by $\{y_n\}_{n\in\mathbb{N}}$, where N belongs to $\mathcal{N}_{\infty}^{\#}$. In case of $N \in \mathcal{N}_{\infty}$, $\{y_n\}_{n\in\mathbb{N}}$ denotes a subsequence of $\{y_n\}_{n\in\mathbb{N}}$ that arises by omitting finitely many members. For example, a subsequence of a subsequence $\{y_n\}_{n\in\mathbb{N}}$ ($N \in \mathcal{N}_{\infty}^{\#}$) can be expressed by some $\bar{N} \in \mathcal{N}_{\infty}^{\#}$ with $\bar{N} \subseteq N$ as $\{y_n\}_{n\in\bar{N}}$. An analogous notation is used for sequences in other spaces, where the limit has to be defined appropriately.

Definition 2.1.1 ([68]) For a sequence $\{A_n\}_{n\in\mathbb{N}} \subseteq \hat{\mathcal{F}}$ the outer limit is the set

$$\operatorname{LIM}_{n \to \infty} \operatorname{SUP} A_n := \left\{ y \in \mathbb{R}^p | \; \exists N \in \mathcal{N}_{\infty}^{\#}, \; \forall n \in N, \; \exists y_n \in A_n : y_n \xrightarrow{N} y \right\}$$

The inner limit of a sequence $\{A_n\}_{n\in\mathbb{N}}\subseteq\hat{\mathcal{F}}$ is the set

$$\operatorname{LIM}_{n \to \infty} \operatorname{INF} A_n := \left\{ y \in \mathbb{R}^p | \exists N \in \mathcal{N}_{\infty}, \forall n \in N, \exists y_n \in A_n : y_n \xrightarrow{N} y \right\}.$$

The limit of the sequence exists if the outer and inner limits coincide. Then we write

$$\lim_{n \to \infty} A_n = \limsup_{n \to \infty} A_n = \limsup_{n \to \infty} A_n.$$

In [68], the closedness of the members of the sequence is not supposed a priori. However, as it can be seen in [68, Proposition 4.4], the outer and inner limits only depend on the closure of the sequence's members.

In contrast to [68], we use capital letters in the notation of the (outer and inner) limit. This is because the notation with small letters is reserved for the (upper and lower) limit in the space \hat{C} to be defined later on. The following characterization of outer and inner limits is very important for the considerations in the next section.

Proposition 2.1.2 ([68]) For a sequence $\{A_n\}_{n\in\mathbb{N}} \subseteq \hat{\mathcal{F}}$ it holds

$$\operatorname{LIM}_{n \to \infty} \operatorname{SUP} A_n = \bigcap_{N \in \mathcal{N}_{\infty}} \operatorname{cl} \bigcup_{n \in N} A_n, \qquad \qquad \operatorname{LIM}_{n \to \infty} \operatorname{INF} A_n = \bigcap_{N \in \mathcal{N}_{\infty}^{\#}} \operatorname{cl} \bigcup_{n \in N} A_n.$$

Proof. See [68, Exercise 4.2.(b)].

We observe that the characterization in the preceeding proposition can be expressed by the supremum and infimum in the space $\hat{\mathcal{F}}$. According to the definition of outer and inner limits of [68] the relation \subseteq has the meaning of "less or equal", i.e., the infimum is related to the intersection and the supremum is related to the closure of the union. Hence, we can write

$$\underset{n \to \infty}{\text{LIM SUP}} A_n = \underset{N \in \mathcal{N}_{\infty}}{\text{INF}} \underset{n \in N}{\text{SUP}} A_n, \qquad \qquad \underset{n \to \infty}{\text{LIM INF}} A_n = \underset{N \in \mathcal{N}_{\infty}^{\#}}{\text{INF}} \underset{n \in N}{\text{SUP}} A_n.$$
(2.1)

This characterization is the starting point in the next section.

2.2 C-Convergence

In the preceding section, we have seen that Painlevé–Kuratowski convergence can be expressed with the aid of the supremum and infimum in the space $\hat{\mathcal{F}}$. Likewise, we introduce corresponding concepts in the space $\hat{\mathcal{C}}$. This means, the supremum and infimum notions in the following definition are based on the considerations in Section 1.2.

Definition 2.2.1 The upper limit and the lower limit of a sequence $\{A_n\}_{n\in\mathbb{N}} \subseteq \hat{\mathcal{C}}^*$ are defined, respectively, by

$$\limsup_{n \to \infty} A_n := \sup_{N \in \mathcal{N}_{\infty}^{\#}} \inf_{n \in N} A_n \quad and \quad \liminf_{n \to \infty} A_n := \sup_{N \in \mathcal{N}_{\infty}} \inf_{n \in N} A_n.$$

The upper limit and the lower limit of a sequence $\{A_n\}_{n\in\mathbb{N}}\subseteq \hat{\mathcal{C}}^\diamond$ are defined, respectively, by

$$\limsup_{n \to \infty} A_n := \inf_{N \in \mathcal{N}_{\infty}} \sup_{n \in N} A_n \quad and \quad \liminf_{n \to \infty} A_n := \inf_{N \in \mathcal{N}_{\infty}^{\#}} \sup_{n \in N} A_n.$$

In the following, we frequently use Convention 1.2.2. For a sequence $\{A_n\}_{n\in\mathbb{N}}\subseteq \hat{\mathcal{C}}$ it is evident that

$$\begin{split} \limsup_{n \to \infty} -A_n &= -\limsup_{n \to \infty} A_n, \qquad \liminf_{n \to \infty} -A_n &= -\liminf_{n \to \infty} A_n, \\ \limsup_{n \to \infty} &\boxplus A_n &= \boxplus \liminf_{n \to \infty} A_n, \qquad \limsup_{n \to \infty} &\boxplus A_n &= \boxplus \liminf_{n \to \infty} A_n. \end{split}$$

In the space $\mathbb{R} \cup \{-\infty, +\infty\}$, we only know the latter equality (with the usual "-" instead of " \square "), where the first two equalities are also valid in the framework of Painlevé–Kuratowski convergence.

Proposition 2.2.2 Let $\{A_n\}_{n \in \mathbb{N}} \subseteq \hat{\mathcal{C}}$ be a sequence. Then,

$$\liminf_{n \to \infty} A_n \le \limsup_{n \to \infty} A_n.$$

Proof. Since $\mathcal{N}_{\infty} \subseteq \mathcal{N}_{\infty}^{\#}$, this follows from the definition.

The upper and lower limits can be used to introduce a convergence concept in $\hat{\mathcal{C}}$.

Definition 2.2.3 A sequence $\{A_n\}_{n\in\mathbb{N}} \subseteq \hat{\mathcal{C}}$ is said to be \mathcal{C} -convergent to some $A \in \hat{\mathcal{C}}$ (with the same orientation as the sequence) if

$$\liminf_{n \in \mathbb{N}} A_n = \limsup_{n \in \mathbb{N}} A_n = A.$$

Then, the limit A is denoted by $\lim_{n \in \mathbb{N}} A_n$ and we write $A_n \to A$ or $A_n \xrightarrow{\mathcal{C}} A$.

Proposition 2.2.4 (upper vs. outer limit) Let $\{A_n\}_{n\in\mathbb{N}} \subseteq \hat{\mathcal{C}}^*$ and $\{B_n\}_{n\in\mathbb{N}} \subseteq \hat{\mathcal{C}}^\diamond$, where $\hat{\mathcal{C}}^*$ and $\hat{\mathcal{C}}^\diamond$ are considered to be subsets of $\hat{\mathcal{F}}$. Then it holds

$$\underset{n \to \infty}{\text{LIM SUP } A_n} \subseteq \liminf_{n \to \infty} A_n, \qquad \underset{n \to \infty}{\text{LIM SUP } B_n} \subseteq \limsup_{n \to \infty} B_n$$
$$\underset{n \to \infty}{\text{LIM INF } A_n} \subseteq \limsup_{n \to \infty} A_n, \qquad \underset{n \to \infty}{\text{LIM INF } B_n} \subseteq \liminf_{n \to \infty} B_n.$$

Proof. Rely on the definition of the upper and lower limits and (2.1).

The following examples show that (in case of existence) the limit with respect to Painlevé–Kuratowski convergence can be different from the limit with respect to C-convergence. It can be seen that neither C-convergence implies Painlevé–Kuratowski convergence nor vice versa.

Example 2.2.5 Painlevé–Kuratowski convergence does not coincide with C–convergence: (i) Let $\{A_n\}_{n\in\mathbb{N}} \subseteq \hat{C}^*(\mathbb{R}^2)$ be defined by $A_n := \{(y_1, y_2) \in \mathbb{R}^2 | y_2 \leq ny_1\}$. By an easy calculation it can be seen that

$$\left\{ (y_1, y_2) \in \mathbb{R}^2 | \ 0 \le y_1 \right\} = \operatorname{LIM} A_n \ne \lim_{n \to \infty} A_n = \mathbb{R}^2.$$



(ii) Let $\{A_n\}_{n \in \mathbb{N}} \subseteq \hat{\mathcal{C}}^{\star}(\mathbb{R}^2)$ be defined by

$$A_n := \begin{cases} \{(y_1, y_2) \in \mathbb{R}^2 | y_2 \le ny_1\} & \text{if} \quad n \text{ is odd} \\ \mathbb{R}^2 & \text{if} \quad n \text{ is even.} \end{cases}$$

In view of (i), it can be easily seen that $\lim_{n\to\infty} A_n = \mathbb{R}^2$, but $\operatorname{LIM}_{n\to\infty} A_n$ does not exist. In fact, we have $\operatorname{LIM}\operatorname{SUP}_{n\to\infty} A_n = \mathbb{R}^2$, but $\operatorname{LIM}\operatorname{INF}_{n\to\infty} A_n = \{(y_1, y_2) \in \mathbb{R}^2 | 0 \le y_1\}$. (iii) Let $\{A_n\}_{n\in\mathbb{N}} \subseteq \hat{\mathcal{C}}^*(\mathbb{R}^2)$ be defined by

$$A_n := \begin{cases} \{(y_1, y_2) \in \mathbb{R}^2 | y_2 \le ny_1\} & \text{if} \quad n \text{ is odd} \\ \{(y_1, y_2) \in \mathbb{R}^2 | 0 \le y_1\} & \text{if} \quad n \text{ is even.} \end{cases}$$

By (i), it can be easily seen that $\operatorname{LIM}_{n\to\infty} A_n = \{(y_1, y_2) \in \mathbb{R}^2 | \ 0 \le y_1\}$, but $\lim_{n\to\infty} A_n$ does not exist. In fact, we have $\lim_{n\to\infty} A_n = \{(y_1, y_2) \in \mathbb{R}^2 | \ 0 \le y_1\}$, but $\liminf_{n\to\infty} A_n = \mathbb{R}^2$.

Proposition 2.2.6 (Convergence of subsequences) Let $\{A_n\}_{n\in\mathbb{N}} \subseteq \hat{\mathcal{C}}$ be a sequence. Then the following statements hold true:

- (i) $\forall \hat{N} \in \mathcal{N}_{\infty}^{\#}$: $\liminf_{n \in \hat{N}} A_n \ge \liminf_{n \to \infty} A_n,$
- (ii) $\forall \hat{N} \in \mathcal{N}_{\infty}^{\#}$: $\limsup_{n \in \hat{N}} A_n \leq \limsup_{n \to \infty} A_n$,
- (iii) If $A_n \to \overline{A}$, then every subsequence of $\{A_n\}_{n \in \mathbb{N}}$ converges to the same limit.

2.2. C-Convergence

Proof. (i), (ii) In case of $\{A_n\}_{n\in\mathbb{N}} \subseteq \hat{\mathcal{C}}^{\star}$ we have

$$\liminf_{n \in \hat{N}} A_n = \sup_{N \in \mathcal{N}_{\infty}} \inf_{n \in N \cap \hat{N}} A_n \ge \sup_{N \in \mathcal{N}_{\infty}} \inf_{n \in N} A_n = \liminf_{n \to \infty} A_n,$$

$$\limsup_{n \in \hat{N}} \sup_{N \in \mathcal{N}_{\infty}^{\#}, N \subseteq \hat{N}} \inf_{n \in N} A_n \le \sup_{N \in \mathcal{N}_{\infty}^{\#}} \inf_{n \in N} A_n = \limsup_{n \to \infty} A_n$$

The case $\{A_n\}_{n \in \mathbb{N}} \subseteq \hat{\mathcal{C}}^{\diamond}$ is similar.

(iii) For all $N \in \mathcal{N}_{\infty}^{\#}$ it holds

$$\limsup_{n \in \mathbb{N}} A_n \stackrel{\text{(ii)}}{\leq} \limsup_{n \to \infty} A_n = \liminf_{n \to \infty} A_n \stackrel{\text{(i)}}{\leq} \liminf_{n \in \mathbb{N}} A_n$$

Proposition 2.2.2 yields equality.

Proposition 2.2.7 (Monotonicity of upper and lower limits) Let be given two sequences $\{A_n\}_{n\in\mathbb{N}}, \{B_n\}_{n\in\mathbb{N}} \subseteq \hat{\mathcal{C}}$ (with the same orientation). Then,

- (i) $\left(\exists \hat{N} \in \mathcal{N}_{\infty}, \forall n \in \hat{N} : A_n \leq B_n\right) \Rightarrow \liminf_{n \to \infty} A_n \leq \liminf_{n \to \infty} B_n,$
- (ii) $\left(\exists \hat{N} \in \mathcal{N}_{\infty}, \forall n \in \hat{N} : A_n \leq B_n\right) \Rightarrow \limsup_{n \to \infty} A_n \leq \limsup_{n \to \infty} B_n.$

Proof. For example, let $\{A_n\}_{n\in\mathbb{N}}, \{B_n\}_{n\in\mathbb{N}} \subseteq \hat{\mathcal{C}}^*$. Let $\hat{N} \in \mathcal{N}_{\infty}$ such that $A_n \leq B_n$ for all $n \in \hat{N}$. Hence, for all $N \in \mathcal{N}_{\infty}$ with $N \subseteq \hat{N}$ we have $\inf_{n \in N} A_n \leq \inf_{n \in N} B_n$.

(i) Given some $y \in \liminf_{n \to \infty} B_n$, we conclude that

$$\forall N \in \mathcal{N}_{\infty} \text{ with } N \subseteq \hat{N} : \quad y \in \inf_{n \in N} B_n \subseteq \inf_{n \in N} A_n.$$

By the definition of \mathcal{N}_{∞} , we have $y \in \inf_{n \in N} A_n$ even for all $N \in \mathcal{N}_{\infty}$, i.e., $y \in \liminf_{n \to \infty} A_n$. (ii) Let be given some $y \in \limsup_{n \to \infty} B_n$. Note that $N \cap \hat{N} \in \mathcal{N}_{\infty}^{\#}$ for all $N \in \mathcal{N}_{\infty}^{\#}$. Hence

$$\forall N \in \mathcal{N}_{\infty}^{\#} : \quad y \in \inf_{n \in N \cap \hat{N}} B_n \subseteq \inf_{n \in N \cap \hat{N}} A_n \subseteq \inf_{n \in N} A_n$$

i.e., $y \in \limsup_{n \to \infty} A_n$.

Proposition 2.2.8 Let be given a set of sequences $\{A_n^{(i)}\}_{n\in\mathbb{N}} \subseteq \hat{\mathcal{C}}$, where *i* belongs to an arbitrary index set *I*. Then it holds

(i)
$$\liminf_{n \to \infty} \inf_{i \in I} A_n^{(i)} \le \inf_{i \in I} \liminf_{n \to \infty} A_n^{(i)},$$
(ii)
$$\liminf_{n \to \infty} \sup_{i \in I} A_n^{(i)} \ge \sup_{i \in I} \liminf_{n \to \infty} A_n^{(i)},$$
(iv)
$$\limsup_{n \to \infty} \sup_{i \in I} A_n^{(i)} \ge \sup_{i \in I} \limsup_{n \to \infty} A_n^{(i)}.$$

Proof. For all $n \in \mathbb{N}$ and all $i \in I$ it holds $\inf_{i \in I} A_n^{(i)} \leq A_n^{(i)}$. Proposition 2.2.7 yields that $\liminf_{n \to \infty} \inf_{i \in I} A_n^{(i)} \leq \liminf_{n \to \infty} A_n^{(i)}$ for all $i \in I$. Hence (i) is true. The proof of (ii) – (iv) is similar.

Remark 2.2.9 (limit of sum \neq **sum of limits)** In the space $\hat{\mathcal{F}}$, the supremum-oriented version of (i) (i.e., we have to replace "liminf" by "LIMSUP" and "inf" by "SUP") even holds with equality if I is finite. This is not true in the present case, see Example 2.2.10 below. In Kuratowski [47], the union of sets is understood as a "sum". This means, for Painlevé-Kuratowski convergence, but not for C-convergence, the limit of the "sum" of two sequences is equal to the "sum" of its limits. In our framework, however, the sum is understood as the closure of the Minkowski-addition. Nevertheless, also in this sense, limit and sum cannot be interchanged, see Example 2.2.11 below. But, even in the space \mathcal{F} , this is only possible by an extension of the convergence concept to so-called "total convergence" and by additional assumptions, see [68, Exercise 4.29]. It remains open if similar extensions are successful for \mathcal{C} -convergence, too. A detailed discussion about topologies for which the basic operations in $\mathcal{C}(X)$ (namely $(A, B) \to A \oplus B$, $(A, B) \to \inf \{A, B\}$ and $(\alpha, A) \to \alpha \cdot A$) in $\mathcal{C}(X)$ are jointly continuous can be found in [4, Section 4.3]. These operations are jointly continuous with respect to the scalar topology (which is related to the scalar convergence, see Section 2.3), but this topology is admissible (i.e., it extends the topology of the underlying space X) if and only if X is finite dimensional, see [4, Exercise 4.3.1 (c)]. If X is infinite dimensional, the linear topology [4] suffices all requirements. In Theorem 2.3.6 below, we give an equivalent characterization of \mathcal{C} -convergence by convergence of support functions. This shows that we can interchange limit and sum at least in many special cases.

Example 2.2.10 Proposition 2.2.8 (i) – (iv) does not hold with equality (for instance, (i) and (ii) in case of supremum orientation): Let $\{A_n\}_{n\in\mathbb{N}}, \{B_n\}_{n\in\mathbb{N}} \subseteq \hat{\mathcal{C}}^*(\mathbb{R}), A_n = \{n\}, B_n = \{-n\}$. Then $\lim_{n\to\infty} A_n = \lim_{n\to\infty} B_n = \emptyset$, hence $\inf \{\lim_{n\to\infty} A_n, \lim_{n\to\infty} B_n\} = \emptyset$. But, $\inf \{A_n, B_n\} = [-n, n]$ and consequently, $\lim_{n\to\infty} \inf \{A_n, B_n\} = \mathbb{R}$.

Example 2.2.11 The sum of limits is different from the limit of the sum: Let the sequences $\{A_n\}_{n\in\mathbb{N}}$ and $\{B_n\}_{n\in\mathbb{N}}$ as in Example 2.2.10. Then $\emptyset = \lim_{n\to\infty} A_n \oplus \lim_{n\to\infty} B_n \neq \lim_{n\to\infty} (A_n \oplus B_n) = \{0\}.$

Example 2.2.12 The sum of convergent sequences is not convergent: Consider two sequences $\{A_n\}_{n\in\mathbb{N}}, \{B_n\}_{n\in\mathbb{N}} \subseteq \hat{\mathcal{C}}^{\star}(\mathbb{R}^2)$ being defined by

$$A_n := \begin{cases} \{(y_1, y_2) \in \mathbb{R}^2 | \ y_2 = ny_1, y_2 \ge 0\} & \text{if} \quad n \text{ is odd} \\ \{(y_1, y_2) \in \mathbb{R}^2 | \ y_1 = 0, y_2 \ge 0\} & \text{if} \quad n \text{ is even}, \end{cases}$$

$$B_n := \begin{cases} \{(y_1, y_2) \in \mathbb{R}^2 | y_2 = -ny_1, y_2 \le 0\} & \text{if } n \text{ is odd} \\ \{(y_1, y_2) \in \mathbb{R}^2 | y_1 = 0, y_2 \le 0\} & \text{if } n \text{ is even} \end{cases}$$

Then, $\{A_n\}_{n\in\mathbb{N}}$ converges to $\overline{A} = \{(y_1, y_2) \in \mathbb{R}^2 | y_1 = 0, y_2 \ge 0\}$ and $\{B_n\}_{n\in\mathbb{N}}$ converges to $\overline{B} = \{(y_1, y_2) \in \mathbb{R}^2 | y_1 = 0, y_2 \le 0\}$. But, the subsequence $\{A_{2n} \oplus B_{2n}\}_{n\in\mathbb{N}}$ converges to $\{(y_1, y_2) \in \mathbb{R}^2 | y_1 = 0\}$ and $\{A_{2n+1} \oplus B_{2n+1}\}_{n\in\mathbb{N}}$ converges to $\{(y_1, y_2) \in \mathbb{R}^2 | y_1 \ge 0\}$, i.e., $\{A_n \oplus B_n\}_{n\in\mathbb{N}}$ is not convergent.



Proposition 2.2.13 (Sequences of singletons) Let $\{y_n\}_{n \in \mathbb{N}} \subseteq \mathbb{R}^p$ be a sequence.

- (i) If $y_n \to \bar{y}$, then $\{y_n\} \xrightarrow{\mathcal{C}} \{\bar{y}\}$.
- (ii) For $\{\{y_n\}\}_{n\in\mathbb{N}} \subseteq \mathcal{C}^*$, $||y_n|| \to +\infty$ implies that $\limsup_{n\to\infty} \{y_n\} = \emptyset$.
- (iii) Conversely, if $\{y_n\} \xrightarrow{\mathcal{C}} \bar{A} \in \hat{\mathcal{C}}^*$, then $\bar{A} = \emptyset$ or $\bar{A} = \{\bar{y}\}$ for some $\bar{y} \in \mathbb{R}^p$. In the latter case we have $y_n \to \bar{y}$.

Proof. (i) For all $N \in \mathcal{N}_{\infty}^{\#}$ it holds $\bar{y} \in \operatorname{cl} \bigcup_{n \in N} \{y_n\} \subseteq \operatorname{cl} \operatorname{conv} \bigcup_{n \in N} \{y_n\}$. Hence $\bar{y} \in \operatorname{lim} \sup_{n \to \infty} \{y_n\}$. Since $y_n \to \bar{y}$, we can write

$$\forall \varepsilon > 0, \ \exists n_0(\varepsilon) : \ \mathrm{cl\,conv} \ \bigcup_{n \ge n_0(\varepsilon)} \{y_n\} \subseteq \mathbb{B}_{\varepsilon}(\bar{y}).$$

Hence

$$\liminf_{n \to \infty} \{y_n\} = \bigcap_{n_0 \in \mathbb{N}} \operatorname{cl\,conv} \, \bigcup_{n \ge n_0} \{y_n\} \subseteq \bigcap_{\varepsilon > 0} \operatorname{cl\,conv} \, \bigcup_{n \ge n_0(\varepsilon)} \{y_n\} \subseteq \bigcap_{\varepsilon > 0} \mathbb{B}_{\varepsilon}(\bar{y}) = \{\bar{y}\}.$$

This yields $\{\bar{y}\} \ge \limsup_{n \to \infty} \{y_n\} \ge \liminf_{n \to \infty} \{y_n\} \ge \{\bar{y}\}.$

(ii) If $||y_n|| \to +\infty$, there exists $N \in \mathcal{N}_{\infty}^{\#}$ such that $y_n/||y_n|| \xrightarrow{N} y \in \mathbb{R}^p$. Using Proposition 2.2.6 (ii) we obtain $\emptyset = \liminf_{n \in N} \{y_n\} \leq \limsup_{n \in N} \{y_n\} \leq \limsup_{n \to \infty} \{y_n\}$.

(iii) The sequence $\{y_n\}_{n\in\mathbb{N}}$ has a subsequence $\{y_n\}_{n\in\hat{N}}$ $(\hat{N}\in\mathcal{N}_{\infty}^{\#})$ such that either $y_n \xrightarrow{N} \bar{y}$ for some $\bar{y}\in\mathbb{R}^p$ or $\|y_n\| \xrightarrow{\hat{N}} +\infty$. By (i) and (ii) we conclude that, respectively, either $\lim_{n\in\hat{N}}\{y_n\}=\{\bar{y}\}$ or $\limsup_{n\in\mathbb{N}}\{y_n\}=\emptyset$ (without loss of generality we assume $\{\{y_n\}\}_{n\in\mathbb{N}}\subseteq \mathcal{C}^*$). Since $\{\{y_n\}\}_{n\in\mathbb{N}}$ is supposed to be convergent, the result follows from Proposition 2.2.6 (iii).

Sometimes, a slight generalization of part (i) of the previous assertions is useful. Taking into account Example 2.2.11 and Example 2.2.12 this is not a direct consequence of the previous assertion.

Proposition 2.2.14 Let $\{y_n\}_{n\in\mathbb{N}} \subseteq \mathbb{R}^p$ be a sequence and $A \in \hat{\mathcal{C}}$. Then, $y_n \to \bar{y}$ implies $\{\{y_n\}+A\} \xrightarrow{\mathcal{C}} \{\bar{y}+A\}.$

Proof. A direct proof can be given similarly to the proof of Proposition 2.2.13 (i). Furthermore, this statement is a direct consequence of Proposition 2.3.1 below. \Box

In the following we restrict ourselves to supremum-oriented sets. Of course, analogous assertions can be shown for infimum oriented sets if we simultaneously switch over from "liminf" to "limsup", "inf" to "sup" and so on.

The following characterization of the lower limit is useful in order to show further properties of the lower and upper limits.

Proposition 2.2.15 Let $\{A_n\}_{n\in\mathbb{N}} \subseteq \hat{\mathcal{C}}^*$ be a sequence. Then, $y \in \liminf_{n\to\infty} A_n$ if and only if the following assertion holds true:

$$\begin{split} \exists \left\{ (\lambda_{0}^{(m)}, ..., \lambda_{p}^{(m)}), (k_{0}^{(m)}, ..., k_{p}^{(m)}), (z_{0}^{(m)}, ..., z_{p}^{(m)}) \right\}_{m \in \mathbb{N}} \subseteq [0, 1]^{p+1} \times \mathbb{N}^{p+1} \times (\mathbb{R}^{p})^{p+1} : \\ y &= \lim_{m \to \infty} \sum_{j=0}^{p} \lambda_{j}^{(m)} z_{j}^{(m)}, \\ \forall j \in \{0, 1, ..., p\}, \forall m \in \mathbb{N} : \quad z_{j}^{(m)} \in A_{k_{j}^{(m)}}, \\ \forall j \in \{0, 1, ..., p\}, \forall m \in \mathbb{N} : \quad k_{j}^{(m)} \geq m, \\ \forall m \in \mathbb{N} : \quad \sum_{j=0}^{p} \lambda_{j}^{(m)} = 1. \end{split}$$

Proof. (i) Let $y \in A := \liminf_{n \to \infty} A_n$. By definition we have $A = \bigcap_{m \in \mathbb{N}} \operatorname{cl} V_m$, where $V_m := \operatorname{conv} \bigcup_{k \ge m} A_k$. Hence $y \in \operatorname{cl} V_m$ for all $m \in \mathbb{N}$. This yields

$$\forall m \in \mathbb{N}, \ \forall \varepsilon > 0 \ \exists v_{m,\varepsilon} \in V_m : \ \|y - v_{m,\varepsilon}\| < \varepsilon.$$

Choosing $\varepsilon := 1/m$ we obtain a sequence $\{v^{(m)}\}_{m \in \mathbb{N}}$, defined by $v^{(m)} := v_{m,1/m}$, which is convergent to $y \in A$. The sequence $\{v^{(m)}\}_{m \in \mathbb{N}}$ is a sequence of convex combinations $v^{(m)}$ of elements in $Z_m := \bigcup_{k \ge m} A_k$. Since $Z_m \subseteq \mathbb{R}^p$, Carathéodory's theorem allows us to write

$$y = \lim_{m \to \infty} v^{(m)}, \qquad v^{(m)} = \sum_{j=0}^{p} \lambda_j^{(m)} z_j^{(m)},$$

where $\lambda_{j}^{(m)} \in [0,1], z_{j}^{(m)} \in Z_{m} \ (j \in \{0,...,p\})$ and $\sum_{j=0}^{p} \lambda_{j}^{(m)} = 1$. Since $z_{j}^{(m)} \in Z_{m} = \bigcup_{k \ge m} A_{k}$, there exists $k_{j}^{(m)} \ge m$ such that $z_{j}^{(m)} \in A_{k_{j}^{(m)}}$.

2.2. C-Convergence

(ii) The opposite direction can be seen as follows. For arbitrarily given $n \in \mathbb{N}$, we have

$$y = \lim_{m \in \mathbb{N}} \sum_{j=0}^{p} \lambda_{j}^{(m)} z_{j}^{(m)} = \lim_{m \ge n} \sum_{j=0}^{p} \lambda_{j}^{(m)} z_{j}^{(m)}.$$

Hence $y \in \operatorname{cl}\operatorname{conv} \bigcup_{k \ge n} A_k$ for all $n \in \mathbb{N}$. This yields $y \in \liminf_{n \to \infty} A_n$.

Remark 2.2.16 Of course, the previous proposition remains true if the condition is formulated for a sequence in $[0, 1]^{\bar{p}+1} \times \mathbb{N}^{\bar{p}+1} \times (\mathbb{R}^p)^{\bar{p}+1}$, where $\bar{p} \ge p$.

The following assertion is used in the following several times.

Proposition 2.2.17 Let L be a linear subspace of \mathbb{R}^p . The map $T_L : \hat{\mathcal{C}}^* \to \hat{\mathcal{C}}^*$, defined by $T_L(A) = (A \oplus L) \cap L^{\perp}$ has the following properties:

- (i) For $y \in \mathbb{R}^p$, $T_L(\{y\})$ is a singleton set,
- (ii) $\bar{T}_L : \mathbb{R}^p \to \mathbb{R}^p, \ \bar{T}_L(y) \in T_L(\{y\})$ is a well-defined linear continuous operator.
- (iii) $T_L(A) = \operatorname{cl} \overline{T}_L(A)$, where $\overline{T}_L(A) := \bigcup_{y \in A} \overline{T}_L(y)$.
- (iv) If L is the lineality space of $A \subseteq \mathbb{R}^p$, then $A = T_L(A) + L$.
- (v) If L is the lineality space of $A \subseteq \mathbb{R}^p$, then the lineality space of $T_L(A) \subseteq \mathbb{R}^p$ is $\{0\}$, i.e., $T_L(A)$ has a pointed recession cone.

Proof. By [1, Corollary 1.6.1], there is a one-to-one linear transformation of \mathbb{R}^p onto itself, which carries L onto the subspace $\overline{L} := \{y \in \mathbb{R}^p | y_{m+1} = 0, y_{m+2} = 0, ..., y_p = 0\}$, where $m = \dim L$. Hence, without loss of generality we can assume that $L = \overline{L}$. Thus, the assertions (i) – (iv) are elementary. By [1, Corollary 8.3.3], we have $0^+T_L(A) \cap -0^+T_L(A) =$ $0^+(A \oplus L) \cap 0^+(-A \oplus L) \cap L^{\perp} = 0^+A \cap -0^+A \cap L^{\perp} = L \cap L^{\perp} = \{0\}$, which proves (v). \Box

The next two theorems tell us that, under additional assumptions, Painlevé–Kuratowski convergence and C-convergence coincide. For the special case $K = \{0\}$, the statement of the next theorem can be found in [2, Lemma 1.1.9]. A noticeable simplification of part (A) of the proof is due to C. Zălinescu¹.

Theorem 2.2.18 Let $K \subseteq \mathbb{R}^p$ be a nonempty closed convex cone and let $\{A_n\}_{n \in \mathbb{N}} \subseteq \mathcal{C}_K^*$ be a sequence such that $\inf_{n \in \mathbb{N}} A_n \in \mathcal{C}_K^*$. Then,

$$\liminf_{n \to \infty} A_n = \operatorname{cl} \operatorname{conv} \operatorname{LIM}_{n \to \infty} \operatorname{SUP} A_n.$$

¹e-mail conversation, October 2004

Proof. Of course, we have $\liminf_{n\to\infty} A_n \supseteq \operatorname{LIM} \operatorname{SUP}_{n\to\infty} A_n$ and hence

$$\liminf_{n \to \infty} A_n \supseteq \operatorname{cl\,conv}\,\operatorname{LIM\,SUP}_{n \to \infty} A_n. \tag{2.2}$$

In the following we can assume that $\liminf_{n\to\infty} A_n \neq \emptyset$, because otherwise (2.2) yields the desired assertion.

(A) First, we prove the opposite inclusion for the special case that K is pointed, i.e., $K \cap -K = \{0\}$. Let $y \in A := \liminf_{n \to \infty} A_n$ be given. By Proposition 2.2.15 we have

$$\exists \left\{ (\lambda_{0}^{(m)}, ..., \lambda_{p}^{(m)}), (k_{0}^{(m)}, ..., k_{p}^{(m)}), (z_{0}^{(m)}, ..., z_{p}^{(m)}) \right\}_{m \in \mathbb{N}} \subseteq [0, 1]^{p+1} \times \mathbb{N}^{p+1} \times (\mathbb{R}^{p})^{p+1} :$$

$$y = \lim_{m \to \infty} v^{(m)}, \quad v^{(m)} = \sum_{j=0}^{p} \lambda_{j}^{(m)} z_{j}^{(m)},$$

$$\forall j \in \{0, 1, ..., p\}, \forall m \in \mathbb{N} : \quad z_{j}^{(m)} \in A_{k_{j}^{(m)}},$$

$$\forall j \in \{0, 1, ..., p\}, \forall m \in \mathbb{N} : \quad k_{j}^{(m)} \ge m,$$

$$\forall m \in \mathbb{N} : \quad \sum_{j=0}^{p} \lambda_{j}^{(m)} = 1.$$

Without loss of generality we can assume that

$$\forall m \in \mathbb{N}: \qquad \left\|\lambda_0^{(m)} z_0^{(m)}\right\| \le \left\|\lambda_1^{(m)} z_1^{(m)}\right\| \le \dots \le \left\|\lambda_p^{(m)} z_p^{(m)}\right\|,\tag{2.4}$$

and $\|\lambda_p^{(m)} z_p^{(m)}\| \neq 0$ for all $m \in \mathbb{N}$. We can successively switch over to subsequences, again indexed by m, such that

$$\forall j \in \{0, 1, ..., p\}: \quad \lambda_j^{(m)} \xrightarrow{m} \lambda_j \in [0, 1] \qquad \text{and} \qquad \frac{\lambda_j^{(m)} z_j^{(m)}}{\left\|\lambda_p^{(m)} z_p^{(m)}\right\|} \xrightarrow{m} y_j \in \mathbb{R}^p.$$
(2.5)

Assume that the sequence $\{\lambda_j^{(m)} z_j^{(m)}\}_{m \in \mathbb{N}}$ is unbounded for some $j \in \{0, 1, ..., p\}$, then it is unbounded for j = p. Hence, there is a subsequence, again indexed by m, such that $\|\lambda_p^{(m)} z_p^{(m)}\| \xrightarrow{m} +\infty$. It follows $\lambda_j^{(m)} / \|\lambda_p^{(m)} z_p^{(m)}\| \xrightarrow{m} 0$. By the characterization of recession cones of [1, Theorem 8.2], applied to the set $\inf_{n \in \mathbb{N}} A_n$, we deduce that $y_j \in K$ for all $j \in \{0, ..., p\}$. From (2.3) we deduce that

$$\frac{v^{(m)}}{\left\|\lambda_p^{(m)} z_p^{(m)}\right\|} = \sum_{j=0}^p \frac{\lambda_j^{(m)} z_j^{(m)}}{\left\|\lambda_p^{(m)} z_p^{(m)}\right\|}.$$

Taking the limit we obtain $0 = \sum_{j=0}^{p} y_j$. Hence $y_p \in K \cap -K = \{0\}$. This contradicts (2.5), which yields that $||y_p|| = 1$. Hence the sequence $\{\lambda_j^{(m)} z_j^{(m)}\}_{m \in \mathbb{N}}$ is bounded for each $j \in \{0, ..., p\}$. Therefore, we can successively extract subsequences (again denoted by) $\{\lambda_j^{(m)} z_j^{(m)}\}_{m \in \mathbb{N}}$ being convergent to $z_j \in \mathbb{R}^p$.

2.2. C-Convergence

If $\lambda_j = 0$, [1, Theorem 8.2] yields that $z_j \in K$. In case of $\lambda_j \neq 0$, we can assume that $\lambda_j^{(m)} \neq 0$ for all $m \in \mathbb{N}$. Hence we have

$$\left\|z_j^{(m)} - \frac{z_j}{\lambda_j}\right\| \le \left\|z_j^{(m)} - \frac{z_j}{\lambda_j^{(m)}}\right\| + \left\|\frac{z_j}{\lambda_j^{(m)}} - \frac{z_j}{\lambda_j}\right\| \xrightarrow{m} 0,$$

i.e., $\{z_j^{(m)}\}_{m\in\mathbb{N}}$, where $z_j^{(m)} \in \bigcup_{n\geq m} A_n$, converges to $c_j := z_j/\lambda_j \in \mathbb{R}^p$. Thus we find a sequence $\{a_j^{(n)}\}_{n\in\mathbb{N}}$ with $a_j^{(n)} \in A_n$ having a subsequence converging to c_j . This means $c_j = z_j/\lambda_j \in \text{LIM} \operatorname{SUP}_{n\to\infty} A_n$. Let us assume that $\lambda_j \neq 0$ for $j \in \{0, ..., r\}$ and $\lambda_j = 0$ for $j \in \{r+1, ..., p\}$. Setting $k := \sum_{j=r+1}^p z_j$ we obtain

$$y = \sum_{j=0}^{j} \lambda_j c_j + k \in \text{conv LIM} \sup_{n \to \infty} A_n + K \subseteq \text{cl conv LIM} \sup_{n \to \infty} A_n + K.$$

From Proposition A.4 (viii), A.1 (iv), A.4 (vi), A.1 (iii) we obtain

$$y \in \operatorname{cl\,conv\,} \underset{n \to \infty}{\operatorname{LIM\,} \operatorname{SUP}\,} A_n + K \subseteq \operatorname{cl\,conv\,} \underset{n \to \infty}{\operatorname{LIM\,} \operatorname{SUP}\,} (A_n + K) \subseteq \operatorname{cl\,conv\,} \underset{n \to \infty}{\operatorname{LIM\,} \operatorname{SUP}\,} A_n.$$
(2.6)

(B) We now turn to the general case, i.e., the lineality space $L := K \cap -K$ of $\inf_{n \in \mathbb{N}} A_n$ is not necessarily $\{0\}$. Let $T_L : \hat{\mathcal{C}}^* \to \hat{\mathcal{C}}^*$ and $\bar{T}_L : \mathbb{R}^p \to \mathbb{R}^p$ be defined as in Proposition 2.2.17. The following quantities hold true:

$$T_{L}(\inf_{n\to\infty}A_{n}) \stackrel{\text{Pr. 2.2.17 (iii)}}{=} \operatorname{cl}\bar{T}_{L}(\operatorname{cl}\operatorname{conv}\bigcup_{n\in\mathbb{N}}A_{n}) \stackrel{\text{Pr. A.5 (i), A.7 (iii)}}{=} \operatorname{cl}\operatorname{conv}\bigcup_{n\in\mathbb{N}}\bar{T}_{L}(A_{n})$$

$$\stackrel{\text{Pr. 2.2.17 (iii)}}{=} \inf_{n\to\infty}T_{L}(A_{n}),$$

$$T_{L}(\liminf_{n\to\infty}A_{n}) \stackrel{\text{Pr. 2.2.17 (iii)}}{=} \operatorname{cl}\bar{T}_{L}(\liminf_{n\to\infty}A_{n}) \stackrel{\text{Pr. A.5 (ii)}}{\subseteq} \operatorname{cl}\bigcap_{N\in\mathcal{N}_{\infty}}\bar{T}_{L}(\inf_{n\in\mathbb{N}}A_{n})$$

$$\stackrel{\text{Pr. A.2 (i)}}{\subseteq} \bigcap_{N\in\mathcal{N}_{\infty}}T_{L}(\inf_{n\in\mathbb{N}}A_{n}) \stackrel{\text{(2.7)}}{=} \liminf_{n\to\infty}T_{L}(A_{n}).$$

$$(2.8)$$

By (2.7), we have $0^+ \inf_{n\to\infty} T_L(A_n) = T_L(K)$. Proposition 2.2.17 (v) yields that $T_L(K)$ is pointed. Applying part (A) to the sequence $\{T_L(A_n)\}_{n\in\mathbb{N}} \subseteq \mathcal{C}^{\star}_{T_L(K)}$ we obtain

$$T_L(\liminf_{n \to \infty} A_n) \stackrel{(2.8)}{\subseteq} \liminf_{n \to \infty} T_L(A_n) \stackrel{\text{part } (A)}{\subseteq} \operatorname{cl \, conv \, LIM \, SUP } T_L(A_n).$$
(2.9)

Of course, we have $A_n + L = A_n$ for all $n \in \mathbb{N}$. This yields

$$\operatorname{cl\,conv\,} \operatorname{LIM\,}_{n \to \infty} \operatorname{SUP} T_L(A_n) = \operatorname{cl\,conv\,} \operatorname{LIM\,}_{n \to \infty} \operatorname{SUP}(A_n \cap L^{\perp}) \subseteq \operatorname{cl\,conv\,} \operatorname{LIM\,}_{n \to \infty} \operatorname{SUP} A_n, \qquad (2.10)$$

By [1, Corollary 8.3.3], we have $0^+ \liminf_{n \to \infty} A_n = \bigcap_{N \in \mathcal{N}_{\infty}} 0^+ \inf_{n \in N} A_n = K$. Thus,

$$\liminf_{n \to \infty} A_n \stackrel{\text{Pr. 2.2.17 (iv)}}{=} T_L(\liminf_{n \to \infty} A_n) + L \stackrel{(2.9), (2.10)}{\subseteq} \operatorname{cl\,conv\,LIM\,SUP} A_n + L.$$

As in (2.6) we obtain cl conv LIM $\operatorname{SUP}_{n\to\infty} A_n + L \subseteq \operatorname{cl conv} \operatorname{LIM} \operatorname{SUP}_{n\to\infty} A_n$.

An analogous result for the upper and inner limits can be obtained even by weaker assumptions. The result is proven using the previous theorem.

Theorem 2.2.19 Let $\{A_n\}_{n \in \mathbb{N}} \subseteq \hat{\mathcal{C}}^*$ be a sequence such that for all $\bar{N} \in \mathcal{N}_{\infty}^{\#}$ there exists some $\tilde{N} \in \mathcal{N}_{\infty}^{\#}$ with $\tilde{N} \subseteq \bar{N}$ and some nonempty closed convex cone $K \subseteq \mathbb{R}^p$ such that

$$\forall n \in \widetilde{N} : A_n \in \mathcal{C}_K^\star$$
 and $\inf_{n \in \widetilde{N}} A_n \in \mathcal{C}_K^\star$.

Then it holds

$$\limsup_{n \to \infty} A_n = \underset{n \to \infty}{\text{LIM INF}} A_n$$

Proof. Clearly, we have $A := \limsup_{n\to\infty} \supseteq \operatorname{LIM} \operatorname{INF}_{n\to\infty} A_n$. To show the opposite inclusion let $\widehat{N} \in \mathcal{N}_{\infty}^{\#}$ be arbitrarily given. By the cluster point description of outer limits [68, Proposition 4.19] there exists some $\overline{N} \in \mathcal{N}_{\infty}^{\#}$ with $\overline{N} \subseteq \widehat{N}$ such that $\{A_n\}_{n\in\overline{N}}$ is convergent (with respect to Painlevé–Kuratowski convergence). By assumption, there exists $\widetilde{N} \in \mathcal{N}_{\infty}^{\#}$ with $\widetilde{N} \subseteq \overline{N}$ such that $0^+(\inf_{n\in\widetilde{N}}A_n) = K$. Of course, $\{A_n\}_{n\in\widetilde{N}}$ is convergent, too. By Theorem 2.2.18 it follows

$$A = \limsup_{n \to \infty} A_n \overset{\Pr. 2.2.6}{\subseteq} \limsup_{n \in \widetilde{N}} A_n \overset{\Pr. 2.2.2}{\subseteq} \liminf_{n \in \widetilde{N}} A_n = \operatorname{cl\,conv\,LIM\,SUP}_{n \in \widetilde{N}} A_n.$$
(2.11)

The convergence of $\{A_n\}_{n\in\widetilde{N}}$ implies the convexity of the set $\operatorname{LIM}\operatorname{SUP}_{n\in\widetilde{N}}A_n$ (because $\operatorname{LIM}\operatorname{INF}_{n\in\widetilde{N}}A_n$ is a convex set, see [68, Proposition 4.15]). Hence we obtain

$$A \subseteq \operatorname{LIM}_{n \in \widetilde{N}} \operatorname{SUP} A_n \subseteq \operatorname{cl} \bigcup_{n \in \widetilde{N}} A_n \subseteq \operatorname{cl} \bigcup_{n \in \widehat{N}} A_n.$$

Since $\widehat{N} \in \mathcal{N}_{\infty}^{\#}$ was chosen arbitrarily, it follows $A \subseteq \text{LIM INF}_{n \to \infty} A_n$.

Corollary 2.2.20 Let $K \subseteq \mathbb{R}^p$ be a nonempty closed convex cone and let $\{A_n\}_{n\in\mathbb{N}} \subseteq \mathcal{C}_K^*$ a sequence such that $\inf_{n\in\mathbb{N}} A_n \in \mathcal{C}_K^*$. Then, $\{A_n\}_{n\in\mathbb{N}}$ is *C*-convergent if and only if it is Painlevé–Kuratowski–convergent. In case of convergence, we have

$$\lim_{n \to \infty} A_n = \lim_{n \to \infty} A_n$$

Proof. Follows from Theorem 2.2.18 and Theorem 2.2.19.

2.3 Scalar convergence vs. *C*-convergence

Convergence of convex closed sets that is determined by the pointwise convergence of the support functions associated to the sets was investigated by many authors such as Wijsman [82], Salinetti/Wets [70] and Sonntag/Zălinescu [74]. An overview and further references can be found in Beer [4]. Following the article by Sonntag and Zălinescu we call this convergence scalar convergence. A sequence $\{A_n\}_{n\in\mathbb{N}} \subseteq \mathcal{C}$ is said to be *S*-convergent to some $\overline{A} \in \mathcal{C}$ if

$$\forall y^* \in \mathbb{R}^p : \qquad \delta^*(y^*|A_n) \to \delta^*(y^*|\bar{A}).$$

Then we write $A_n \xrightarrow{S} \overline{A}$. In this section, we investigate the relationship between scalar convergence and \mathcal{C} -convergence. We start with a result of Sonntag and Zălinescu [74]. The following extension of this result is due to C. Zălinescu².

Proposition 2.3.1 ([74]) Let $\{A_n\}_{n \in \mathbb{N}} \subseteq \mathcal{C}$. Then $A_n \xrightarrow{S} \overline{A}$ implies $A_n \xrightarrow{\mathcal{C}} \overline{A}$.

Proof. For example, let $\{A_n\}_{n\in\mathbb{N}} \subseteq \mathcal{C}^*$. In [74, Proposition 1] it is shown that $A_n \xrightarrow{S} \bar{A}$ implies that $\bar{A} = \liminf_{n\to\infty} A_n$.

To show that $\bar{A} = \limsup_{n \to \infty} A_n$, let $N \in \mathcal{N}_{\infty}^{\#}$. Then, $\bar{A} = S - \lim_{n \in N} A_n$, and so $\bar{A} = \liminf_{n \in N} A_n \ge \inf_{n \in N} A_n$ for all $N \in \mathcal{N}_{\infty}^{\#}$. Hence we have $\bar{A} = \liminf_{n \in N} A_n \ge \sup_{N \in \mathcal{N}_{\infty}^{\#}} \inf_{n \in N} A_n = \limsup_{n \to \infty} A_n$. It follows $\bar{A} = \liminf_{n \to \infty} A_n \le \limsup_{n \to \infty} A_n \le \bar{A}$, i.e., $A_n \xrightarrow{\mathcal{C}} \bar{A}$.

We next want to investigate the opposite inclusion. Assuming that C-convergence implies scalar convergence we can deduce that the sum of two C-convergent sequences is convergent as well. This, however, contradicts Example 2.2.12. Indeed, the sequence $\{A_n\}_{n\in\mathbb{N}} \subseteq C$ in Example 2.2.12 C-converges to $\overline{A} := \{(y_1, y_2) \in \mathbb{R}^2 | y_1 = 0, y_2 \ge 0\}$, but for $y^* = (1, 0)$ the sequence $\{\delta^*(y^*|A_n)\}_{n\in\mathbb{N}}$ is not convergent. However, we can show that C-convergence implies the pointwise convergence of the support function, except at relative boundary points of the domain of the support function of the C-limit. Moreover, C-convergence is equivalent to this property. We start with some auxiliary assertions.

Proposition 2.3.2 Let $K \subseteq \mathbb{R}^p$ be a nonempty pointed closed convex cone. Then, for $y^* \in$ ri K° and $k \in K \setminus \{0\}$ it holds $\langle y^*, k \rangle < 0$.

Proof. We have int $K^{\circ} \neq \emptyset$ and consequently $y^* \in \operatorname{int} K^{\circ}$. Hence there exists some $\varepsilon > 0$ such that $y^* + \varepsilon k \in K^{\circ}$. This yields $\langle y^* + \varepsilon k, k \rangle = \langle y^*, k \rangle + \varepsilon ||k||^2 \leq 0$, hence $\langle y^*, k \rangle < 0$. \Box

Proposition 2.3.3 Let $\{A_n\}_{n\in\mathbb{N}} \subseteq \hat{\mathcal{C}}^{\star}$ such that $\liminf_{n\to\infty} A_n \neq \emptyset$ and let $\{y_n\}_{n\in\mathbb{N}} \subseteq \mathbb{R}^p \setminus \{0\}$ be a sequence such that $y_n \in A_n$ for all $n \in \mathbb{N}$, $||y_n|| \to \infty$ and $y_n/||y_n|| \to k$. Then, $k \in 0^+ \liminf_{n\to\infty} A_n$.

Proof. We show that $y + \mu k \in \liminf_{n \to \infty} A_n$ for arbitrarily given $y \in \liminf_{n \to \infty} A_n$ and $\mu > 0$. Using the characterization of Proposition 2.2.15, we obtain

$$y + \mu k = \lim_{m \to \infty} \sum_{j=0}^{p} \lambda_j^{(m)} z_j^{(m)} + \lim_{m \to \infty} \mu \frac{y_m}{\|y_m\|}.$$

Setting $\lambda_{p+1}^{(m)} := \frac{\mu}{\|y_m\|}, \, z_{p+1}^{(m)} := y_m, \, k_{p+1}^{(m)} := m$ and

$$\forall j \in \{0, ..., p+1\}: \qquad \tilde{\lambda}_j^{(m)} := \frac{\lambda_j^{(m)}}{(1+\lambda_{p+1}^{(m)})}$$

²e-mail conversation, October 2004

we obtain

$$y + \mu k = \lim_{m \to \infty} (1 + \lambda_{p+1}^{(m)}) \sum_{j=0}^{p+1} \tilde{\lambda}_j^{(m)} z_j^{(m)} = \lim_{m \to \infty} \sum_{j=0}^{p+1} \tilde{\lambda}_j^{(m)} z_j^{(m)}.$$

By Proposition 2.2.15 and taking into account Remark 2.2.16 we obtain that $y + \mu k \in \liminf_{n \to \infty} A_n$.

Proposition 2.3.4 Let $\{A_n\}_{n\in\mathbb{N}} \subseteq \hat{\mathcal{C}}^*$ such that $\liminf_{n\to\infty} A_n \neq \emptyset$. Denoting by L the lineality space of $\liminf_{n\to\infty} A_n$, we define $T_L : \hat{\mathcal{C}}^* \to \hat{\mathcal{C}}^*$ as in Proposition 2.2.17. Then,

$$T_L(\liminf_{n \to \infty} A_n) = \liminf_{n \to \infty} T_L(A_n)$$

Proof. Set $\bar{A} := \liminf_{n\to\infty} A_n$. As in (2.8) we obtain $T_L(\bar{A}) \ge \liminf_{n\to\infty} T_L(A_n)$. It follows that $T_L(\bar{A}) \ge \liminf_{n\to\infty} (A_n \oplus L)$. For all $N \in \mathcal{N}_\infty$ we have $\inf_{n\in N} A_n \le \bar{A}$ and hence $0^+(\inf_{n\in N} A_n) \supseteq 0^+ \bar{A} \supseteq L$. With the aid of Proposition 1.2.4 we obtain $\inf_{n\in N} (A_n \oplus L) = (\inf_{n\in N} A_n) \oplus L = \inf_{n\in N} A_n$ for all $N \in \mathcal{N}_\infty$. Hence $\liminf_{n\to\infty} (A_n \oplus L) = \liminf_{n\to\infty} A_n = \bar{A}$. It follows that $T_L(\bar{A}) \ge \liminf_{n\to\infty} T_L(A_n) \ge \bar{A}$. Of course, $\liminf_{n\to\infty} T_L(A_n) \ge L^{\perp}$ and so $T_L(\bar{A}) \ge \liminf_{n\to\infty} T_L(A_n) \ge \bar{A} \cap L^{\perp} = T_L(\bar{A})$.

Lemma 2.3.5 Let $\{A_n\}_{n \in \mathbb{N}} \subseteq \hat{\mathcal{C}}^{\star}$ such that $\liminf_{n \to \infty} A_n \neq \emptyset$. Then,

$$\forall y^* \in \operatorname{ri} \left(0^+ \liminf_{n \to \infty} A_n \right)^{\circ} : \qquad \liminf_{n \to \infty} -\delta^*(y^* | A_n) = -\delta^*(y^* | \liminf_{n \to \infty} A_n).$$

Proof. Set $\overline{A} := \liminf_{n \to \infty} A_n$. From Proposition 1.3.7 we easily obtain

$$\forall y^* \in \mathbb{R}^p: \qquad \liminf_{n \to \infty} -\delta^*(y^* | A_n) \le -\delta^*(y^* | \bar{A}).$$

It remains to show that $\limsup_{n\to\infty} \delta^*(y^*|A_n) \leq \delta^*(y^*|\bar{A})$ for all $y^* \in \operatorname{ri}(0^+\bar{A})^\circ$.

(A) We first prove the case that $0^+\bar{A}$ is pointed. Assume the assertion is not true. This means there exists some $y^* \in \operatorname{ri}(0^+\bar{A})^\circ$ such that $\limsup_{n\to\infty} \delta^*(y^*|A_n) > \delta^*(y^*|\bar{A})$. Hence, there is some $\varepsilon > 0$ and some $N \in \mathcal{N}_{\infty}^{\#}$ such that $\delta^*(y^*|A_n) > \delta^*(y^*|\bar{A}) + \varepsilon$ for all $n \in N$. It follows

$$\forall n \in N, \quad \exists y_n \in A_n : \quad \langle y^*, y_n \rangle > \delta^*(y^* | \bar{A}) + \frac{\varepsilon}{2}.$$
(2.12)

We distinguish between two cases.

(i) If there exists some $\tilde{N} \in \mathcal{N}_{\infty}^{\#}$ with $\tilde{N} \subseteq N$ such that $y_n \to \bar{y} \in \mathbb{R}^p$ we conclude that $\bar{y} \in \text{LIM} \operatorname{SUP}_{n \to \infty} A_n \subseteq \liminf_{n \to \infty} A_n = \bar{A}$. This contradicts (2.12).

(ii) Otherwise there is some $\tilde{N} \in \mathcal{N}_{\infty}^{\#}$ with $\tilde{N} \subseteq N$ such that $||y_n|| \neq 0$, $||y_n|| \to \infty$ and $|y_n/||y_n|| \to k \in \mathbb{R}^p$. From Proposition 2.3.3 we deduce that $k \in 0^+\bar{A}$. Proposition 2.3.2 yields $\langle y^*, k \rangle < 0$. But, from (2.12) we deduce that $\langle y^*, k \rangle \geq 0$, a contradiction.

(B) If $0^+\bar{A}$ is not pointed, consider its lineality space $L := 0^+\bar{A} \cap (-0^+\bar{A}) \neq \{0\}$. Let $T_L : \hat{\mathcal{C}}^* \to \hat{\mathcal{C}}^*$ be defined as in Proposition 2.2.17. Proposition 2.3.4 yields that $T_L(\bar{A}) =$
$\liminf_{n\to\infty} T_L(A_n). \quad \text{From [1, Corollary 8.3.3] and [1, Corollary 16.4.2] we deduce that$ $<math display="block"> (0^+T_L(\bar{A}))^\circ = (0^+\bar{A})^\circ \oplus L. \quad \text{By [1, Theorem 6.3] and [1, Corollary 6.6.2] we conclude that ri <math>(0^+T_L(\bar{A}))^\circ = \text{ri} (0^+\bar{A})^\circ + L.$ This implies that ri $(0^+T_L(\bar{A}))^\circ \supseteq \text{ri} (0^+\bar{A})^\circ.$ Moreover, from $L \subseteq 0^+\bar{A}$ we conclude that $L^\perp \supseteq (0^+\bar{A})^\circ.$ For arbitrary $A \in \mathcal{C}$ it is true that $\delta^*(y^*|T_L(A)) = \delta^*(y^*|A)$ for all $y^* \in L^\perp.$ Since $0^+T_L(\bar{A})$ is pointed, for all $y^* \in \text{ri} (0^+\bar{A})^\circ$ $(\subseteq \text{ri} (0^+T_L(\bar{A}))^\circ \cap L^\perp)$ we have

$$\limsup_{n \to \infty} \delta^*(y^*|A_n) = \limsup_{n \to \infty} \delta^*(y^*|T_L(A_n)) \stackrel{\text{part (A)}}{\leq} \delta^*(y^*|T_L(\bar{A})) = \delta^*(y^*|\bar{A}).$$

. . .

This completes the proof.

It follows the main result of this section, an equivalent characterization of C-convergence. For $A \in C$ we denote by $\operatorname{rb} A := A \setminus \operatorname{ri} A$ the relative boundary of A.

Theorem 2.3.6 If $\{A_n\}_{n \in \mathbb{N}} \subseteq \hat{\mathcal{C}}$ and $\bar{A} \in \mathcal{C}$, the following statements are equivalent:

(i) $A_n \xrightarrow{\mathcal{C}} \bar{A}$,

(ii) $\forall y^* \in \mathbb{R}^p \setminus \operatorname{rb}(0^+ \overline{A})^\circ : \qquad \delta^*(y^* | A_n) \to \delta^*(y^* | \overline{A}).$

Proof. We adopt our notation to the case $\{A_n\}_{n\in\mathbb{N}} \subseteq \hat{\mathcal{C}}^*$. (i) \Rightarrow (ii). By Proposition 1.3.7 we easily deduce that

$$\forall y^* \in \mathbb{R}^p: \qquad \delta^*(y^*|\bar{A}) = \delta^*(y^*|\limsup_{n \to \infty} A_n) \le \liminf_{n \to \infty} \delta^*(y^*|A_n). \tag{2.13}$$

By Lemma 1.3.1 (i) we obtain $\delta^*(y^*|A_n) \to \delta^*(y^*|\bar{A}) = +\infty$ for all $y^* \in \mathbb{R}^p \setminus (0^+\bar{A})^\circ$. It remains to show that $\delta^*(y^*|A_n) \to \delta^*(y^*|\bar{A})$ for all $y^* \in \operatorname{ri}(0^+\bar{A})^\circ$. This follows from Lemma 2.3.5 and (2.13).

(ii) \Rightarrow (i). Let $B_n := \inf_{k \ge n} A_k$. Then, $\{B_n\}_{n \in \mathbb{N}} \subseteq \hat{\mathcal{C}}^*$ and $\{-\delta^*(\cdot | B_n)\}_{n \in \mathbb{N}}$ are increasing. It follows

$$\forall y^* \in \mathbb{R}^p \setminus \operatorname{rb} (0^+ \bar{A})^\circ : -\delta^*(y^* | \bar{A}) = \liminf_{n \to \infty} -\delta^*(y^* | A_n)$$

$$= \sup_{n \in \mathbb{N}} \inf_{k \ge n} -\delta^*(y^* | A_n)$$

$$\stackrel{\operatorname{Pr. 1.3.7}}{=} \sup_{n \in \mathbb{N}} -\delta^*(y^* | B_n) = \lim_{n \to \infty} -\delta^*(y^* | B_n).$$

$$(2.14)$$

Hence, for all $n \in \mathbb{N}$ we obtain

$$\forall y^* \in \mathbb{R}^p \setminus \operatorname{rb}(0^+ \bar{A})^\circ : \qquad \delta^*(y^* | \bar{A}) \le \delta^*(y^* | B_n).$$
(2.15)

Since $B_{n+1} \subseteq B_n$ and $B_n \neq \emptyset$ for all $n \in \mathbb{N}$, there exists some $n_0 \in \mathbb{N}$ such that the lineality space of B_n is constant for all $n \geq n_0$. Without loss of generality we can set $L := 0^+ B_n \cap -0^+ B_n$ for all $n \in \mathbb{N}$. We next show that $0^+ \overline{A} \cap -0^+ \overline{A} = L$. For $y^* \in \mathbb{R}^p \setminus L^{\perp}$ we have $\delta^*(y^*|B_n) = +\infty$ for all $n \in \mathbb{N}$. By (2.14) we have either $\delta^*(y^*|\overline{A}) = +\infty$ or

 $y^* \in \operatorname{rb}(0^+\bar{A})^\circ$. Taking into account Lemma 1.3.1 (i) we obtain $\operatorname{ri}(0^+\bar{A})^\circ \subseteq L^{\perp}$, hence $L \subseteq 0^+A$. It follows that $L \subseteq 0^+\bar{A} \cap -0^+\bar{A}$. Further, from (2.15) and Lemma 1.3.1 (i) we deduce that $\operatorname{ri}(0^+B_n)^\circ \subseteq (0^+\bar{A})^\circ$ and hence $0^+\bar{A} \subseteq 0^+B_n$ for all $n \in \mathbb{N}$. This yields that $0^+\bar{A} \cap -0^+\bar{A} \subseteq L$.

Proceeding as in the proof of Lemma 2.3.5 we have $(0^+T_L(\bar{A}))^\circ = (0^+\bar{A})^\circ \oplus L$. Since $(0^+\bar{A})^\circ \subseteq L^{\perp}$, it follows $(0^+T_L(\bar{A}))^\circ \cap L^{\perp} = (0^+\bar{A})^\circ$. From [1, Theorem 6.5] we conclude that ri $(0^+T_L(\bar{A}))^\circ \cap L^{\perp} = \operatorname{ri}(0^+\bar{A})^\circ$. Moreover, we have $\delta^*(y^*|\bar{A}) = \delta^*(y^*|T_L(\bar{A}))$ for all $y^* \in L^{\perp}$. For all $y^* \in \mathbb{R}^p$ and all $l \in L$ it holds $\delta^*(y^*|T_L(\bar{A})) = \delta^*(y^*+l \mid T_L(\bar{A}))$. Analogous assertions are valid for B_n (instead of \bar{A}). Together with (2.15) we obtain $\delta^*(y^*|T_L(\bar{A})) \leq \delta^*(y^*|T_L(B_n))$ for all $y^* \in \mathbb{R}^p \setminus \operatorname{rb}(0^+T_L(\bar{A}))^\circ$. Lemma 1.3.1 (iv) yields that $T_L(\bar{A}) \subseteq T_L(B_n)$ for all $n \in \mathbb{N}$. With the aid of Proposition 2.2.17 (iv) we conclude that $\bar{A} \subseteq B_n$ for all $n \in \mathbb{N}$ and so $\bar{A} \subseteq \bigcap_{n \in \mathbb{N}} B_n = \liminf_{n \to \infty} A_n$.

On the other hand, for all $y^* \in \mathbb{R}^p \setminus \operatorname{rb}(0^+ \overline{A})^\circ$ we have

$$-\delta^*(y^*|\liminf_{n\to\infty}A_n) = -\delta^*(y^*|\sup_{n\in\mathbb{N}}B_n) \stackrel{\text{Pr. 1.3.7}}{\geq} \sup_{n\in\mathbb{N}} -\delta^*(y^*|B_n) \stackrel{\text{(2.14)}}{=} -\delta^*(y^*|\bar{A}).$$

From Lemma 1.3.1 (iii) we deduce that $\liminf_{n\to\infty} A_n \subseteq \bar{A}$. Hence $\liminf_{n\to\infty} A_n = \bar{A}$. As in the second part of the proof of Proposition 2.3.1 we obtain $\bar{A} = \limsup_{n\to\infty} A_n$ and hence $A_n \xrightarrow{\mathcal{C}} \bar{A}$.

Chapter 3

Functions with values in $\hat{\mathcal{C}}$

In this chapter, we investigate $\hat{\mathcal{C}}$ -valued functions, i.e., functions with values in the space $\hat{\mathcal{C}}$ of closed convex subsets of \mathbb{R}^p . In our optimization problems, based on set relations, the objective functions are of this type. Moreover, $\hat{\mathcal{C}}$ -valued functions frequently occur in other fields of Optimization and Nonlinear Analysis. For instance, the sublevel map of a (epi-)closed quasi-convex vector-valued function (compare Luc [57, Theorem 5.8 and Proposition 6.3 (i)]) and many kinds of subdifferential maps $\partial f(\cdot)$ (e.g. the classical one [1] and that of Clarke [18]) are $\hat{\mathcal{C}}$ -valued functions.

The aim of our investigations is to generalize some important concepts and assertions from Convex Analysis to the case of \hat{C} -valued functions (instead of extended real-valued functions). In doing so, we try to point out the similar structures. It turns out that certain well-known concepts for set-valued maps, such as semi-continuity notions, are not appropriate in this context. Therefore, we introduce some new concepts. Our semi-continuity notions are compared with the usual notions of upper, lower, outer and inner semi-continuity of a set-valued map.

This chapter is organized as follows. In the first section, we summarize some well-known semi-continuity notions of set-valued maps. We point out that these concepts are adapted to the framework of $\hat{\mathcal{F}}$ -valued functions. In the next section, we introduce a semi-continuity concept for $\hat{\mathcal{C}}$ -valued functions. This concept is based on the results of Chapter 2 and seems to be new. In Section 3.3, we present a sufficient condition, called *local boundedness*, such that semi-continuity for $\hat{\mathcal{C}}$ -valued functions (see Section 3.2) coincides with the classical concepts based on the outer and inner limits (see Section 3.1). Section 3.4 is devoted to convex functions. It can be seen that many assertions, which are known for extended real-valued convex functions, can be generalized to the context of this work. So we introduce conjugates for $\hat{\mathcal{C}}$ -valued functions and, finally, in Section 3.6 we prove a biconjugation theorem, which is completely analogous to the classical variant.

3.1 Semi-continuity of $\hat{\mathcal{F}}$ -valued functions

In this section, we summarize some well-known facts on semi-continuity of set-valued maps. Our main reference is the book by Rockafellar and Wets [68]. Even though in [68] arbitrary set-valued maps $f : \mathbb{R}^n \rightrightarrows \mathbb{R}^p$ are considered, we formulate these results only for $\hat{\mathcal{F}}$ -valued functions, i.e., we suppose the closedness of the values. This is no loss of generality, because semi-continuity properties are not influenced by the closure operation to the values of the map f, see [68, Proposition 4.4, Definition 5.4].

Although in [68] the terms *outer* and *inner limit* (instead of upper and lower limit) are used, it might be beneficial to identify the relation \subseteq by "less or equal". This is suggestive by the usage of the notation "limsup" and "liminf" in [68] and, furthermore, we use concepts like supremum, infimum, epigraph and hypograph, which are based on this identification. Let us recall the concepts of outer and inner limits as well as some of their basic properties. As in Section 2.1, we use capital letters in order to distinguish these concepts from the new concepts to be introduced in the next section.

Throughout this section (and this chapter) let $X = \mathbb{R}^n$, although many assertions are also valid in a more general context. The notation $\bigcup_{x_n \to \bar{x}}$, for instance, means the union over all sequences converging to \bar{x} .

Definition 3.1.1 ([68]) Let $f: X \to \hat{\mathcal{F}}$. The outer limit of f at $\bar{x} \in X$ is defined by

$$\underset{x \to \bar{x}}{\text{LIM}} \sup f(x) := \bigcup_{x_n \to \bar{x}} \underset{n \to \infty}{\text{LIM}} \sup f(x_n),$$

and the inner limit of f at $\bar{x} \in X$ is defined by

$$\underset{x \to \bar{x}}{\text{LIM}} \underset{x \to \bar{x}}{\text{INF}} f(x) := \bigcap_{x_n \to \bar{x}} \underset{n \to \infty}{\text{LIM}} \underset{n \to \infty}{\text{INF}} f(x_n).$$

Note that the outer limit (and obviously also the inner limit) is always a closed subset of \mathbb{R}^p , i.e., it belongs to $\hat{\mathcal{F}}$, see [68, Proposition 4.4]. Together with (2.1) this yields the following description of outer and inner limits by the supremum and infimum in $(\hat{\mathcal{F}}, \subseteq)$.

$$\operatorname{LIM}_{x \to \bar{x}} \operatorname{SUP} f(x) = \operatorname{SUP}_{x_n \to \bar{x}} \operatorname{INF}_{N \in \mathcal{N}_{\infty}} \operatorname{SUP}_{n \in N} f(x_n),$$

$$\operatorname{LIM}_{x \to \bar{x}} \operatorname{INF}_{x_n \to \bar{x}} \operatorname{INF}_{N \in \mathcal{N}_{\infty}^{\#}} \operatorname{SUP}_{n \in N} f(x_n).$$
(3.1)

Definition 3.1.2 ([68]) A function $f: X \to \hat{\mathcal{F}}$ is said to be outer semi-continuous (osc) at $\bar{x} \in X$ if

$$f(\bar{x}) \supseteq \operatorname{LIM}_{x \to \bar{x}} \operatorname{SUP} f(x)$$

A function $f: X \to \hat{\mathcal{F}}$ is said to be inner semi-continuous (isc) at $\bar{x} \in X$ if

$$f(\bar{x}) \subseteq \operatorname{LIM} \operatorname{INF} f(x).$$

If f is osc (isc) at every $\bar{x} \in X$ we simply say f is osc (isc).

By the identification of \subseteq with "less or equal" it is clear that outer and inner semi-continuity can be considered as generalizations of upper and lower semi-continuity of extended realvalued functions to the set-valued case. As noticed in [68], Choquet [17] already used the term *upper semi-continuous* instead of *outer semi-continuous*. But, what is often called *upper semi-continuous* in the literature differs from this concept, see [68, p.193 and 5.7(b)]. The inner semi-continuity, however, is mostly called lower semi-continuity in the literature. It is quite natural to ask for an equivalent description of semi-continuity by the "epigraph" and "hypograph" as it is well-known for extended real-valued functions.

Definition 3.1.3 The epigraph of a function $f: X \to \hat{\mathcal{F}}$ is defined to be the set

$$\operatorname{epi} f := \left\{ (x, A) \in X \times \hat{\mathcal{F}} | A \supseteq f(x) \right\},$$

and the hypograph of a function $f: X \to \hat{\mathcal{F}}$ is the set

hyp
$$f := \left\{ (x, A) \in X \times \hat{\mathcal{F}} | A \subseteq f(x) \right\}.$$

Note that, for all $x \in X$, we have $(x, \emptyset) \in \text{hyp } f$ and $(x, \mathbb{R}^p) \in \text{epi } f$. For a characterization of semi-continuity we need to know what is meant by closedness of the epigraph and hypograph.

Definition 3.1.4 A subset $\mathcal{A} \subseteq X \times \hat{\mathcal{F}}$ is closed if for every sequence $\{(x_n, A_n)\}_{n \in \mathbb{N}} \subseteq \mathcal{A}$ with $x_n \to \bar{x} \in X$ and $A_n \to \bar{A} \in \hat{\mathcal{F}}$ (with respect to Painlevé–Kuratowski convergence) it is true that $(\bar{x}, \bar{A}) \in \mathcal{A}$. The closure of a set $\mathcal{A} \subseteq X \times \hat{\mathcal{F}}$ is defined to be the set of all $(\bar{x}, \bar{A}) \in X \times \hat{\mathcal{F}}$ such that there is a sequence $\{(x_n, A_n)\}_{n \in \mathbb{N}} \subseteq \mathcal{A}$ with $x_n \to \bar{x}$ and $A_n \to \bar{A}$. The closure of \mathcal{A} is denoted by cl \mathcal{A} .

It follows the characterization of outer semi–continuity by the hypograph (and the graph) of the function.

Proposition 3.1.5 For all functions $f: X \to \hat{\mathcal{F}}$ the following statements are equivalent:

- (i) hyp f is closed,
- (ii) f is osc,
- (iii) gr $f \subseteq X \times \mathbb{R}^p$ is closed.

Proof. (i) \Leftrightarrow (ii). Follows from [68, Exercise 5.6 (c)].

(ii) \Leftrightarrow (iii). See [68, Theorem 5.7 (a)].

Likewise, inner semi-continuity can be characterized by the closedness of the epigraph. Note that the description by the graph fails in this case, i.e., a function $f: X \to \hat{\mathcal{F}}$ that is isc does not necessarily have a closed graph, see [68, Fig. 5–3. (b)].

Proposition 3.1.6 For all functions $f: X \to \hat{\mathcal{F}}$ the following statements are equivalent:

- (i) epi f is closed,
- (ii) f is isc.

Proof. Follows from [68, Exercise 5.6 (d)].

We next consider the outer semi-continuous hull of a $\hat{\mathcal{F}}$ -valued function. We observe that the outer semi-continuous hull is defined analogously to the upper semi-continuous hull of extended real-valued functions and it has comparable properties.

Definition 3.1.7 Let $f : X \to \hat{\mathcal{F}}$. The outer semi-continuous hull of f is the function $(\operatorname{osc} f) : X \to \hat{\mathcal{F}}$ defined by

$$(\operatorname{osc} f)(x) := \operatorname{LIM}_{x' \to x} \operatorname{SUP} f(x').$$

Let us collect some basic properties of the outer semi-continuous hull.

Proposition 3.1.8 Let $f: X \to \hat{\mathcal{F}}$. Then it holds

- (i) gr $(\operatorname{osc} f) = \operatorname{cl} (\operatorname{gr} f),$
- (ii) hyp $(\operatorname{osc} f) \supseteq \operatorname{cl} (\operatorname{hyp} f)$,
- (iii) $(\operatorname{osc} f)$ is osc,
- (iv) $\forall x \in X$: $(\operatorname{osc} f)(x) \supseteq f(x)$,
- (v) f is osc at $\bar{x} \in X \iff (\operatorname{osc} f)(\bar{x}) = f(\bar{x}),$
- (vi) gr f convex \Rightarrow gr (osc f) convex.

Proof. (i) See [68, page 154, 5(2) and 5(3)].

(ii) Let $(\bar{x}, \bar{A}) \in \operatorname{cl}(\operatorname{hyp} f)$. Then, there exist $\{x_n\}_{n \in \mathbb{N}} \subseteq X$ and $\{A_n\}_{n \in \mathbb{N}} \subseteq \hat{\mathcal{F}}$ such that $\bar{x} = \lim_{n \to \infty} x_n$, $\bar{A} = \operatorname{LIM}_{n \to \infty} A_n$ and $A_n \subseteq f(x_n)$ for all $n \in \mathbb{N}$. Hence,

$$(\operatorname{osc} f)(\bar{x}) = \operatorname{LIM}_{x \to \bar{x}} \operatorname{SUP} f(x) \supseteq \operatorname{LIM}_{n \to \infty} \operatorname{SUP} f(x_n)$$
$$\supseteq \operatorname{LIM}_{n \to \infty} \operatorname{SUP} A_n = \operatorname{LIM}_{n \to \infty} A_n = \bar{A},$$

i.e., $(\bar{x}, \bar{A}) \in \text{hyp}(\text{osc } f)$.

- (iii) By (i), $\operatorname{gr}(\operatorname{osc} f)$ is closed. Hence, $(\operatorname{osc} f)$ is osc, by Proposition 3.1.5.
- (iv) Choosing the special sequence $x_n \equiv x$, we obtain

$$(\operatorname{osc} f)(x) = \operatorname{LIM}_{x' \to x} \operatorname{SUP}_{x} f(x') \supseteq \operatorname{LIM}_{n \to \infty} \operatorname{SUP}_{x' \to x} f(x_n) = \operatorname{LIM}_{n \to \infty} \operatorname{SUP}_{x' \to x} f(x) = f(x).$$

(v) By definition, f is osc at \bar{x} if and only if $f(\bar{x}) \supseteq (\operatorname{osc} f)(\bar{x})$. By (iv), this equivalent to $f(\bar{x}) = (\operatorname{osc} f)(\bar{x})$.

(vi) Since gr f is convex, cl(gr f) is convex, too. Hence, the convexity of gr(osc f) follows from (i).

The next example shows that the opposite inclusion in assertion (ii) of the previous proposition does not hold true, in general.

Example 3.1.9 Let $f : \mathbb{R} \to \hat{\mathcal{F}}(\mathbb{R})$ be defined by

$$f(x) = \begin{cases} \{-1\} & \text{if } x < 0\\ \{1\} & \text{if } x > 0\\ \emptyset & \text{if } x = 0. \end{cases}$$

Then, $(0, \{-1, 1\})$ belongs to hyp $(\operatorname{osc} f)$ but it does not belong to $\operatorname{cl}(\operatorname{hyp} f)$.

Remark 3.1.10 As noticed in [68], an analogous definition of the inner semi-continuous hull, namely by $(\text{isc } f)(x) := \text{LIM INF}_{x' \to x} f(x')$, is not constructive in the sense that (isc f) is not necessarily isc. In the framework of $\hat{\mathcal{C}}$ -valued functions we will have similar problems. An example is given there.

Remark 3.1.11 In Proposition 3.1.8 (vi) we consider the convexity of gr f. Recall that gr f is convex if and only if for all $\lambda \in [0, 1]$ and all $x_1, x_2 \in X$ is holds $f(\lambda x_1 + (1 - \lambda)x_2) \supseteq \lambda f(x_1) \oplus (1 - \lambda)f(x_2)$. Having in mind that \subseteq has the meaning of "less or equal", this can be interpreted as concavity of f although in the literature this property is often called convexity of f. In view of the analogy to extended real-valued functions, Proposition 3.1.8 (vi) should be expressed as follows: f concave \Rightarrow (osc f) concave.

3.2 Semi-continuity

Based on the considerations of Section 2.2 we introduce the concepts of upper and lower limits for functions with values in $\hat{\mathcal{C}}$. We first show that the concept of outer limit is not appropriate to define semi-continuity in the framework of $\hat{\mathcal{C}}$ -valued functions. As in the previous section, we set $X = \mathbb{R}^n$, although many assertions are also valid in a more general context.

Example 3.2.1 Let $f : \mathbb{R} \to \mathcal{C}(\mathbb{R})$,

$$f(x) := \begin{cases} \{-1\} & \text{if } x < 0\\ \{0\} & \text{if } x = 0\\ \{1\} & \text{if } x > 0. \end{cases}$$

Then the outer semi–continuous hull of f, namely

$$(\operatorname{osc} f)(x) = \begin{cases} \{-1\} & \text{if } x < 0\\ \{-1, 0, 1\} & \text{if } x = 0\\ \{1\} & \text{if } x > 0, \end{cases}$$

is not convex-valued.

This might suggest to redefine the outer semi-continuous hull as follows:

$$(\widetilde{\operatorname{osc}} f)(x) := \operatorname{cl\,conv\,LIM\,SUP}_{x' \to x} f(x')$$

However, $(\widetilde{\operatorname{osc}} f)$ has not necessarily a closed graph (and hence it is not osc) as the following example shows.

Example 3.2.2 Let $f : \mathbb{R} \to \hat{\mathcal{C}}(\mathbb{R})$,

$$f(x) := \begin{cases} \left\{ \begin{array}{ll} \frac{1}{x} \right\} & \text{if} & \exists n \in \mathbb{N} : x \in \left[2^{-2n}, 2^{-2n+1} \right) \\ \left\{ -\frac{1}{x} \right\} & \text{if} & \exists n \in \mathbb{N} : x \in \left[2^{-2n+1}, 2^{-2n+2} \right) \\ \emptyset & \text{else.} \end{cases} \end{cases}$$

Then the modified outer semi–continuous hull $(\widetilde{\operatorname{osc}} f)$ of f is obtained as

$$(\widetilde{\operatorname{osc}} f)(x) = \begin{cases} \left\{\frac{1}{x}\right\} & \text{if} & \exists n \in \mathbb{N} : x \in (2^{-2n}, 2^{-2n+1}) \\ \left\{-\frac{1}{x}\right\} & \text{if} & \exists n \in \mathbb{N} : x \in (2^{-2n+1}, 2^{-2n+2}) \\ \left[-\frac{1}{x}, \frac{1}{x}\right] & \text{if} & \exists n \in \mathbb{N} : x = 2^{-n} \\ \emptyset & \text{else.} \end{cases}$$

It is easily seen that $\operatorname{gr}(\widetilde{\operatorname{osc}} f)$ is not closed. Indeed, the sequence $\{(2^{-n}, 0)\}_{n \in \mathbb{N}}$ belongs to the graph of $(\widetilde{\operatorname{osc}} f)$, but its limit (0, 0) does not.



We next introduce a new notion of upper and lower limits, which is adapted to the framework of $\hat{\mathcal{C}}$ -valued functions. In the following, we frequently use Convention 1.2.2.

Definition 3.2.3 Let $f: X \to \hat{\mathcal{C}}$ and let $\bar{x} \in X$. The upper limit of f at \bar{x} is defined by

$$\limsup_{x \to \bar{x}} f(x) := \sup_{x_n \to \bar{x}} \limsup_{n \to \infty} f(x_n)$$

and the lower limit of f at \bar{x} is defined by

$$\liminf_{x \to \bar{x}} f(x) := \inf_{x_n \to \bar{x}} \liminf_{n \to \infty} f(x_n).$$

The limit of f at \bar{x} exists if the upper and lower limits coincide. Then we write

$$\lim_{x \to \bar{x}} f(x) = \limsup_{x \to \bar{x}} f(x) = \liminf_{x \to \bar{x}} f(x).$$

In case $\hat{\mathcal{C}} = \hat{\mathcal{C}}(\mathbb{R})$, these concepts coincide with the classical upper and lower limits of extended real-valued functions $f : X \to \mathbb{R} \cup \{-\infty, +\infty\}$ if we identify $f(\cdot)$ with the function $(\{f(\cdot)\} + \mathbb{R}_+) : X \to \hat{\mathcal{C}}^{\star}$ (where we set $\{+\infty\} + \mathbb{R}_+ = \emptyset$ and $\{-\infty\} + \mathbb{R}_+ = \mathbb{R}$).

As in the case of $\hat{\mathcal{F}}$ -valued functions, the upper and lower limits can be expressed by the supremum and infimum.

Proposition 3.2.4 Let $f: X \to \hat{\mathcal{C}}^{\star}$ and $\bar{x} \in X$. Then,

$$\limsup_{x \to \bar{x}} f(x) = \sup_{x_n \to \bar{x}} \sup_{N \in \mathcal{N}_{\infty}^{\#}} \inf_{n \in N} f(x_n),$$
$$\liminf_{x \to \bar{x}} f(x) = \inf_{x_n \to \bar{x}} \sup_{N \in \mathcal{N}_{\infty}} \inf_{n \in N} f(x_n).$$

Let $f: X \to \hat{\mathcal{C}}^{\diamond}$ and $\bar{x} \in X$. Then,

$$\limsup_{x \to \bar{x}} f(x) = \sup_{x_n \to \bar{x}} \inf_{N \in \mathcal{N}_{\infty}} \sup_{n \in N} f(x_n),$$
$$\liminf_{x \to \bar{x}} f(x) = \inf_{x_n \to \bar{x}} \inf_{N \in \mathcal{N}_{\infty}^{\#}} \sup_{n \in N} f(x_n).$$

Proof. Follows from Definition 2.2.1 and Definition 3.2.3.

In case of outer limits for $\hat{\mathcal{F}}$ -valued functions (see Definition 3.1.1), we observe that the set $\bigcup_{x_n \to \bar{x}} \text{LIM} \operatorname{SUP}_{n \to \infty} f(x_n)$ is always closed, i.e., the closure operation, which is implicitly contained in the infimum in formula (3.1), is superfluous. An analogous result is valid for $\hat{\mathcal{C}}$ -valued functions.

Proposition 3.2.5 Let $f: X \to \hat{\mathcal{C}}^*$ and $\bar{x} \in X$. Then it holds

$$\liminf_{x \to \bar{x}} f(x) = \bigcup_{x_n \to \bar{x}} \liminf_{n \to \infty} f(x_n)$$

Proof. We have to show that $\bigcup_{x_n \to \bar{x}} \liminf_{n \to \infty} f(x_n)$ is convex and closed.

(i) Convexity. Let $y_1, y_2 \in \bigcup_{x_n \to \bar{x}} \liminf_{n \to \infty} f(x_n)$ and $\lambda \in [0, 1]$ be given. Hence there exist sequences $\{x_n^{(i)}\}_{n \in \mathbb{N}} \subseteq X$, (i = 1, 2) with $x_n^{(i)} \to \bar{x}$ such that $y_i \in \liminf_{n \to \infty} f(x_n^{(i)})$. We define a sequence $\{x_n^{(3)}\}_{n \in \mathbb{N}} \subseteq X$ by

$$\left\{x_n^{(3)}\right\}_{n\in\mathbb{N}} := \left\{x_1^{(1)}, x_1^{(2)}, x_2^{(1)}, x_2^{(2)}, x_3^{(1)}, x_3^{(2)}, \ldots\right\}.$$

Since $\{x_n^{(i)}\}_{n\in\mathbb{N}}$, (i = 1, 2) are subsequences of $\{x_n^{(3)}\}_{n\in\mathbb{N}}$, Proposition 2.2.6 yields that $\liminf_{n\to\infty} f(x_n^{(i)}) \subseteq \liminf_{n\to\infty} f(x_n^{(3)})$, (i = 1, 2). Hence we obtain $\lambda y_1 + (1 - \lambda)y_2 \in \liminf_{n\to\infty} f(x_n^{(3)})$. From $x_n^{(3)} \to \bar{x}$ it follows that $\lambda y_1 + (1 - \lambda)y_2 \in \bigcup_{x_n \to \bar{x}} \liminf_{n\to\infty} f(x_n)$. (ii) *Closedness.* Let $\{y_m\}_{m\in\mathbb{N}} \subseteq \bigcup_{x_n\to\bar{x}} \liminf_{n\to\infty} f(x_n)$ be convergent to some $\bar{y} \in \mathbb{R}^p$. For all $m \in \mathbb{N}$ there exists a sequence $\{x_n^{(m)}\}_{n\in\mathbb{N}} \subseteq X$ such that $\bar{x} = \lim_{n\to\infty} x_n^{(m)}$ and

 $y_m \in \liminf_{n \to \infty} f(x_n^{(m)})$. Hence, we can construct a strictly increasing function $n_0 : \mathbb{N} \to \mathbb{N}$ by

$$\forall m \in \mathbb{N}, \ \exists n_0(m) \in \mathbb{N}, \ \forall n \ge n_0(m), \ \forall k \in \{1, ..., m\}: \ \left\| x_n^{(k)} - \bar{x} \right\| < \frac{1}{m}$$

An "inverse function" of $n_0 : \mathbb{N} \to \mathbb{N}$ can be defined by $m_0(n) := \sup \{m \in \mathbb{N} | n \ge n_0(m)\}$. Of course, $m_0 : \mathbb{N} \to \mathbb{N} \cup \{-\infty\}$ is (not necessarily strictly) increasing and we have $m_0(n) \to \infty$ for $n \to \infty$. Consider the sequence $\{\bar{x}_n\}_{n \in \mathbb{N}} \subseteq X$ defined by

$$\left\{\bar{x}_n\right\}_{n\in\mathbb{N}} := \left\{x_1^{(1)}, x_1^{(2)}, ..., x_1^{(m_0(1))}, x_2^{(1)}, x_2^{(2)}, ..., x_2^{(m_0(2))}, ..., x_n^{(1)}, x_n^{(2)}, ..., x_n^{(m_0(n))}, ...\right\}.$$

where, without loss of generality, it can be assumed that $m_0(n) \neq -\infty$ for all $n \in \mathbb{N}$. Clearly, the sequence $\{\bar{x}_n\}_{n \in \mathbb{N}}$ converges to \bar{x} and it has the following property:

$$\forall m \in \mathbb{N} : \left\{ x_n^{(m)} \right\}_{n \ge n_0(m)} \subseteq \left\{ \bar{x}_n \right\}_{n \in \mathbb{N}}.$$

By Proposition 2.2.6 (i), for all $m \in \mathbb{N}$ we have $\liminf_{n \to \infty} f(x_n^{(m)}) \subseteq \liminf_{n \to \infty} f(\bar{x}_n)$ and, consequently, the sequence $\{y_m\}_{m \in \mathbb{N}}$ is a subset of $\liminf_{n \to \infty} f(\bar{x}_n)$. Since $\liminf_{n \to \infty} f(\bar{x}_n)$ is a closed subset of \mathbb{R}^p it follows that $\bar{y} \in \liminf_{n \to \infty} f(\bar{x}_n) \subseteq \bigcup_{x_n \to \bar{x}} \liminf_{n \to \infty} f(x_n)$. \Box

The following relationship between the outer and inner limits in $\hat{\mathcal{F}}$ and the upper and lower limits in $\hat{\mathcal{C}}$ is an easy consequence of the definition. For $f: X \to \hat{\mathcal{C}}^{\star}$ we have

$$\liminf_{x \to \bar{x}} f(x) \supseteq \underset{x \to \bar{x}}{\text{LIM}} \underset{x \to \bar{x}}{\text{SUP}} f(x) \quad \text{and} \quad \limsup_{x \to \bar{x}} f(x) \supseteq \underset{x \to \bar{x}}{\text{LIM}} \underset{x \to \bar{x}}{\text{INF}} f(x).$$
(3.2)

In case of $f: X \to \hat{\mathcal{C}}^\diamond$, it always holds

$$\liminf_{x\to \bar{x}} f(x) \supseteq \operatornamewithlimits{LIM\,INF}_{x\to \bar{x}} f(x) \qquad \text{and} \qquad \limsup_{x\to \bar{x}} f(x) \supseteq \operatornamewithlimits{LIM\,SUP}_{x\to \bar{x}} f(x).$$

We next define semi–continuity concepts for functions $\hat{\mathcal{C}}$ -valued functions.

Definition 3.2.6 A function $f: X \to \hat{\mathcal{C}}$ is lower semi–continuous (lsc) at $\bar{x} \in X$ if

$$f(\bar{x}) \le \liminf_{x \to \bar{x}} f(x).$$

A function $f: X \to \hat{\mathcal{C}}$ is said to be upper semi-continuous (usc) at $\bar{x} \in X$ if

$$f(\bar{x}) \ge \limsup_{x \to \bar{x}} f(x).$$

A function $f: X \to \hat{\mathcal{C}}$ is said to be continuous at $\bar{x} \in X$ if f is simultaneously lsc and usc at \bar{x} . If f is lsc (usc, continuous) at every $\bar{x} \in X$ we say f is lsc (usc, continuous).

Of course, f is continuous at $\bar{x} \in X$ if and only if $f(\bar{x}) = \lim_{x \to \bar{x}} f(x)$.

Proposition 3.2.7 Let $f: X \to \hat{\mathcal{C}}^*$ and $\bar{x} \in X$. If f is isc (lsc) at \bar{x} , then f is usc (osc) at the same point.

3.2. Semi-continuity

Proof. Follows from (3.2).

Another benefit of our semi-continuity concept is the following characterization, which will be used in the proof of the biconjugation theorem in Section 3.6 below. The proof is based on the assertions of Section 2.3.

Theorem 3.2.8 Let $f: X \to \hat{\mathcal{C}}^{\star}$ and let $\bar{x} \in \text{dom } f$. Then the following statements are equivalent:

- (i) f is lsc at \bar{x} ,
- (ii) For all $y^* \in \operatorname{ri} \left(0^+ f(\bar{x})\right)^\circ$ the function

 $\bar{f}_{y^*}: X \to \mathbb{R} \cup \{-\infty, +\infty\}, \quad \bar{f}_{y^*}(x) := -\delta^* (y^* | f(x))$

is lsc at \bar{x} .

Proof. Let be given an arbitrary sequence $\{x_n\}_{n\in\mathbb{N}}$ with $x_n\to \bar{x}$.

(i) \Rightarrow (ii). Let the sequence $\{\tilde{x}_n\}_{n\in\mathbb{N}}$ be defined by $\tilde{x}_{2n} := x_n$ and $\tilde{x}_{2n+1} := \bar{x}$. From (i) we deduce that $f(\bar{x}) = \liminf_{n \to \infty} f(\tilde{x}_n)$. Lemma 2.3.5 implies that

$$\forall y^* \in \operatorname{ri}\left(0^+ f(\bar{x})\right)^\circ : \quad -\delta^*\left(y^*|f(\bar{x})\right) = \liminf_{n \to \infty} -\delta^*\left(y^*|f(\bar{x}_n)\right) \le \liminf_{n \to \infty} -\delta^*\left(y^*|f(x_n)\right).$$

(ii) \Rightarrow (i). With aid of Proposition 1.3.7 we obtain

$$\forall y^* \in \operatorname{ri}\left(0^+ f(\bar{x})\right)^\circ : -\delta^*\left(y^* | f(\bar{x})\right) \le \liminf_{n \to \infty} -\delta^*\left(y^* | f(x_n)\right) \le -\delta^*\left(y^* | \liminf_{n \to \infty} f(x_n)\right).$$

om Lemma 1.3.1 (iii) we deduce that $f(\bar{x}) \le \liminf_{n \to \infty} f(x_n)$.

From Lemma 1.3.1 (iii) we deduce that $f(\bar{x}) \leq \liminf_{n \to \infty} f(x_n)$.

The next assertion about nested lower limits is essential for an expedient definition of the lower semi-continuous hull of a $\hat{\mathcal{C}}^*$ -valued function. An analogous assertion for the upper limit is not true, see the Example 3.2.15 below.

Proposition 3.2.9 Let $f: X \to \hat{\mathcal{C}}^*$ and $\bar{x} \in X$. Then it holds

$$\liminf_{x \to \bar{x}} f(x) = \liminf_{x \to \bar{x}} \left(\liminf_{w \to x} f(w)\right).$$

Proof. Clearly, we have $f(x) \ge \liminf_{w \to x} f(w)$ for all $x \in X$ and hence

$$\liminf_{x \to \bar{x}} f(x) \ge \liminf_{x \to \bar{x}} \left(\liminf_{w \to x} f(w)\right).$$

In order to show the opposite inequality let $\{x_n\}_{n\in\mathbb{N}}$ with $x_n\to \bar{x}$ be arbitrarily given. For all $y^* \in \operatorname{ri} \left(0^+ \liminf_{n \to \infty} f(x_n) \right)^\circ$ it holds

$$-\delta^* \left(y^* \left| \liminf_{n \to \infty} \left(\liminf_{w \to x_n} f(w) \right) \right) \stackrel{\text{Pr. 1.3.7}}{\geq} \liminf_{n \to \infty} \left(\liminf_{w \to x_n} -\delta^* \left(y^* | f(w) \right) \right) \\ = \liminf_{n \to \infty} -\delta^* \left(y^* | f(x_n) \right) \\ \stackrel{\text{Lem. 2.3.5}}{=} -\delta^* \left(y^* | \liminf_{n \to \infty} f(x_n) \right).$$

Lemma 1.3.1 (iii) yields that $\liminf_{n\to\infty} (\liminf_{w\to x_n} f(w)) \subseteq \liminf_{n\to\infty} f(x_n)$. Hence $\liminf_{n\to\infty} (\liminf_{w\to x_n} f(w)) \ge \inf_{x_n\to\bar{x}} \liminf_{n\to\infty} f(x_n) = \liminf_{x\to\bar{x}} f(x)$. \Box

We next introduce the *lower semi-continuous hull* of a \hat{C}^{\star} -valued function.

Definition 3.2.10 Let $f : X \to \hat{\mathcal{C}}^*$. The lower semi-continuous hull of f is the function $(\operatorname{lsc} f) : X \to \hat{\mathcal{C}}^*$, defined by

$$(\operatorname{lsc} f)(x) := \liminf_{x' \to x} f(x').$$

In order to show that the lower semi-continuous hull of a function $f: X \to \hat{\mathcal{C}}^*$ has similar (but not completely the same) properties as the outer semi-continuous hull we define the epigraph of f as well as its closedness.

Definition 3.2.11 The epigraph of a function $f: X \to \hat{\mathcal{C}}^{\star}$ is defined to be the set

$$\operatorname{epi} f := \left\{ (x, A) \in X \times \hat{\mathcal{C}}^{\star} | A \ge f(x) \right\}.$$

Definition 3.2.12 A subset $\mathcal{A} \subseteq X \times \hat{\mathcal{C}}^*$ is closed if for every sequence $\{(x_n, A_n)\}_{n \in \mathbb{N}} \subseteq \mathcal{A}$ with $x_n \to \bar{x} \in X$ and $A_n \to \bar{A} \in \hat{\mathcal{C}}^*$ (with respect to \mathcal{C} -convergence) it is true that $(\bar{x}, \bar{A}) \in \mathcal{A}$. The closure of a set $\mathcal{A} \subseteq X \times \hat{\mathcal{C}}^*$ is defined to be the set of all $(\bar{x}, \bar{A}) \in X \times \hat{\mathcal{C}}^*$ such that there is a sequence $\{(x_n, A_n)\}_{n \in \mathbb{N}} \subseteq \mathcal{A}$ with $x_n \to \bar{x}$ and $A_n \to \bar{A}$. The closure of \mathcal{A} is denoted by $cl \mathcal{A}$.

Let us collect some properties of the lower semi-continuous hull of a $\hat{\mathcal{C}}^{\star}$ -valued function.

Proposition 3.2.13 For $f: X \to \hat{\mathcal{C}}^*$ the following statements hold true:

- (i) $\operatorname{gr}(\operatorname{lsc} f) \supseteq \operatorname{cl}(\operatorname{gr} f)$,
- (ii) $\operatorname{epi}(\operatorname{lsc} f) \supseteq \operatorname{cl}(\operatorname{epi} f)$,
- (iii) $(\operatorname{lsc} f)$ is lsc,
- (iv) $\forall x \in X$: $(\operatorname{lsc} f)(x) \le f(x),$
- (v) f is lsc at $\bar{x} \in X \iff (\operatorname{lsc} f)(\bar{x}) = f(\bar{x}),$
- (vi) $\operatorname{gr}(\operatorname{lsc} f)$ is closed,
- (vii) epi(lsc f) is closed.

Proof. (i) Let $(\bar{x}, \bar{y}) \in cl(gr f)$. Then there exists a sequence $\{(x_n, y_n)\}_{n \in \mathbb{N}} \subseteq gr f$ converging to (\bar{x}, \bar{y}) . Let $\{y_n\}, \{\bar{y}\} \in \mathcal{C}^*$. For all $n \in \mathbb{N}$, we have $\{y_n\} \ge f(x_n)$. Hence

$$\{\bar{y}\} = \lim_{n \to \infty} \{y_n\} = \liminf_{n \to \infty} \{y_n\} \stackrel{\text{rr. 2.2.1}}{=} \liminf_{n \to \infty} f(x_n)$$

$$\geq \inf_{x'_n \to \bar{x}} \liminf_{n \to \infty} f(x'_n) = \liminf_{x \to \bar{x}} f(x) = (\operatorname{lsc} f)(\bar{x}), \qquad (3.3)$$

i.e., $(\bar{x}, \bar{y}) \in \operatorname{gr}(\operatorname{lsc} f)$.

(ii) Let $(\bar{x}, \bar{A}) \in cl (epi f)$. Then there exist $\{x_n\}_{n \in \mathbb{N}} \subseteq X$ and $\{A_n\}_{n \in \mathbb{N}} \subseteq \hat{\mathcal{C}}^{\star}$ such that $\bar{x} = \lim_{n \to \infty} x_n$, $\bar{A} = \lim_{n \to \infty} A_n$ and $A_n \geq f(x_n)$ for all $n \in \mathbb{N}$. Similarly to (3.3), we obtain $(\bar{x}, \bar{A}) \in epi (lsc f)$.

(iii) Let $\bar{x} \in X$ be arbitrarily given. Then, by Proposition 3.2.9, we have

$$(\operatorname{lsc} f)(\bar{x}) = \liminf_{x \to \bar{x}} f(x) = \liminf_{x \to \bar{x}} \left(\liminf_{w \to x} f(w)\right) = \liminf_{x \to \bar{x}} (\operatorname{lsc} f)(x)$$

(iv) Choosing the special sequence $x_n \equiv x$ we obtain

$$(\operatorname{lsc} f)(x) = \liminf_{x' \to x} f(x') \le \liminf_{n \to \infty} f(x_n) = \liminf_{n \to \infty} f(x) = f(x).$$

(v) By definition, f is lsc at \bar{x} if and only if $f(\bar{x}) \leq (\operatorname{lsc} f)(\bar{x})$. By (iv) this equivalent to $f(\bar{x}) = (\operatorname{lsc} f)(\bar{x})$.

(vi) Let $\{(x_n, y_n)\}_{n \in \mathbb{N}} \subseteq \operatorname{gr}(\operatorname{lsc} f)$ with $(x_n, y_n) \to (\bar{x}, \bar{y}) \in X \times \mathbb{R}^p$ be given. Proceeding as in (i), but replacing f by $(\operatorname{lsc} f)$, we obtain $\{\bar{y}\} \ge (\operatorname{lsc}(\operatorname{lsc} f))(\bar{x})$. From (iii) we conclude that $(\operatorname{lsc}(\operatorname{lsc} f))(\bar{x}) = (\operatorname{lsc} f)(\bar{x})$. Hence $(\bar{x}, \bar{y}) \in \operatorname{gr}(\operatorname{lsc} f)$.

(vii) Let $\{(x_n, A_n)\}_{n \in \mathbb{N}} \subseteq \operatorname{epi}(\operatorname{lsc} f)$ with $\bar{x} = \lim_{n \in \mathbb{N}} x_n$, $\bar{A} = \lim_{n \in \mathbb{N}} A_n$. Proceeding as in (ii), but replacing f by $(\operatorname{lsc} f)$, we obtain $(\bar{x}, \bar{A}) \in \operatorname{epi}(\operatorname{lsc}(\operatorname{lsc} f))$. From (iii) we conclude that $(x, \bar{A}) \in \operatorname{epi}(\operatorname{lsc} f)$.

The next example shows that neither the closedness of epi f nor the closedness of gr f implies that f is lsc.

Example 3.2.14 Let $f : \mathbb{R} \to \hat{\mathcal{C}}^{\star}(\mathbb{R})$ be defined by

$$f(x) = \begin{cases} \{1/x\} & \text{if } x \neq 0\\ \emptyset & \text{if } x = 0. \end{cases}$$

It can be easily seen that $\operatorname{gr} f \subseteq \mathbb{R} \times \mathbb{R}$ is closed. Moreover, we observe that $\operatorname{epi} f \subseteq \mathbb{R} \times \hat{\mathcal{C}}^*(\mathbb{R})$ is closed. Indeed, let $\{(x_n, A_n)\}_{n \in \mathbb{N}} \subseteq \operatorname{epi} f$ be a sequence converging to $(0, \overline{A})$. (the case $\overline{x} \neq 0$ is obvious). We can assume that there is a subsequence $\{x_n\}_{n \in \hat{N}} (\hat{N} \in \mathcal{N}_{\infty}^{\#})$ of $\{x_n\}_{n \in \mathbb{N}}$ such that $x_n > 0$ for all $n \in \hat{N}$ or $x_n < 0$ for all $n \in \hat{N}$ (otherwise we obviously have $\overline{A} = \emptyset$). Then, by Proposition 2.2.6, it follows $\overline{A} = \lim_{n \to \infty} A_n = \limsup_{n \to \infty} A_n \ge \limsup_{n \in \hat{N}} A_n \ge \lim_{n \in \hat{N}} A_n \ge \lim_{n \in \hat{N}} A_n$. Now it is easy to see that $\liminf_{n \in \hat{N}} A_n = \emptyset$ and hence $\overline{A} = \emptyset$. Of course, it holds $(0, \emptyset) \in \operatorname{epi} f$. However, f is not lsc, because (lsc f) can be easily calculated as

$$(\operatorname{lsc} f)(x) = \begin{cases} \{1/x\} & \text{if } x \neq 0\\ \mathbb{R} & \text{if } x = 0. \end{cases}$$

In Remark 3.1.10 (due to [68]) we noticed that an inner semi-continuous hull of a $\hat{\mathcal{F}}$ -valued function that is analogously defined to the outer semi-continuous hull is not necessarily inner semi-continuous. There are analogous problems with the upper semi-continuous hull of $\hat{\mathcal{C}}^{\star}$ -valued functions. This is due to the fact that there is no analogous assertion to Proposition 3.2.9 for upper limits, as the following example shows.

Example 3.2.15 For functions $f: X \to \hat{\mathcal{C}}^{\star}$, in general, we have

$$\limsup_{x \to \bar{x}} f(x) \neq \limsup_{x \to \bar{x}} \left(\limsup_{w \to x} f(w)\right).$$

Indeed, consider the function $f : \mathbb{R}^2 \to \hat{\mathcal{C}}^{\star}(\mathbb{R})$, defined by

$$f(x) := \begin{cases} \{\|x\|\} & \text{if } x_1 \ge 0\\ \{-\|x\|\} & \text{if } x_1 < 0. \end{cases}$$

Then it holds

$$\limsup_{w \to x} f(w) := \begin{cases} \{ \|x\| \} & \text{if } x_1 > 0 \text{ or } x_2 = 0 \\ \{ -\|x\| \} & \text{if } x_1 < 0 \\ \emptyset & \text{if } x_1 = 0 \text{ and } x_2 \neq 0. \end{cases}$$

Hence we obtain

$$\{0\} = \limsup_{x \to 0} f(x) \neq \limsup_{x \to 0} \left(\limsup_{w \to x} f(w)\right) = \emptyset.$$

3.3 Locally bounded functions

The concept of local boundedness of a set-valued map plays an important role in Variational Analysis, see [68]. As an easy consequence of the definition ([68, Definition 5.14]), local boundedness of a map $f : \mathbb{R}^n \Rightarrow \mathbb{R}^p$ at \bar{x} implies that $f(\bar{x})$ is a bounded subset of \mathbb{R}^p . This means, local boundedness is (at least locally) adapted to set-valued maps with bounded values. Therefore we introduce a slightly generalized concept, adapted to the framework of $\hat{\mathcal{C}}^*$ -valued functions. It turns out that this concept provides a sufficient condition for the coincidence of lower (upper) semi-continuity with outer (inner) semi-continuity. In this section, we set $X = \mathbb{R}^n$.

Definition 3.3.1 A function $f: X \to \hat{\mathcal{C}}^*$ is said to be locally bounded at $\bar{x} \in \text{dom } f$ if there exists a neighborhood $V \in \mathcal{N}(\bar{x})$ such that the following conditions are satisfied:

- (i) $0^+ \inf_{x \in V} f(x) \subseteq 0^+ f(\bar{x}),$
- (ii) $\forall x \in V \cap \operatorname{dom} f : 0^+ f(x) \supseteq 0^+ f(\bar{x}).$

Remark 3.3.2 If $f: X \to \hat{\mathcal{C}}^*$ locally bounded at $\bar{x} \in \text{dom } f$, (i) and (ii) of the previous definition are always satisfied with equality. (Indeed, the opposite inclusion in (i) follows from $\inf_{x \in V} f(x) \leq f(\bar{x})$ and Proposition A.9 and the opposite inclusion in (ii) follows from (i).) In view of the embedding theorem in the form of Corollary 1.3.4 this means, locally, the values of f can be embedded into a partially ordered linear space (extended by a largest element) and the embedding yields a set in the linear space which is bounded below.

We next clarify the relations between our local boundedness concept and the classical one.

Proposition 3.3.3 Let $f: X \to \hat{C}^*$ such that $f(\bar{x})$ is bounded for some $\bar{x} \in \text{dom } f$. Then f is locally bounded in the sense of the above definition if and only if f is locally bounded in the sense of [68] (where f is understood to be a set-valued map $f: X \rightrightarrows \mathbb{R}^p$ in the latter case).

Proof. In the present situation, condition (i) of Definition 3.3.1 has the meaning of $0^+ \operatorname{cl} \operatorname{conv} \bigcup_{x \in V} f(x) = \{0\}$ and condition (ii) is always satisfied. This is equivalent to the boundedness of $\bigcup_{x \in V} f(x)$.

Theorem 3.3.4 Let $f: X \to \hat{\mathcal{C}}^*$ be locally bounded at $\bar{x} \in \text{dom } f$. Then,

$$\liminf_{x \to \bar{x}} f(x) = \operatorname{cl} \operatorname{conv} \operatorname{LIMSUP}_{x \to \bar{x}} f(x).$$

Proof. Clearly, we have $\liminf_{x\to\bar{x}} f(x) \supseteq \operatorname{cl} \operatorname{conv} \operatorname{LIM} \operatorname{SUP}_{x\to\bar{x}} f(x)$. To show the opposite inclusion let $y \in \liminf_{x\to\bar{x}} f(x)$ be given. Then there exists a sequence $\{x_n\}_{n\in\mathbb{N}}$ with $x_n\to\bar{x}$ such that $y \in \liminf_{n\to\infty} f(x_n)$. Assuming that there exists some $n_0 \in \mathbb{N}$ such that $f(x_n) = \emptyset$ for all $n \ge n_0$, we obtain $\liminf_{n\to\infty} f(x_n) = \emptyset$, which contradicts $y \in \liminf_{x\to\bar{x}} f(x)$. Hence, by $\{x_{n_k}\}_{k\in\mathbb{N}} := \{x_n\}_{n\in\mathbb{N}} \cap \operatorname{dom} f$, we obtain a subsequence of $\{x_n\}_{n\in\mathbb{N}}$. By the definition of the lower limit, we have $\liminf_{n\to\infty} f(x_n) = \liminf_{k\to\infty} f(x_{n_k})$. By the local boundedness, we find $k_0 \in \mathbb{N}$ such that, setting $K := 0^+ f(\bar{x})$, $f(x_{n_k})$ and $\inf_{k\geq k_0} f(x_{n_k})$ belong to $\hat{\mathcal{C}}_K^*$ for all $k \ge k_0$. Theorem 2.2.18 yields $y \in \operatorname{cl} \operatorname{conv} \operatorname{LIM} \operatorname{SUP}_{k\to\infty} f(x_{n_k})$, hence we obtain $y \in \operatorname{cl} \operatorname{conv} \bigcup_{x_n\to\bar{x}} \operatorname{LIM} \operatorname{SUP}_{n\to\infty} f(x_n) = \operatorname{LIM} \operatorname{SUP}_{x\to\bar{x}} f(x)$.

In the next example we show that the assertion of the preceeding theorem can fail if one of the conditions in the definition of the local boundedness concept is not satisfied.

Example 3.3.5 Let $f : \mathbb{R} \to \mathcal{C}^{\star}(\mathbb{R})$ be defined by

$$f(x) = \begin{cases} \left\{ \frac{1}{x} \right\} & \text{if } x \neq 0\\ \left\{ 0 \right\} & \text{if } x = 0, \end{cases}$$

i.e., (ii) is satisfied, but (i) is not. Then we have

$$\mathbb{R} = \liminf_{x \to 0} f(x) \neq \text{cl conv LIM}_{x \to 0} \text{SUP} f(x) = \{0\}.$$

Example 3.3.6 Let $f : \mathbb{R} \to \mathcal{C}^{\star}(\mathbb{R}^2)$ be defined by

$$f(x) = \begin{cases} \{y \in \mathbb{R}^2 | y_2 = 1, y_1 = 1/x\} & \text{if } x \neq 0\\ \{y \in \mathbb{R}^2 | y_2 = 0\} & \text{if } x = 0, \end{cases}$$

i.e., (i) is satisfied, but (ii) is not. An easy calculation shows that

$$\{y \in \mathbb{R}^2 | \ 0 \le y_2 \le 1\} = \liminf_{x \to 0} f(x) \ne \text{cl conv LIM}_{x \to 0} \text{SUP} \ f(x) = \{y \in \mathbb{R}^2 | \ y_2 = 0\}$$



The local boundedness of a function $f: X \to \hat{\mathcal{C}}^*$ at a point $\bar{x} \in \text{dom } f$ also implies that $\limsup_{x \to \bar{x}} f(x) = \text{LIM INF}_{x \to \bar{x}} f(x)$ (see Corollary 3.3.9 below). However, as shown in the next theorem, a weaker assumption is already sufficient.

Theorem 3.3.7 Let $f: X \to \hat{\mathcal{C}}^*$ be a function and let $\bar{x} \in \text{dom } f$ such that for all sequences $\{x_n\}_{n \in \mathbb{N}} \subseteq X$ with $x_n \to \bar{x}$ there exists a subsequence $\{x_{n_k}\}_{k \in \mathbb{N}}$ and a nonempty closed convex cone $K \subseteq \mathbb{R}^p$ with

$$\forall k \in \mathbb{N} : f(x_{n_k}) \in \mathcal{C}_K^{\star} \qquad and \qquad \inf_{k \in \mathbb{N}} f(x_{n_k}) \in \mathcal{C}_K^{\star}.$$

Then it holds

$$\limsup_{x\to \bar{x}} f(x) = \operatornamewithlimits{LIMINF}_{x\to \bar{x}} F f(x)$$

Proof. Of course, $\limsup_{x\to \bar{x}} f(x) \supseteq \operatorname{LIM}\operatorname{INF}_{x\to \bar{x}} f(x)$. In order to show the opposite inclusion let $y \in \mathbb{R}^p \setminus \operatorname{LIM}\operatorname{INF}_{x\to \bar{x}} f(x)$ be given (the case $\operatorname{LIM}\operatorname{INF}_{x\to \bar{x}} f(x) = \mathbb{R}^p$ is obvious). Hence there exists a sequence $\{x_n\}_{n\in\mathbb{N}} \subseteq X$ with $x_n \to \bar{x}$ such that $y \notin \operatorname{LIM}\operatorname{INF}_{n\to\infty} f(x_n)$. Every subsequence of $\{x_n\}_{n\in\mathbb{N}}$ is again a sequence converging to \bar{x} , hence our assumption ensures that Theorem 2.2.19 is applicable. It follows that $y \notin \limsup_{n\to\infty} f(x_n)$. \Box

The next example shows that the assertion of the previous theorem can fail if the assumption is not satisfied.

Example 3.3.8 Let $f : \mathbb{R} \to \mathcal{C}^*(\mathbb{R}^2)$ be defined by

$$f(x) = \begin{cases} \mathbb{R}^2 & \text{if } x \le 0\\ \text{conv}\left\{\left(-1, -\frac{1}{x}\right), \left(1, \frac{1}{x}\right)\right\} & \text{if } x > 0, \end{cases}$$

i.e., the condition in the previous theorem is not satisfied. Then we have

$$\{y \in \mathbb{R}^2 | -1 \le y_2 \le 1\} = \limsup_{x \to 0} f(x) \ne \underset{x \to 0}{\text{LIM} \text{INF}} f(x) = \{y \in \mathbb{R}^2 | y_2 = 0\}.$$

3.4. Convex functions



Corollary 3.3.9 Let $f: X \to \hat{\mathcal{C}}^*$ be locally bounded at $\bar{x} \in \text{dom } f$. Then,

$$\limsup_{x \to \bar{x}} f(x) = \operatorname{LIM}_{x \to \bar{x}} \operatorname{INF} f(x).$$

Proof. By the local boundedness of f at \bar{x} , for every sequence $\{x_n\}_{n\in\mathbb{N}}$ with $x_n \to \bar{x}$ there exists some $n_0 \in \mathbb{N}$ such that

$$\forall n \ge n_0: f(x_n) \in \mathcal{C}_K^{\star}$$
 and $\inf_{n \ge n_0} f(x_n) \in \mathcal{C}_K^{\star}$

where $K := 0^+ f(\bar{x})$. Hence, Theorem 3.3.7 yields the desired assertion.

Corollary 3.3.10 Let $f: X \to \hat{\mathcal{C}}^*$ be locally bounded at every $\bar{x} \in \text{dom } f$. Then the following statements are equivalent:

- (i) epi $f \subseteq X \times \hat{\mathcal{C}}^{\star}$ is closed,
- (ii) f is lsc,
- (iii) gr $f \subseteq X \times \mathbb{R}^p$ is closed.

Proof. (i) ⇒ (iii). Elementary (using Proposition 2.2.13 (i)).
(iii) ⇒ (ii). Proposition 3.1.5 yields that f is osc. By Theorem 3.3.4, f is lsc.
(ii) ⇒ (i). Follows from Proposition 3.2.13 (v), (vii).

3.4 Convex functions

In this section, we investigate convex and concave $\hat{\mathcal{C}}$ -valued functions. As a main result, we show that semi-continuity in the sense of Section 3.2 coincides with the classical concepts of outer and inner semi-continuity. We next introduce convex and concave $\hat{\mathcal{C}}$ -valued functions. We proceed analogously to the scalar case. We set $X = \mathbb{R}^n$ even though many assertions are also valid in more general context.

Definition 3.4.1 A function $f: X \to \hat{\mathcal{C}}$ is said to be convex if

$$\forall \lambda \in [0,1], \, \forall x_1, x_2 \in X: \quad f(\lambda \cdot x_1 + (1-\lambda) \cdot x_2) \le \lambda f(x_1) \oplus (1-\lambda) f(x_2).$$

A concave function is analogously defined (replacing $\leq by \geq$).

The following figure shows (the graph of) a convex and a concave function $f: \mathbb{R} \to \hat{\mathcal{C}}^{\star}(\mathbb{R})$.



Proposition 3.4.2 A function $f : X \to \hat{\mathcal{C}}$ is convex (concave) if and only if $\operatorname{epi} f \subseteq X \times \hat{\mathcal{C}}$ (hyp $f \subseteq X \times \hat{\mathcal{C}}$) is convex.

Proof. The proof is immediate.

Usually, convexity of a set–valued map is defined by convexity of its graph. The relationship is as follows.

Proposition 3.4.3 A function $f : X \to \hat{C}^*$ is convex if and only if gr f is likewise. A function $f : X \to \hat{C}^\diamond$ is concave if and only if gr f is convex.

Proof. Follows from the definition.

It can be easily seen that a concave function $f: X \to \hat{\mathcal{C}}^*$ and a convex function $f: X \to \hat{\mathcal{C}}^*$ do not have convex graphs, in general. Functions of this (or similar) type are called "fans" by Ioffe, compare [38]. In the most cases these functions have worse properties than its counterparts (with convex graph). In Chapter 4, for instance, they occur as objective functions of dual problems.

The following proposition shows that the values of a convex \hat{C}^* -valued function essentially have the same recession cone. An analogous result for concave \hat{C}^* -valued functions is not true. A systematic study of assertions of this type can be found in [54].

Proposition 3.4.4 Let $f: X \to \hat{\mathcal{C}}^*$ be convex. If $\bar{x} \in \operatorname{ridom} f$, then $0^+f(x) \subseteq 0^+f(\bar{x})$ for all $x \in \operatorname{dom} f$ and $0^+f(x) = 0^+f(\bar{x})$ for all $x \in \operatorname{ri}(\operatorname{dom} f)$.

Proof. Note that dom f is convex. Let $x \in \text{dom } f$ be arbitrarily given and, by hypothesis, $\bar{x} \in \text{ri dom } f$. By [1, Theorem 6.4], there exists $\mu > 1$ such that $\hat{x} := \mu \bar{x} + (1 - \mu)x \in \text{dom } f$.

Set $\lambda := 1/\mu \in (0, 1)$. The convexity of f yields $f(\bar{x}) \supseteq \lambda f(\hat{x}) \oplus (1-\lambda)f(x)$. Since $\hat{x} \in \text{dom } f$ we can choose some $\hat{y} \in f(\hat{x})$, hence $f(\bar{x}) \supseteq \lambda \{\hat{y}\} + (1-\lambda)f(x) := C_x$. Of course, $f(x) \subseteq \mathbb{R}^p$ is a nonempty closed convex set for each $x \in \text{dom } f$. Therefore, Proposition A.9 yields $0^+C_x \subseteq 0^+f(\bar{x})$. With the aid of [1, Theorem 8.1] we conclude that $0^+C_x = 0^+f(x)$, hence $0^+f(x) \subseteq 0^+f(\bar{x})$.

Assume there is some $\tilde{x} \in \text{ri dom } f$ with $0^+ f(\tilde{x}) \subsetneq 0^+ f(\bar{x})$, then the first part yields $0^+ f(x) \subseteq 0^+ f(\tilde{x})$ for all $x \in \text{dom } f$, whence the contradiction $0^+ f(\bar{x}) \subsetneq 0^+ f(\bar{x})$. \Box

Theorem 3.4.5 Let $f: X \to \hat{\mathcal{C}}^{\star}$ be convex. Then, for all $\bar{x} \in X$ it holds

$$\liminf_{x \to \bar{x}} f(x) = \operatorname{LIM}_{x \to \bar{x}} \operatorname{SUP} f(x).$$

Proof. Of course, we always have $\liminf_{x\to \bar{x}} f(x) \supseteq \operatorname{LIM} \operatorname{SUP}_{x\to \bar{x}} f(x)$. To show the opposite inclusion let $y \in \liminf_{x\to \bar{x}} f(x)$ be given. Hence there exists a sequence $\{x_n\}_{n\in\mathbb{N}}$ with $x_n \to \bar{x}$ such that $y \in \liminf_{n\to\infty} f(x_n)$. By Proposition 2.2.15 this can be written as

$$\exists \left\{ (\lambda_0^{(m)}, ..., \lambda_p^{(m)}), (k_0^{(m)}, ..., k_p^{(m)}), (z_0^{(m)}, ..., z_p^{(m)}) \right\}_{m \in \mathbb{N}} \subseteq [0, 1]^{p+1} \times \mathbb{N}^{p+1} \times (\mathbb{R}^p)^{p+1} : \\ y = \lim_{m \to \infty} \sum_{j=0}^p \lambda_j^{(m)} z_j^{(m)}, \\ \forall j \in \{0, 1, ..., p\}, \forall m \in \mathbb{N} : \quad z_j^{(m)} \in f(x_{k_j^{(m)}}), \\ \forall j \in \{0, 1, ..., p\}, \forall m \in \mathbb{N} : \quad k_j^{(m)} \ge m, \\ \forall m \in \mathbb{N} : \quad \sum_{j=0}^p \lambda_j^{(m)} = 1. \end{cases}$$

We define two sequences $\{y_m\}_{m\in\mathbb{N}}\subseteq\mathbb{R}^p$ and $\{\tilde{x}_m\}_{m\in\mathbb{N}}\subseteq X$ by

$$y_m := \sum_{j=0}^p \lambda_j^{(m)} z_j^{(m)} \qquad \tilde{x}_m := \sum_{j=0}^p \lambda_j^{(m)} x_{k_j^{(m)}}.$$

Then we have $y_m \to y$, $\tilde{x}_m \to \bar{x}$ and the convexity of f yields that

$$y_m = \sum_{j=0}^p \lambda_j^{(m)} z_j^{(m)} \in \sum_{j=0}^p \lambda_j^{(m)} f\left(x_{k_j^{(m)}}\right) \subseteq f\left(\sum_{j=0}^p \lambda_j^{(m)} x_{k_j^{(m)}}\right) = f\left(\tilde{x}_m\right)$$

for all $m \in \mathbb{N}$. By [68, 5(1)], this means $y \in \text{LIM} \operatorname{SUP}_{x \to \bar{x}} f(x)$.

Corollary 3.4.6 Let $f: X \to \hat{\mathcal{C}}^*$ be convex. Then the following statements hold true:

(i) $(\operatorname{lsc} f) = (\operatorname{osc} f),$

(ii) $(\operatorname{lsc} f)$ is convex,

(iii) (lsc f): $X \to \hat{\mathcal{C}}_K^{\star}$ for some nonempty closed convex cone $K \subseteq \mathbb{R}^p$.

Proof. (i) Follows from Theorem 3.4.5.

(ii) f convex \Leftrightarrow gr f convex \Rightarrow cl (gr f) = gr (osc f) convex \Leftrightarrow osc f = lsc f convex.

(iii) Since lsc f is osc and convex, its graph is closed and convex. If dom (lsc f) = \emptyset there is nothing to prove, otherwise, there exists some $\bar{x} \in \operatorname{ri} \operatorname{dom} (\operatorname{lsc} f)$. From Proposition 3.4.4 we deduce that $0^+(\operatorname{lsc} f)(x) \subseteq 0^+(\operatorname{lsc} f)(\bar{x}) =: K$ for all $x \in \operatorname{dom} (\operatorname{lsc} f)$. It remains to prove the opposite inclusion for all $x \in \operatorname{dom} (\operatorname{lsc} f)$. Indeed, let $\hat{y} \in 0^+(\operatorname{lsc} f)(\bar{x})$ and $\bar{y} \in (\operatorname{lsc} f)(\bar{x})$ be arbitrarily chosen. By [1, Theorem 8.3] we have $\bar{y} + \lambda \hat{y} \in (\operatorname{lsc} f)(\bar{x})$ for all $\lambda \geq 0$ and equivalently $(0, \hat{y}) \in 0^+ \operatorname{gr} (\operatorname{lsc} f)$. Given some $x \in \operatorname{dom} (\operatorname{lsc} f)$ we can choose $y \in (\operatorname{lsc} f)(x)$. Since $(0, \hat{y}) \in 0^+ \operatorname{gr} (\operatorname{lsc} f)$, [1, Theorem 8.3] yields that $y + \lambda \hat{y} \in (\operatorname{lsc} f)(x)$ for all $\lambda \geq 0$ and equivalently $\hat{y} \in 0^+(\operatorname{lsc} f)(x)$.

Corollary 3.4.7 For some nonempty closed convex cone $K \subseteq \mathbb{R}^p$ and a function $f: X \to \tilde{\mathcal{C}}_K^{\star}$, the following statements are equivalent:

- (i) f is convex and lsc,
- (ii) gr $f \subseteq X \times \mathbb{R}^p$ is convex and closed,
- (iii) epi $f \subseteq X \times \hat{\mathcal{C}}^{\star}$ is convex and closed,
- (iv) For all $y^* \in \operatorname{ri} K^\circ$ the function

 $\bar{f}_{y^*}: X \to \mathbb{R} \cup \{-\infty, +\infty\}, \quad \bar{f}_{y^*}(x) := -\delta^* \left(y^* | f(x) \right)$

is convex and closed.

Proof. (i) \Leftrightarrow (ii) \Leftrightarrow (iii). The equivalence of the convexity assertions follows from Proposition 3.4.2 and Proposition 3.4.3. The equivalence of the lower semi-continuity and closedness assertions follows similarly to the proof of Corollary 3.3.10 (using Corollary 3.4.6 (i) instead of Theorem 3.3.4).

(i) \Leftrightarrow (iv). From Corollary 1.3.4 (convexity) and Theorem 3.2.8 taking into account Corollary 3.4.6 (iii).

Theorem 3.4.8 Let $f: X \to \hat{\mathcal{C}}^*$ be convex. Then the following assertions hold true:

- (i) f is lsc at every $\bar{x} \in \operatorname{ri}(\operatorname{dom} f)$,
- (ii) f is continuous at every $\bar{x} \in int (dom f)$.

Proof. (i) Let $\bar{x} \in \operatorname{ri}(\operatorname{dom} f)$ be given and let $K := 0^+ f(\bar{x})$. By Theorem 3.2.8, it remains to show that, for all $y^* \in \operatorname{ri} K^\circ$, \bar{f}_{y^*} is lsc at \bar{x} . From Proposition 3.4.4 we deduce that $0^+ f(x) = K$ for all $x \in \operatorname{ri}(\operatorname{dom} f)$. Hence, for all $y^* \in \operatorname{ri} K^\circ$ it is true that $\bar{x} \in \operatorname{ri} \operatorname{dom} \bar{f}_{y^*}$, whence, by [1, Theorem 7.4], \bar{f}_{y^*} is lsc at \bar{x} .

(ii) By [68, Theorem 5.9 (b)], f is isc at $\bar{x} \in int (\text{dom } f)$. Hence, by Proposition 3.2.7, f is usc at \bar{x} . Now the assertion follows from (i).

We close this section with some assertions with respect to local boundedness of convex functions. These statements are not used in the following, however, they illuminate some additional connections between the previous assertions and known results for set–valued maps and could be of independent interest.

Theorem 3.4.9 Let $f : X \to \hat{\mathcal{C}}^*$ be convex and lsc. Then, f is locally bounded at every $\bar{x} \in \text{dom } f$.

Proof. Let $\bar{x} \in \text{dom } f$, $V := \{\bar{x}\} + \mathbb{B}$ and $K := 0^+ f(\bar{x})$. By Proposition 3.2.13 (v) and Corollary 3.4.6 (iii) we have $0^+ f(x) = K$ for all $x \in \text{dom } f$. Hence, condition (ii) in the definition of the local boundedness is satisfied. It remains to show $0^+ \inf_{x \in V} f(x) \subseteq K$. First we show that

$$\inf_{x \in V} f(x) = \bigcup_{x \in V} f(x).$$
(3.4)

We have to show that $\bigcup_{x \in V} f(x)$ is convex and closed. Since V and f are convex, the set $\bigcup_{x \in V} f(x)$ is convex. Since V is compact and gr f is closed, we deduce that $\bigcup_{x \in V} f(x)$ is closed. From (3.4) it follows that

$$0^{+} \inf_{x \in V} f(x) = 0^{+} \bigcup_{x \in V} f(x).$$

Let $k \in 0^+ \bigcup_{x \in V} f(x)$ be given. By [1, Theorem 8.2], k is the limit of a sequence $\{\lambda_n y_n\}$ where $\lambda_n \downarrow 0$ and $y_n \in \bigcup_{x \in V} f(x)$. Clearly, for all $n \in \mathbb{N}$ there exists $x_n \in V$ such that $y_n \in f(x_n)$. Since V is bounded, we have $(\lambda_n x_n, \lambda_n y_n) \to (0, k)$. Applying [1, Theorem 8.2] to the closed convex set gr $f \subseteq X \times \mathbb{R}^p$, we obtain $(0, k) \in 0^+$ gr f. With the aid of [1, Theorem 8.3] we deduce that $\bar{y} + \lambda k \in f(\bar{x} + \lambda \cdot 0) = f(\bar{x})$ for all $\lambda \ge 0$ and arbitrary $\bar{y} \in f(\bar{x})$, which is equivalent to $k \in 0^+ f(\bar{x}) = K$.

Corollary 3.4.10 If $f: X \to \hat{\mathcal{C}}^{\star}$ is convex, then f locally bounded at every $\bar{x} \in \operatorname{ri}(\operatorname{dom} f)$.

Proof. Theorem 3.4.9 yields that $\operatorname{lsc} f$ is locally bounded at every $x \in \operatorname{dom}(\operatorname{lsc} f)$. By Theorem 3.4.8 (i), we know that $f(\bar{x}) = (\operatorname{lsc} f)(\bar{x})$ for all $\bar{x} \in \operatorname{ri}(\operatorname{dom} f)$.

Corollary 3.4.11 Let $f: X \to \hat{\mathcal{C}}^{\star}$ be convex. Then, for all $\bar{x} \in X$ it holds

$$\limsup_{x \to \bar{x}} f(x) = \underset{x \to \bar{x}}{\text{LIM}} \inf_{x \to \bar{x}} f(x).$$

Proof. If $\bar{x} \in \operatorname{ri}(\operatorname{dom} f)$, this follows from Corollary 3.4.10 and Corollary 3.3.9. Otherwise, we have $\limsup_{x\to \bar{x}} f(x) = \operatorname{LIM}\operatorname{INF}_{x\to \bar{x}} f(x) = \emptyset$.

3.5 Conjugates

In this section, we introduce conjugates of $\hat{\mathcal{C}}$ -valued functions. We observe that a lot of properties being well-known for the classical conjugate notion remain valid. The lack of linearity with respect to the image space is compensated by the concept of oriented sets against the background of the considerations on embedding, see Section 1.2 and Section 1.3. We set $X = X^* = \mathbb{R}^n$ even though many assertions are also valid in more general spaces.

Definition 3.5.1 Let $f: X \to \hat{\mathcal{C}}$ and $c \in \mathbb{R}^p$. The function $f_c^*: X^* \to \hat{\mathcal{C}}$, defined by

$$f_c^*(x^*) := \sup_{x \in X} \left\{ \langle x^*, x \rangle \cdot \{c\} \boxminus f(x) \right\},\$$

where the set $\{c\}$ is considered to have the opposite orientation of f, is said to be the conjugate of f with respect to c. The function $f_c^{**}: X \to \hat{\mathcal{C}}$, defined by

$$f_c^{**}(x) := (f_c^*)_c^*,$$

is said to be the biconjugate of f with respect to c.

Clearly, f and f_c^* are contrarily oriented, but f and f_c^{**} have the same orientation. For instance, if $f: X \to \hat{\mathcal{C}}^*$ and $\{c\} \in \mathcal{C}^*$, we can express the biconjugate by using supremum oriented sets only:

$$f_{c}^{**}(x) = \sup_{x^{*} \in X^{*}} \left\{ \langle x, x^{*} \rangle \left\{ c \right\} + \inf_{x' \in X} \left\{ - \left\langle x^{*}, x' \right\rangle \left\{ c \right\} + f(x') \right\} \right\}.$$
(3.5)

As an easy consequence of the definition we obtain the Fenchel–Young inequality

$$\forall x \in X, \ x^* \in X^*, \ c \in \mathbb{R}^p: \ f_c^*(x^*) \ge \langle x^*, x \rangle \cdot \{c\} \boxminus f(x).$$

$$(3.6)$$

The following proposition collects some further properties of conjugates.

Proposition 3.5.2 Let $f, f_1, f_2 : X \to \hat{\mathcal{C}}$ (with the same orientation) be given. Then, for all $c \in \mathbb{R}^p$ the following statements hold true:

- (i) f_c^* is convex and lsc (even if f is not),
- (ii) f_c^{**} is convex and lsc,
- (iii) $f_1 \le f_2 \Rightarrow (f_1)_c^* \ge (f_2)_c^*$,
- (iv) $f_1 \le f_2 \Rightarrow (f_1)_c^{**} \le (f_2)_c^{**}$,

3.5. Conjugates

(v) $f_c^{**} \leq f$,

(vi) $f_c^{**} = f_{-c}^{**}$.

Proof. (i) and (ii). Convexity. For all $\lambda \in [0, 1]$ and all $x_1^*, x_2^* \in X^*$ it holds

$$\begin{aligned} f_c^* \left(\lambda x_1^* + (1 - \lambda) x_2^* \right) \\ &= \sup_{x \in X} \left\{ \left\langle \lambda x_1^* + (1 - \lambda) x_2^*, x \right\rangle \{c\} \boxminus f(x) \right\} \\ &= \sup_{x \in X} \left\{ \lambda \left(\left\langle x_1^*, x \right\rangle \{c\} \boxminus f(x) \right) \oplus (1 - \lambda) \left(\left\langle x_2^*, x \right\rangle \{c\} \boxminus f(x) \right) \right\} \right\} \\ &\leq \sup_{x_1, x_2 \in X} \left\{ \lambda \left(\left\langle x_1^*, x_1 \right\rangle \{c\} \boxminus f(x_1) \right) \oplus (1 - \lambda) \left(\left\langle x_2^*, x_2 \right\rangle \{c\} \boxminus f(x_2) \right) \right\} \\ &\stackrel{(1.1)}{\leq} \sup_{x_1 \in X} \left\{ \lambda \left(\left\langle x_1^*, x_1 \right\rangle \{c\} \boxminus f(x_1) \right) \right\} \oplus \sup_{x_2 \in X} \left\{ (1 - \lambda) \left(\left\langle x_2^*, x_2 \right\rangle \{c\} \boxminus f(x_2) \right) \right\} \\ &= \lambda f_c^* (x_1^*) \oplus (1 - \lambda) f_c^* (x_2^*). \end{aligned}$$

Lower semi-continuity. For all $\bar{x}^* \in X^*$ it holds

$$\lim_{x^* \to \bar{x}^*} f_c^*(x^*) = \inf_{\substack{x_n^* \to \bar{x}^* \\ \geq \\ Pr. 2.2.8 \text{ (iii)} \\ \geq \\ Pr. 2.2.8 \text{ (iii)} \\ \geq \\ Pr. 2.2.8 \text{ (iii)} \\ = \\ pr. 2.2.8 \text{ (iii)} \\ Pr. 2.2.8 \text{ (iii)} \\ = \\ pr. 2.2.8 \text{ (iii)} \\ Pr. 2.2.14 \text{ (iii)} \\$$

(iii) and (iv). Since $f_1 \leq f_2$ implies $\Box f_1 \geq \Box f_2$, this is obvious.

(v) From (3.5) we deduce

$$f_{c}^{**}(x) = \sup_{x^{*} \in X^{*}} \left\{ \langle x, x^{*} \rangle \left\{ c \right\} + \inf_{x' \in X} \left\{ - \left\langle x^{*}, x' \right\rangle \left\{ c \right\} + f(x') \right\} \right\}$$

$$\leq \sup_{x^{*} \in X^{*}} \left\{ \langle x, x^{*} \rangle \left\{ c \right\} - \left\langle x^{*}, x \right\rangle \left\{ c \right\} + f(x) \right\} = f(x).$$

(vi) Replace $\langle x, x^* \rangle \{c\}$ and $\langle x^*, x' \rangle \{c\}$ in formula (3.5), respectively, by $\langle x, -x^* \rangle \{-c\}$ and $\langle -x^*, x' \rangle \{-c\}$.

Example 3.5.3 (set-valued support function) The following set-valued map is often used in the literature. Given a subset $S \subseteq X$, the *set-valued indicator function* is defined by

$$\Delta(x \mid S) := \begin{cases} \{0\} & \text{if} \quad x \in S \\ \emptyset & \text{else.} \end{cases}$$

If $\Delta(\cdot | S)$ is considered to be a function with values in $\hat{\mathcal{C}}^{\star}$, $\Delta(\cdot | S)$ is convex if and only if the set S is likewise. The conjugate of $\Delta(\cdot | S) : X \to \hat{\mathcal{C}}^{\star}$ with respect to $c \in \mathbb{R}^p$ is the function

Chapter 3. Functions with values in $\widehat{\mathcal{C}}$

 $\Delta_c^*(\,\cdot\,|S):X^*\to\mathcal{C}^\diamond,$

$$\Delta_c^*(x^*|S) = \sup_{x \in S} \left\{ \langle x^*, x \rangle \left\{ c \right\} \right\} = \left[-\delta^*(x^*, -S), \delta^*(x^*, S) \right] \cdot \left\{ c \right\},$$

where $\{c\} \in \mathcal{C}^{\diamond}$ and, by convention,

$$\begin{aligned} \forall \alpha \in \mathbb{R} : \quad [-\infty, \alpha] \{c\} &:= (-\infty, \alpha] \{c\} = \bigcup_{\lambda \le \alpha} \left(\lambda \{c\}\right), \\ \forall \alpha \in \mathbb{R} : \quad [\alpha, +\infty] \{c\} &:= [\alpha, +\infty) \{c\} = \bigcup_{\lambda \ge \alpha} \left(\lambda \{c\}\right), \\ [-\infty, +\infty] \{c\} &:= (-\infty, +\infty) \{c\} = \mathbb{R} \cdot \{c\} = \bigcup_{\lambda \in \mathbb{R}} \left(\lambda \{c\}\right), \\ [+\infty, -\infty] \{c\} &:= \emptyset \quad \text{(this case occurs if } S = \emptyset). \end{aligned}$$

In order to emphasize the analogy to the scalar case, $\Delta_c^*(\cdot | S)$ is called the *(set-valued) support* function (with respect to $c \in \mathbb{R}^p$) of f. If S is closed and convex, one can easily verify (using [1, Theorem 13.1]) that the biconjugate $\Delta_c^{**}(\cdot | S)$ is equal to $\Delta(\cdot | S)$. Of course, this is also a consequence of the set-valued biconjugation theory to be developed in the next section.

Remark 3.5.4 An axiomatic approach to duality and conjugation was given by Martínez– Legaz and Singer [62]. Recently, this theory was extended up to the framework of complete lattices, see [26], which is general enough in order to apply these results to optimization with set relations. The spaces \hat{C}^* and \hat{C}° can be considered as two complete lattices. The change of orientation corresponds to the bijection s in [26]. In axiomatic duality a so-called "condition inf-d" plays a crucial role. In our framework this is condition is equivalent to the requirement that (1.1) is satisfied with equality. Another observation is that in [26, Theorem 4.1] coupling functions with a special form are considered, which corresponds to the form of our coupling functions, namely $\langle \cdot, \cdot \rangle \{c\}$.

3.6 Biconjugation theorem

In this section, we prove a biconjugation theorem for functions $f : X \to \hat{\mathcal{C}}^*$. In the next chapter, this result is used to prove strong duality assertions for convex optimization problems based on set relations. As in the previous section, we set $X = X^* = \mathbb{R}^n$. We begin with an auxiliary assertion.

Lemma 3.6.1 Let $K \subseteq \mathbb{R}^p$ be a nonempty closed convex cone which is not a linear subspace of \mathbb{R}^p . Then it holds

$$(y \in \operatorname{ri} K \wedge y^* \in \operatorname{ri} K^\circ) \Rightarrow \langle y, y^* \rangle < 0.$$

Proof. Assume the contrary. By the definition of the polar cone, this means that there exists $\bar{y} \in \operatorname{ri} K$ and $\bar{y}^* \in \operatorname{ri} K^\circ$ such that $\langle \bar{y}, \bar{y}^* \rangle = 0$. We show that

$$\forall y^* \in K^\circ : \quad \langle \bar{y}, y^* \rangle = 0. \tag{3.7}$$

Assuming that (3.7) is not true, we find some $\tilde{y}^* \in K^\circ$ such that $\langle \bar{y}, \tilde{y}^* \rangle < 0$. Since $\bar{y}^* \in \operatorname{ri} K^\circ$, there exists some $\mu > 1$ such that $\hat{y}^* := \mu \bar{y}^* + (1 - \mu) \tilde{y}^* \in K^\circ$. Hence $\langle \bar{y}, \hat{y}^* \rangle > 0$, which contradicts $\hat{y}^* \in K^\circ$.

With the aid of (3.7) and the bipolar theorem [1, Theorem 14.1] we obtain $-\bar{y} \in K^{\circ\circ} = K$. Since K is a convex cone and $\bar{y} \in K \cap (-K)$, we have $K = K + \{-\bar{y}\}$. From $\bar{y} \in \operatorname{ri} K$ we conclude $0 \in \operatorname{ri} K + \{-\bar{y}\} = \operatorname{ri} (K + \{-\bar{y}\}) = \operatorname{ri} K$. This implies $\lim K = \operatorname{aff} K = K$, i.e. K is a linear subspace of \mathbb{R}^p , a contradiction.

In order to formulate the biconjugation theorem, the vector $c \in \mathbb{R}^p$, which is involved in the definition of the biconjugate, has to be chosen appropriately. Therefore we define the set C_f , which depends on the function $f: X \to \hat{C}^*$, as follows:

$$K_{f} := \inf_{x \in \text{dom}\,f} 0^{+} f(x),$$

$$C_{f} := \begin{cases} \operatorname{ri} K_{f} \cup -\operatorname{ri} K_{f} & \text{if} \quad K_{f} \subsetneq \mathbb{R}^{p} \text{ is a not linear subspace of } \mathbb{R}^{p} \text{ or } K_{f} = \mathbb{R}^{p} \\ \mathbb{R}^{p} \setminus K_{f} & \text{if} \quad K_{f} \subsetneq \mathbb{R}^{p} \text{ is a linear subspace of } \mathbb{R}^{p} \text{ or } K_{f} = \emptyset. \end{cases}$$

Let us enumerate some basic properties of the sets K_f and C_f . These statements easily follow from the definition and some results of Section 3.4.

- (i) C_f is nonempty,
- (ii) If $f(\cdot) \equiv \emptyset$, then $C_f = \mathbb{R}^p$,
- (iii) If $f(x) = \mathbb{R}^p$ for some $x \in X$, then $C_f = \mathbb{R}^p$ (this case does not occur in Theorem 3.6.2, but in Corollary 3.6.3),
- (iv) If $f: X \to \hat{\mathcal{C}}_K^{\star}$, then $K = K_f$,
- (v) If f is convex, then $K_f = 0^+ f(x)$ for all $x \in \operatorname{ri} \operatorname{dom} f$,
- (vi) If f is convex and lsc, then $K_f = 0^+ f(x)$ for all $x \in \text{dom } f$.

It follows the main result of this section and one of the main results of this work.

Theorem 3.6.2 (Biconjugation theorem) For all functions $f : X \to \hat{C}^* \setminus \{\mathbb{R}^p\}$ the following statements are equivalent:

- (i) f is convex and lsc,
- (ii) For all $c \in C_f$ it holds $f = f_c^{**}$,
- (iii) There exists $c \in \mathbb{R}^p$ such that $f = f_c^{**}$.

Proof. If f is identically \emptyset we directly conclude that $f = f_c^{**}$ for all $c \in C_f = \mathbb{R}^p$ and, of course, f is convex and lsc. Therefore, dom f is assumed to be nonempty, and hence $K_f \neq \emptyset$.

(i) \Rightarrow (ii): From Corollary 3.4.6 (iii) and Proposition 3.2.13 (v) we deduce that $x \mapsto 0^+ f(x)$ is constant on dom f, i.e., setting $K := K_f$, we have $f : X \to \hat{\mathcal{C}}_K^{\star}$. From Corollary 3.4.7 we deduce that for all $y^* \in \operatorname{ri} K^\circ$ the function $\bar{f}_{y^*} : X \to \mathbb{R} \cup \{-\infty, +\infty\}, \quad \bar{f}_{y^*}(x) := -\delta^*(y^*|f(x))$ is convex and closed. By a classical biconjugation theorem, e.g. [1, Theorem 12.2], it follows that

$$\forall y^* \in \operatorname{ri} K^\circ : \qquad \bar{f}_{y^*} = \bar{f}_{y^*}^{**}. \tag{3.8}$$

Let $x \in X$ and $c \in C_f$ be arbitrarily chosen. In dependence on $K \neq \mathbb{R}^p$, we distinguish between two cases.

(A) Let $K \subsetneq \mathbb{R}^p$ be a not a linear subspace of \mathbb{R}^p . In view of Proposition 3.5.2 (vi) we can assume that $c \in \mathrm{ri} K$. For arbitrarily given $y^* \in \mathrm{ri} K^\circ$, Lemma 3.6.1 yields that $\langle y^*, c \rangle < 0$. Hence there exists $\alpha_{y^*} > 0$ such that $\langle \alpha_{y^*} y^*, c \rangle = -1$. This can be rewritten as

$$\forall t \in \mathbb{R}: \quad -\delta^* \left(\alpha_{y^*} y^* | \{t \cdot c\} \right) = - \left\langle \alpha_{y^*} y^*, t \cdot c \right\rangle = t.$$
(3.9)

For $\alpha := \alpha_{y^*} > 0$ we have

$$\begin{split} \alpha \cdot \left(-\delta^{*} \left(y^{*} | f(x) \right) \right) &= -\delta^{*} \left(\alpha y^{*} | f(x) \right) &= \bar{f}_{\alpha y^{*}}(x) \stackrel{(3.8)}{=} \bar{f}_{\alpha y^{*}}^{**}(x) \\ &= \sup_{x^{*} \in X^{*}} \left\{ \langle x, x^{*} \rangle + \inf_{x' \in X} \left\{ -\langle x^{*}, x' \rangle + \bar{f}_{\alpha y^{*}}(x') \right\} \right\} \\ \stackrel{(3.9)}{=} \sup_{x^{*} \in X^{*}} \left\{ -\delta^{*} \left(\alpha y^{*} | \langle x, x^{*} \rangle \left\{ c \right\} \right) + \inf_{x' \in X} \left\{ -\delta^{*} \left(\alpha y^{*} | - \langle x^{*}, x' \rangle \left\{ c \right\} \right) - \delta^{*} \left(\alpha y^{*} | f(x') \right) \right\} \right\} \\ &= \sup_{x^{*} \in X^{*}} \left\{ -\delta^{*} \left(\alpha y^{*} | \langle x, x^{*} \rangle \left\{ c \right\} \right) + \inf_{x' \in X} \left\{ -\delta^{*} \left(\alpha y^{*} | - \langle x^{*}, x' \rangle \left\{ c \right\} + f(x') \right\} \right) \right\} \\ \stackrel{\text{Pr. 1.3.7}}{=} \sup_{x^{*} \in X^{*}} \left\{ -\delta^{*} \left(\alpha y^{*} | \langle x, x^{*} \rangle \left\{ c \right\} \right) - \delta^{*} \left(\alpha y^{*} | \inf_{x' \in X} \left\{ -\langle x^{*}, x' \rangle \left\{ c \right\} + f(x') \right\} \right) \right\} \\ &= \sup_{x^{*} \in X^{*}} \left\{ -\delta^{*} \left(\alpha y^{*} | \langle x, x^{*} \rangle \left\{ c \right\} + \inf_{x' \in X} \left\{ -\langle x^{*}, x' \rangle \left\{ c \right\} + f(x') \right\} \right) \right\} \\ \stackrel{\text{Pr. 1.3.7}}{\leq} -\delta^{*} \left(\alpha y^{*} | \sup_{x^{*} \in X^{*}} \left\{ \langle x, x^{*} \rangle \left\{ c \right\} + \inf_{x' \in X} \left\{ -\langle x^{*}, x' \rangle \left\{ c \right\} + f(x') \right\} \right\} \right) \\ \stackrel{\text{(3.5)}}{=} -\delta^{*} \left(\alpha y^{*} | f_{c}^{**}(x) \right) = \alpha \cdot \left(-\delta^{*} \left(y^{*} | f_{c}^{**}(x) \right) \right). \end{split}$$

It follows that $\delta^*(y^*|f_c^{**}(x)) \leq \delta^*(y^*|f(x))$ for all $y^* \in \operatorname{ri} K^\circ$. If $f(x) = \emptyset$ we conclude that $\delta^*(y^*|f_c^{**}(x)) = -\infty$ for some y^* and hence $f_c^{**}(x) = \emptyset$. If $x \in \operatorname{dom} f$, we have $0^+f(x) = K$. Hence, by Lemma 1.3.1 (iii) we deduce that $f(x) \leq f_c^{**}(x)$. Finally, 3.5.2 (v) yields $f(x) = f_c^{**}(x)$.

(B) Let $K \subsetneq \mathbb{R}^p$ be a linear subspace of \mathbb{R}^p and let $c \in \mathbb{R}^p \setminus K$. Letting $\{c\} \in \mathcal{C}^*$, we define the set $B := \mathbb{R}_+ \{c\} \in \mathcal{C}^*$. Consider the function $\tilde{f} : X \to \hat{\mathcal{C}}^*$, $\tilde{f}(x) := f(x) + B$ (by [1, Corollary 9.1.2] this sum is closed). Of course, \tilde{f} is convex and lsc and $\tilde{f}(x) \neq \mathbb{R}^p$. For $x \in \text{dom } f$, [1, Corollary 9.1.2] yields $\tilde{K} := 0^+ \tilde{f}(x) = 0^+ f(x) + B = K + B$. In particular, \tilde{K} is not a linear space and $c \in \text{ri } \tilde{K}$. Therefore, \tilde{f} satisfies the assumptions of case (A). For arbitrary $x \in X$ it

3.6. Biconjugation theorem

follows

$$\begin{aligned} f(x) + B &= \tilde{f}(x) \stackrel{\text{part (A)}}{=} \tilde{f}_c^{**}(x) \\ &= \sup_{\substack{x^* \in X^*}} \left\{ \langle x, x^* \rangle \left\{ c \right\} + \inf_{\substack{x' \in X}} \left\{ - \left\langle x^*, x' \right\rangle \left\{ c \right\} + f(x') + B \right\} \right\} \\ &\stackrel{\text{Pr. 1.2.4, (1.1)}}{\leq} f_c^{**}(x) \oplus B &\leq f_c^{**}(x). \end{aligned}$$

By the same arguments (replace c by -c) we obtain $f(x) - B \leq f_{-c}^{**}(x) = f_c^{**}(x)$. Together we have $(f(x)-B)\cap(f(x)+B) \leq f_c^{**}(x)$. It remains to show that $f(x) \leq (f(x)-B)\cap(f(x)+B)$. Let $y \in (f(x)+B)\cap(f(x)-B)$ be given (the case $f(x) = \emptyset$ is immediate). This means $y = y_1 + r_1c = y_2 - r_2c$ for $y_1, y_2 \in f(x)$ and real numbers $r_1, r_2 \geq 0$. If $r_1 + r_2 = 0$ there is nothing to prove. For $r_1 + r_2 > 0$ it follows

$$y = \frac{r_2}{r_1 + r_2}(y_1 + r_1c) + \frac{r_1}{r_1 + r_2}(y_2 - r_2c) = \frac{r_2}{r_1 + r_2}y_1 + \frac{r_1}{r_1 + r_2}y_2 \in f(x).$$

Hence $f(x) \leq f_c^{**}(x)$ and Proposition 3.5.2 (v) yields equality.

- (ii) \Rightarrow (iii): Since C_f is always nonempty, this is obvious.
- (iii) \Rightarrow (i): Follows from Proposition 3.5.2 (ii).

It follows a local variant of the biconjugation theorem. In this assertion, the assumption $f(x) \neq \mathbb{R}^p$ can be omitted.

Corollary 3.6.3 Let $f: X \to \hat{\mathcal{C}}^*$ be convex and let $\bar{x} \in \text{dom } f$. Then, the following statements are equivalent:

- (i) f is lsc at $\bar{x} \in X$,
- (ii) For all $c \in C_f$ it holds $f(\bar{x}) = f_c^{**}(\bar{x})$,
- (iii) There exists $c \in \mathbb{R}^p$ such that $f(\bar{x}) = f_c^{**}(\bar{x})$.

Proof. (i) \Rightarrow (ii). If $f(\bar{x}) = \mathbb{R}^p$, Proposition 3.5.2 (v) yields that $f(\bar{x}) = f_c^{**}(\bar{x}) = \mathbb{R}^p$ for all $c \in C_f$ ($= \mathbb{R}^p$). Therefore let $f(\bar{x}) \neq \mathbb{R}^p$. By Corollary 3.4.6 and Proposition 3.2.13 (iii), (lsc f) : $X \to \hat{C}^*$ is convex, lsc and \hat{C}_K^* -valued for some nonempty closed convex cone $K \subseteq \mathbb{R}^p$. Since f is lsc at $\bar{x} \in \text{dom } f$ we obtain $K = 0^+ f(\bar{x}) \neq \mathbb{R}^p$. Hence (lsc f)(x) $\neq \mathbb{R}^p$ for all $x \in X$. Theorem 3.6.2 yields (lsc f) = (lsc f) $_c^{**}$ for all $c \in C_{(\text{lsc } f)}$. Using Proposition 3.2.13 (iv) and Proposition 3.5.2 (iv) we conclude that $f(\bar{x}) = (\text{lsc } f)(\bar{x}) = (\text{lsc } f)_c^{**}(\bar{x}) \leq f_c^{**}(\bar{x})$. Proposition 3.5.2 (v) yields that $f(\bar{x}) = f_c^{**}(\bar{x})$ for all $c \in C_{(\text{lsc } f)}$. It remains to show that $C_{(\text{lsc } f)} = C_f$. Therefore, we show that $K = K_f$ ($= \inf_{x \in \text{dom } f} 0^+ f(x)$). Since $K = 0^+ f(\bar{x})$ we have $K \geq K_f$. From (lsc f) $\leq f$ we conclude that $K = 0^+(\text{lsc } f)(x) \leq 0^+ f(x)$ for all $x \in \text{dom } f$. Hence $K \leq K_f$.

(ii) \Rightarrow (iii). Since C_f is always nonempty, this is obvious.

(iii) \Rightarrow (i). By Proposition 3.5.2 (ii), f_c^{**} is lsc. From $f_c^{**} \leq f$ we obtain $\liminf_{x \to \bar{x}} f_c^{**}(x) \leq \liminf_{x \to \bar{x}} f(x)$. Hence $f(\bar{x}) = f_c^{**}(\bar{x}) \leq \liminf_{x \to \bar{x}} f_c^{**}(x) \leq \liminf_{x \to \bar{x}} f(x)$. \Box

Finally, we express the biconjugation theorem by conventional notations. The set C_f is defined as above, but in the definition of K_f , $0^+ f(x)$ is replaced by $0^+(\operatorname{cl} \operatorname{conv} f(x))$ (in order to have a correct definition of C_f for a set-valued map $f : \mathbb{R}^n \rightrightarrows \mathbb{R}^p$, which has not closed convex values a priori).

Corollary 3.6.4 Let $f : \mathbb{R}^n \rightrightarrows \mathbb{R}^p$ be a set-valued map with nonempty graph and $f(x) \neq \mathbb{R}^p$ for all $x \in X$. Then, the following statements are equivalent:

- (i) $\operatorname{gr} f$ is closed and convex,
- (ii) For all $c \in C_f$ it holds

$$\forall x \in \mathbb{R}^n : f(x) = \bigcap_{x^* \in \mathbb{R}^n} \left\{ \langle x, x^* \rangle \{c\} - \operatorname{cl\,conv} \bigcup_{x' \in \mathbb{R}^n} \left\{ \langle x^*, x' \rangle \{c\} - f(x') \right\} \right\},\$$

(iii) There exists $c \in \mathbb{R}^p$ such that f can be expressed as in (ii).

Proof. Follows from Theorem 3.6.2 and the results of Section 3.4.

Chapter 4

Duality

In this chapter, we investigate optimization problems with $\hat{\mathcal{C}}$ -valued objective function. Given an objective function $f: X \to \hat{\mathcal{C}}$ and a set $S \subseteq X$ of feasible points, we are interested in the sets $\inf_{x \in S} f(x)$ and $\sup_{x \in S} f(x)$.

We already know that these problems are not of the same type, because one of them involves the union and the other one the intersection. In this chapter, we prove weak as well as strong duality assertions. In order to obtain strong duality it is essential to start either with an infimum problem for \hat{C}^* -valued functions or with a supremum problem for \hat{C}° -valued functions. Otherwise, it seems to be not possible to obtain comparable duality assertions, see Remark 4.1.4 below. However, the other type of problem occurs as the dual problem. Exemplary, we consider primal problems with \hat{C}^* -valued functions.

The main tool for proving strong duality assertions is the biconjugation theorem developed in Section 3.6. We consider a Fenchel duality approach as well as a Lagrange duality approach. The problem formulation as well as the proofs are very analogous to the well-known scalar case. In the last section of this chapter, we calculate some special cases.

4.1 Fenchel duality

In this section, we prove weak and strong duality assertions for optimization problems with \hat{C}^{\star} -valued objective function. We speak about Fenchel duality because our duality theorem has the same structure as the classical Fenchel duality theorem for extended real-valued functions, for instance, see [1, Theorem 31.1] or [10, Theorem 3.3.5]. This means, for instance, that the dual problem of a given problem of the form $\inf_{x \in X} \{f(x) - g(x)\}$ (or similar) is expressed by the conjugates of f and g. In contrast to that, this structure gets lost in many other generalizations of the classical Fenchel duality, for instance, see [78]. In this section, we set $X = X^* = \mathbb{R}^n$ and $U = U^* = \mathbb{R}^m$, because our main attention is drawn to the image space. Nevertheless, an extension to more general spaces should be possible. As in the scalar theory, the constraint qualification has to be strengthened in case of infinite dimensional X and U.

For given functions $f : X \to \hat{\mathcal{C}}^*$ and $g : U \to \hat{\mathcal{C}}^*$, a linear map $A : X \to U$ and a vector $c \in \mathbb{R}^p$, let

 $p: X \to \hat{\mathcal{C}}^{\star}$ and $d_c: U^* \to \hat{\mathcal{C}}^{\star}$

be defined, respectively, by

$$p(x) = f(x) \oplus g(Ax)$$
 and $d_c(u^*) = \boxminus \left(f_c^*(A^*u^*) \oplus g_c^*(-u^*)\right).$

We consider the following optimization problems, the primal problem

(P)
$$P := \inf_{x \in X} p(x),$$

and the dual problem associated to (P)

$$(\mathbf{D}_{\mathbf{c}}) \qquad \qquad D_{c} := \sup_{u^* \in U^*} d_c(u^*).$$

Analogous to the scalar optimization theory, we introduce the value function by

$$v: U \to \hat{\mathcal{C}}^{\star}, \qquad v(u) := \inf_{x \in X} \big\{ f(x) \oplus g(Ax - u) \big\}.$$

The following proposition collects some properties of the value function.

Proposition 4.1.1 The value function $v: U \to \hat{\mathcal{C}}^*$ has the following properties:

- (i) If f and g are convex, then v is convex,
- (ii) v(0) = P,
- (iii) $\forall c \in \mathbb{R}^p, \ \forall u^* \in U^*: \quad v_c^*(u^*) = \boxminus d_c(u^*),$
- (iv) $\forall c \in \mathbb{R}^p : v_c^{**}(0) = D_c.$

Proof. (i) For arbitrary $u_1, u_2 \in U$ and $\lambda \in (0, 1)$ it holds

$$\begin{split} \lambda v(u_1) \oplus (1-\lambda) v(u_2) \\ &= \lambda \inf_{x_1 \in X} \left\{ f(x_1) \oplus g(Ax_1 - u_1) \right\} \oplus (1-\lambda) \inf_{x_2 \in X} \left\{ f(x_2) \oplus g(Ax_2 - u_2) \right\} \\ \stackrel{\text{Pr. 1.2.4}}{=} \inf_{x_1, x_2 \in X} \left\{ \lambda f(x_1) \oplus \lambda g(Ax_1 - u_1) f(x_1) \oplus (1-\lambda) f(x_2) \oplus (1-\lambda) g(Ax_2 - u_2) \right\} \\ \stackrel{f, \ g \ \text{convex}}{\geq} \inf_{x_1, x_2 \in X} \left\{ f\left(\lambda x_1 + (1-\lambda) x_2\right) \oplus g\left(A\left(\lambda x_1 + (1-\lambda) x_2\right) - (\lambda u_1 + (1-\lambda) u_2)\right) \right\} \\ &= \inf_{x \in X} \left\{ f\left(x\right) \oplus g\left(Ax - (\lambda u_1 + (1-\lambda) u_2)\right) \right\} = v(\lambda u_1 + (1-\lambda) u_2). \end{split}$$

(ii) Obvious.

4.1. Fenchel duality

(iii) Let $c \in \mathbb{R}^p$ ($\{c\} \in \mathcal{C}^*$) and $u^* \in U^*$ be given. Then we have

$$\begin{split} v_{c}^{*}(u^{*}) &= \sup_{u \in U} \left\{ \boxplus \langle u^{*}, u \rangle \{c\} \boxminus v(u) \right\} \\ \stackrel{(1.2)}{=} & \boxplus_{u \in U} \left\{ \langle -u^{*}, u \rangle \{c\} + v(u) \right\} \\ &= & \lim_{u \in U} \left\{ \langle -u^{*}, u \rangle \{c\} + \inf_{x \in X} \{f(x) \oplus g(Ax - u)\} \right\} \\ \stackrel{\text{Pr. 1.2.4}}{=} & \boxplus_{u \in U} \inf_{x \in X} \left\{ \langle -u^{*}, u \rangle \{c\} + f(x) \oplus g(Ax - u) \right\} \\ \stackrel{\text{Pr. A.4 (v)}}{=} & \boxplus_{x \in X} \inf_{u \in U} \left\{ \left(f(x) - \langle A^{*}u^{*}, x \rangle \{c\} \right) \oplus \left(g(Ax - u) - \langle -u^{*}, Ax - u \rangle \{c\} \right) \right\} \\ \stackrel{\text{Pr. 1.2.4}}{=} & \liminf_{x \in X} \inf_{u \in U} \left\{ f(x) - \langle A^{*}u^{*}, x \rangle \{c\} \right\} \oplus \inf_{u' \in U} \left\{ g(u') - \langle -u^{*}, u' \rangle \{c\} \right\} \\ &= & f_{c}^{*}(A^{*}u^{*}) \oplus g_{c}^{*}(-u^{*}) = \quad \boxminus d_{c}(u^{*}). \end{split}$$

(iv) It holds $v_c^{**}(0) = \sup_{u^* \in U^*} \boxminus v_c^*(u^*) \stackrel{\text{(iii)}}{=} \sup_{u^* \in U^*} d_c(u^*) = D_c.$

It follows the main result of this section. This assertion was announced in the form of Corollary 4.1.3 in [55] and is to be published in the present form in [56]. We present two different proofs, the first one makes use of the biconjugation theorem developed in Section 3.6 and the second one is a "direct proof" using a scalar duality theorem.

Theorem 4.1.2 (Duality theorem) The problems (P) and (D_c) (with arbitrary $c \in \mathbb{R}^p$) satisfy the weak duality inequality, i.e., $D_c \leq P$.

Furthermore, let f and g be convex, let

$$0 \in \operatorname{ri}\left(\operatorname{dom} g - A \operatorname{dom} f\right) \tag{4.1}$$

and, in dependence on $K := 0^+ P$, let the element $c \in \mathbb{R}^p$ be chosen as follows:

- (i) $c \in \operatorname{ri} K \cup -\operatorname{ri} K$, if $K \subsetneq \mathbb{R}^p$ is not a linear subspace of \mathbb{R}^p or $K = \mathbb{R}^p$,
- (ii) $c \in \mathbb{R}^p \setminus K$, if $K \subsetneq \mathbb{R}^p$ is a linear subspace of \mathbb{R}^p .

Then, we have strong duality, i.e., $D_c = P$.

Proof. Consider the value function $v: U \to \hat{\mathcal{C}}^*$ of problem (P). By Proposition 4.1.1 (ii), (iv), for all $c \in \mathbb{R}^p$ we have P = v(0) and $v_c^{**}(0) = D_c$. Since $v_c^{**} \leq v$, we obtain the weak duality inequality. From Proposition 4.1.1 (i) we conclude that v is convex. It is easy to check that dom v = dom g - Adom f. By virtue of (4.1), this implies $0 \in \text{ri dom } v$. Theorem 3.4.8 (i) yields that v is lsc at 0. By Proposition 3.4.4, we conclude that $K = 0^+P = 0^+v(0) = \inf_{u \in \text{dom } v} 0^+v(u)$, i.e., the choice of the vector c by (i) and (ii) ensures that c belongs to the set C_v , defined in Section 3.6. The local variant of the biconjugation theorem (Corollary 3.6.3) yields that $v(0) = v_c^{**}(0)$, i.e., we have strong duality.

We continue with a second proof of Theorem 4.1.2 wherein the strong duality assertion is obtained from a corresponding scalar strong duality result using the embedding procedure as in the proof of the biconjugation theorem. This proof is independent of the considerations about semi-continuity in Chapter 3 and it was already published in the author's paper [56].

Second proof of Theorem 4.1.2. The weak duality can be proven as above or by the Fenchel–Young inequality (3.6). The proof of the strong duality assertion is organized as follows. We start with case (i). Then we show that case (ii) is a consequence of case (i).

(i) In case of $K = \mathbb{R}^p$ there is nothing to prove because the strong duality immediately follows from the weak duality assertion. Therefore, let $K \subsetneq \mathbb{R}^p$ be not a linear subspace of \mathbb{R}^p . It is easy to verify that $D_c = D_{-c}$. Hence it suffices to consider the case $c \in \operatorname{ri} K$.

With the aid of Proposition 1.3.7 it follows that

$$\forall y^* \in \mathbb{R}^p : -\delta^*(y^*|P) = -\delta^*(y^*|\inf_{x \in X} p(x)) = \inf_{x \in X} \{-\delta^*(y^*|p(x))\}.$$

By the extended real-valued functions $\bar{f}_{y^*}: X \to \mathbb{R} \cup \{-\infty, +\infty\}$ and $\bar{g}_{y^*}: U \to \mathbb{R} \cup \{-\infty, +\infty\}$ being defined, respectively, by $\bar{f}_{y^*}(x) := -\delta^*(y^*|f(x))$ and $\bar{g}_{y^*}(u) := -\delta^*(y^*|g(u))$ this can be rewritten as a collection of scalar optimization problems

$$\forall y^* \in \mathbb{R}^p : -\delta^*(y^*|P) = \inf_{x \in X} \{ \bar{f}_{y^*}(x) + \bar{g}_{y^*}(Ax) \}.$$
(4.2)

The convexity of f and g implies the convexity of \bar{f}_{y^*} and \bar{g}_{y^*} , respectively. Clearly, we have dom $f = \operatorname{dom} \bar{f}_{y^*}$ and dom $g = \operatorname{dom} \bar{g}_{y^*}$. Hence, (4.1) implies that $0 \in \operatorname{ri}(\operatorname{dom} \bar{g}_{y^*} - A \operatorname{dom} \bar{f}_{y^*})$. A scalar duality theorem, for instance [10, Theorem 3.3.5], yields that

$$\forall y^* \in \mathbb{R}^p: \ -\delta^*(y^*|P) = \sup_{u^* \in U^*} \left\{ -\bar{f}_{y^*}^*(A^*u^*) - \bar{g}_{y^*}^*(-u^*) \right\}.$$
(4.3)

Let $y^* \in \operatorname{ri} K^\circ$ be arbitrarily given. Since $c \in \operatorname{ri} K$, Lemma 3.6.1 yields that $\langle y^*, c \rangle < 0$. Hence, there exists $\alpha_{y^*} > 0$ such that $\langle \alpha_{y^*} y^*, c \rangle = -1$. This can be rewritten as

$$\forall t \in \mathbb{R}: \quad -\delta^* \left(\alpha_{y^*} y^* | \{t \cdot c\} \right) = - \left\langle \alpha_{y^*} y^*, t \cdot c \right\rangle = t. \tag{4.4}$$

4.1. Fenchel duality

For $\alpha := \alpha_{y^*} > 0$ and letting $\{c\} \in \mathcal{C}^*$ we have

$$\begin{split} \alpha \cdot \left(-\delta^{*}\left(y^{*}|P\right)\right) &= -\delta^{*}\left(\alpha y^{*}|P\right) \stackrel{(4.3)}{=} \sup_{u^{*} \in U^{*}} \left\{-\bar{f}_{\alpha y^{*}}^{*}(A^{*}u^{*}) - \bar{g}_{\alpha y^{*}}^{*}(-u^{*})\right\} \\ &= \sup_{u^{*} \in U^{*}} \left\{\inf_{x \in X} \left\{-\langle A^{*}u^{*}, x \rangle + \bar{f}_{\alpha y^{*}}(x)\right\} + \inf_{u \in U} \left\{\langle u^{*}, u \rangle + \bar{g}_{\alpha y^{*}}(u)\right\}\right\} \\ \stackrel{(4.4)}{=} \sup_{u^{*} \in U^{*}} \left\{\inf_{x \in X} \left\{-\delta^{*}\left(\alpha y^{*}| - \langle A^{*}u^{*}, x \rangle \cdot \{c\}\right) - \delta^{*}\left(\alpha y^{*}|f(x)\right)\right\} \\ &+ \inf_{u \in U} \left\{-\delta^{*}\left(\alpha y^{*}| \langle u^{*}, u \rangle \cdot \{c\}\right) - \delta^{*}\left(\alpha y^{*}|g(u)\right)\right\}\right\} \\ &= \sup_{u^{*} \in U^{*}} \left\{\inf_{x \in X} \left\{-\delta^{*}\left(\alpha y^{*}| - \langle A^{*}u^{*}, x \rangle \{c\} + f(x)\right)\right\} + \inf_{u \in U} \left\{-\delta^{*}\left(\alpha y^{*}| \langle u^{*}, u \rangle \{c\} + g(u)\right)\right\}\right\} \\ &= \sup_{u^{*} \in U^{*}} \left\{-\delta^{*}\left(\alpha y^{*}| \inf_{x \in X} \left\{-\langle A^{*}u^{*}, x \rangle \{c\} + f(x)\right\}\right) - \delta^{*}\left(\alpha y^{*}| \inf_{u \in U} \left\{\langle u^{*}, u \rangle \{c\} + g(u)\right\}\right)\right\} \\ &= \sup_{u^{*} \in U^{*}} \left\{-\delta^{*}\left(\alpha y^{*}| \inf_{x \in X} \left\{-\langle A^{*}u^{*}, x \rangle \{c\} + f(x)\right\}\right\} \oplus \inf_{u \in U} \left\{\langle u^{*}, u \rangle \{c\} + g(u)\right\}\right)\right\} \\ &= -\delta^{*}\left(\alpha y^{*}| \sup_{u^{*} \in U^{*}} \left\{\inf_{x \in X} \left\{-\langle A^{*}u^{*}, x \rangle \{c\} + f(x)\right\}\right\} \oplus \inf_{u \in U} \left\{\langle u^{*}, u \rangle \{c\} + g(u)\right\}\right\} \\ &= -\delta^{*}\left(\alpha y^{*}| \sup_{u^{*} \in U^{*}} \left\{\inf_{x \in X} \left\{-\langle A^{*}u^{*}, x \rangle \{c\} + f(x)\right\}\right\right\} \\ &= -\delta^{*}\left(\alpha y^{*}| \sup_{u^{*} \in U^{*}} \left\{\inf_{x \in X} \left\{-\langle A^{*}u^{*}, x \rangle \{c\} + f(x)\right\}\right\right\} \\ &= -\delta^{*}\left(\alpha y^{*}| \sup_{u^{*} \in U^{*}} \left\{\inf_{x \in X} \left\{-\langle A^{*}u^{*}, x \rangle \{c\} + f(x)\right\}\right\right\} \\ &= -\delta^{*}\left(\alpha y^{*}| \sup_{u^{*} \in U^{*}} \left\{\inf_{x \in X} \left\{-\langle A^{*}u^{*}, x \rangle \{c\} + f(x)\right\}\right\right\} \\ &= -\delta^{*}\left(\alpha y^{*}| \sup_{u^{*} \in U^{*}} \left\{\inf_{x \in X} \left\{-\langle A^{*}u^{*}, x \rangle \{c\} + f(x)\right\}\right\} \\ &= -\delta^{*}\left(\alpha y^{*}| \sup_{u^{*} \in U^{*}} \left\{\inf_{x \in X} \left\{-\langle A^{*}u^{*}, x \rangle \{c\} + f(x)\right\}\right\right\} \\ &= -\delta^{*}\left(\alpha y^{*}| \sup_{u^{*} \in U^{*}} \left\{\inf_{x \in X} \left\{\inf_{x \in X} \left\{-\langle A^{*}u^{*}, x \rangle \{c\} + f(x)\right\}\right\right\}\right\} \\ &= -\delta^{*}\left(\alpha y^{*}| \sup_{u^{*} \in U^{*}} \left\{\inf_{x \in X} \left\{\inf_{x \in X} \left\{-\langle A^{*}u^{*}, x \rangle \{c\} + f(x)\right\}\right\right\} \\ &= -\delta^{*}\left(\alpha y^{*}| \sup_{u^{*} \in U^{*}} \left\{\inf_{x \in X} \left\{\inf_{u^{*} \in U^{*} \in U^{*}} \left\{\inf_{u^{*} \in U^{*} \inU^{*} \in U^{*} \in U^{*} \inU^{*} \inU^{*} \inU^{*} \inU^{*} \inU^{*} \inU^$$

It follows that $\delta^*(y^*|D_c) \leq \delta^*(y^*|P)$ for all $y^* \in \operatorname{ri} K^\circ$. Lemma 1.3.1 (iii) yields $P \supseteq D_c$. By the weak duality inequality we obtain $P = D_c$.

(ii) Let $K \subsetneq \mathbb{R}^p$ be a linear subspace of \mathbb{R}^p and let $c \in \mathbb{R}^p \setminus K$. Consider the set $B := \mathbb{R}_+ \{c\} \subseteq \mathcal{C}^*$. We define a new objective function by $\tilde{p} : X \to \hat{\mathcal{C}}^*$, $\tilde{p}(x) := p(x) \oplus B = f(x) \oplus (g(Ax) \oplus B)$. By Proposition 1.2.4, we have $\tilde{P} := \inf_{x \in X} \tilde{p}(x) = (\inf_{x \in X} p(x)) \oplus B = P \oplus B$. With the aid of [1, Corollary 9.1.2] we conclude that $\tilde{P} = P \oplus B = P + B$ and $\tilde{K} := 0^+ \tilde{P} = 0^+ P + B = K + B$. Clearly, \tilde{K} is not a linear space and $c \in \operatorname{ri} \tilde{K}$. It is an easy task to show that $\tilde{g} : U \to \hat{\mathcal{C}}^*$, $\tilde{g}(\cdot) := g(\cdot) \oplus B$ is convex and (4.1) remains true for the new problem, hence, we have strong duality by part (i) of this theorem. For the conjugate $\tilde{g}_c^* : U^* \to \hat{\mathcal{C}}^\diamond$ of \tilde{g} it holds

$$\tilde{g}_c^*(u^*) = \boxminus \inf_{u \in U} \left\{ - \langle u, u^* \rangle \left\{ c \right\} \oplus g(u) \oplus B \right\} \stackrel{\operatorname{Pr. } 1.2.4}{=} g_c^*(u^*) \oplus \boxminus B.$$

Hence, the dual objective function $\tilde{d}: U^* \to \hat{\mathcal{C}}^*$ for the problem $\inf_{x \in X} \tilde{p}(x)$ is given by

$$\tilde{d}_c(u^*) = \boxminus f_c^*(A^*u^*) \oplus \boxminus g_c^*(-u^*) \oplus B = d_c(u^*) \oplus B.$$

Since $0 \in B$ we deduce that $\tilde{d}_c \leq d_c$, hence $\tilde{D}_c := \sup_{u^* \in U^*} \tilde{d}_c(u^*) \leq D_c$. The strong duality assertion for the problem $\inf_{x \in X} \tilde{p}(x)$ yields $P + B = \tilde{P} \leq \tilde{D}_c \leq D_c$. Likewise (replace c by -c), it follows $P - B \leq D_{-c} = D_c$. Hence, $(P + B) \cap (P - B) \leq D_c$. As in the last part of the proof of Theorem 3.6.2 we can show that $P \leq (P + B) \cap (P - B)$, hence we have $P \leq D_c$. By the weak duality assertion it follows $P = D_c$.

We next express the preceeding theorem by conventional notations. Although the analogy to the scalar theory is more difficult to see, this form could be more convenient for possible applications. Moreover, we try to avoid the convex hull and closure operations if they are superfluous. Let $f : \mathbb{R}^n \rightrightarrows \mathbb{R}^p$ be a set-valued map. We say f has closed (convex) values if $f(x) \subseteq \mathbb{R}^p$ is closed (convex) for all $x \in \mathbb{R}^n$. Clearly, if f has a closed (convex) graph, then f has closed (convex) values. The opposite implication is not true, in general. The map fhas closed values and a convex graph if and only if f can be interpreted as a convex function $f : \mathbb{R}^n \to \hat{\mathcal{C}}^{\star}$.

Corollary 4.1.3 For given set-valued maps $f : \mathbb{R}^n \rightrightarrows \mathbb{R}^p$ and $g : \mathbb{R}^m \rightrightarrows \mathbb{R}^p$, a linear map $A : \mathbb{R}^n \rightarrow \mathbb{R}^m$ and a vector $c \in \mathbb{R}^p$, we have

$$\bigcup_{x \in \mathbb{R}^n} \left(f(x) + g(Ax) \right) \subseteq \bigcap_{u^* \in \mathbb{R}^m} \left(\bigcup_{x \in \mathbb{R}^n} \left(f(x) - \langle A^*u^*, x \rangle \left\{ c \right\} \right) + \bigcup_{u \in \mathbb{R}^m} \left(g(u) + \langle u^*, u \rangle \left\{ c \right\} \right) \right)$$

If f and g additionally have convex graphs and closed values and satisfy the condition $0 \in$ ri (dom g - A dom f) and, in dependence on $K := 0^+$ (cl conv $\bigcup_{x \in X} (f(x) + g(Ax))$), the vector $c \in \mathbb{R}^p$ is chosen as in Theorem 4.1.2, we have strong duality, i.e.,

$$\operatorname{cl}\bigcup_{x\in\mathbb{R}^n} (f(x) + g(Ax)) = \bigcap_{u^*\in\mathbb{R}^m} \operatorname{cl}\left(\bigcup_{x\in\mathbb{R}^n} (f(x) - \langle A^*u^*, x\rangle \{c\}) + \bigcup_{u\in\mathbb{R}^m} (g(u) + \langle u^*, u\rangle \{c\})\right).$$

Proof. For all $u^* \in \mathbb{R}^m$, we have

$$\bigcup_{x \in \mathbb{R}^n} \left(f(x) + g(Ax) \right) = \bigcup_{x \in \mathbb{R}^n} \left(f(x) - \langle A^* u^*, x \rangle \left\{ c \right\} + g(Ax) + \langle u^*, Ax \rangle \left\{ c \right\} \right)$$

$$\subseteq \bigcup_{x \in \mathbb{R}^n} \left(f(x) - \langle A^* u^*, x \rangle \left\{ c \right\} \right) + \bigcup_{u \in \mathbb{R}^m} \left(g(u) + \langle u^*, u \rangle \left\{ c \right\} \right).$$

Taking the intersection over all $u^* \in \mathbb{R}^m$ we obtain the weak duality inclusion.

Let f and g have convex graphs and closed values. This means f and g can be interpreted as convex functions $f : \mathbb{R}^n \to \hat{\mathcal{C}}^*$ and $g : \mathbb{R}^m \to \hat{\mathcal{C}}^*$. In the present case, we can show that

$$\inf_{x \in X} \left\{ f(x) \oplus g(Ax) \right\} = \operatorname{cl} \bigcup_{x \in X} \left(f(x) + g(Ax) \right).$$
(4.5)

Indeed, by elementary arguments we deduce that $P := \bigcup_{x \in X} \operatorname{cl} (f(x) + g(Ax))$ is convex. Furthermore, from Proposition A.2 (ii) we conclude that $\operatorname{cl} \bigcup_{x \in X} \operatorname{cl} p(x) = \operatorname{cl} \bigcup_{x \in X} p(x)$, where p(x) = f(x) + g(Ax). Together we obtain (4.5). By analogous arguments the right-hand side of the strong duality equality equals D_c in Theorem 4.1.2.

Remark 4.1.4 It seems to be not possible to obtain comparable duality assertions if we start with an infimum problem with a \hat{C}^{\diamond} -valued objective function. The reason for that seems to be that (1.1) is not satisfied with equality in this case. Equality in (1.1) is equivalent to the condition "inf-d", which plays an important role in the axiomatic duality theory by Getán, Martinez-Legaz and Singer [26].

4.2 Lagrange duality

This section is devoted to Lagrange duality of optimization problems with $\hat{\mathcal{C}}^{\star}$ -valued objective function and set-valued constraints. As before we set $X = X^* = \mathbb{R}^n$ and $U = U^* = \mathbb{R}^m$.

Let $f: X \to \hat{\mathcal{C}}^*$, let $G: X \rightrightarrows U$ be a set-valued map and let $C_U \subseteq U$ be a nonempty closed convex cone. We consider the following optimization problem:

(P)
$$P := \inf_{x \in S} f(x), \qquad S := \{ x \in X | G(x) \cap -C_U \neq \emptyset \}.$$

Constraints of this type have been considered by many authors such as [7], [20], [57], [32], [41], [22]. We next recall the notion of C-convexity. In the following, this concept is used to describe convexity with respect to the constraints.

Definition 4.2.1 (e.g. [41]) Let X, Y be real linear spaces and let $C \subseteq Y$ be a convex cone. A set-valued map $H: X \rightrightarrows Y$ is said to be C-convex if

$$\forall x_1, x_2 \in X, \ \forall \lambda \in [0,1]: \qquad H\big(\lambda x_1 + (1-\lambda)x_2\big) + C \supseteq \lambda H(x_1) + (1-\lambda)H(x_2).$$

The Lagrangian of the problem (P) (with respect to $c \in \mathbb{R}^p$) is defined by

$$L_c: X \times U^* \to \hat{\mathcal{C}}^*, \qquad L_c(x, u^*) := f(x) \oplus \Box \Delta_c^* \big(u^* \big| - C_U - G(x) \big),$$

where $\Delta_c^*(\cdot | -C_U - G(x)) : U^* \to \hat{\mathcal{C}}^\diamond$ is the set-valued support function (see Section 3.5) of the set $-C_U - G(x)$. Hence, the Lagrangian can be expressed (for $\{c\} \in \mathcal{C}^\star$) by

$$L_{c}(x, u^{*}) = f(x) \oplus \inf_{u \in G(x) + C_{U}} \left\{ \langle u^{*}, u \rangle \left\{ c \right\} \right\}.$$

Perhaps, this definition looks a little unusual. If f is an extended real-valued function and G a vector-valued function, our constraint reduces to $G(x) \in -C_U$ (or $G(x) \leq_{C_U} 0$) and the Lagrangian is usually defined by $L(x, u^*) := f(x) + \langle u^*, G(x) \rangle$, but only those $u^* \in U^*$ are involved into the theory that belong to $-C_U^\circ$. If we define the Lagrangian analogously to above by $L(x, u^*) := f(x) - \delta^*(u^* | -C_U - G(x))$ (using the convention $-\infty + \infty = +\infty$), we obtain

$$L(x, u^{*}) = f(x) + \langle u^{*}, G(x) \rangle - \delta^{*} (u^{*} | -C_{U}),$$

where the support function of $-C_U$ has the simple form

$$\delta^*(u^*| - C_U) = \begin{cases} 0 & \text{if } u^* \in -C_U^\circ \\ +\infty & \text{else.} \end{cases}$$

The set-valued support function, however, has a more complicated form, namely, letting L be the lineality space of C_U° and $\{c\} \in \mathcal{C}^{\diamond}$, it can be expressed by

$$\Delta_c^*(u^*| - C_U) = \begin{cases} \mathbb{R}_+ \cdot \{c\} & \text{if } u^* \in C_U^\circ \setminus L \\ \mathbb{R}_- \cdot \{c\} & \text{if } u^* \in -C_U^\circ \setminus L \\ 0 & \text{if } u^* \in L \\ \mathbb{R} \cdot \{c\} & \text{else.} \end{cases}$$

This means, we do not obtain a meaningful analogue to the Lagrangian in the classical form. By Proposition 1.2.4, the Lagrangian can be written as

$$L_{c}(x,u^{*}) = f(x) \oplus \inf_{u \in G(x)} \left\{ \langle u^{*}, u \rangle \left\{ c \right\} \right\} \oplus \inf_{u \in C_{U}} \left\{ \langle u^{*}, u \rangle \left\{ c \right\} \right\},$$

where $\{c\} \in \mathcal{C}^{\star}$. An easy calculation shows that

$$L_{c}(x,u^{*}) = f(x) \oplus \left[\inf_{u \in G(x)} \langle u^{*}, u \rangle, \sup_{u \in G(x)} \langle u^{*}, u \rangle\right] \{c\} \oplus \begin{cases} \mathbb{R}_{+} \cdot \{c\} & \text{if } u^{*} \in -C_{U}^{\circ} \setminus L \\ \mathbb{R}_{-} \cdot \{c\} & \text{if } u^{*} \in C_{U}^{\circ} \setminus L \\ 0 & \text{if } u^{*} \in L \\ \mathbb{R} \cdot \{c\} & \text{else}, \end{cases}$$

where the interval $[\alpha, \beta]$ for $\alpha, \beta \in \mathbb{R} \cup \{-\infty, +\infty\}$ is defined as in Section 3.5.

In the scalar theory we re-obtain the original problem from the Lagrangian by taking the supremum over all $u^* \in U^*$ (or all $u^* \in -C_U^\circ$). A corresponding result is given in the following two propositions.

Proposition 4.2.2 For all $x \in S$ and all $c \in \mathbb{R}^p$ it holds

$$\sup_{u^* \in U^*} L_c(x, u^*) = f(x)$$

Proof. From $x \in S$ we deduce that $0 \in -C_U - G(x)$. For arbitrary $c \in \mathbb{R}^p$ it holds

$$\sup_{u^* \in U^*} L_c(x, u^*) \stackrel{(1.1)}{\leq} f(x) \oplus \sup_{u^* \in U^*} \Box \Delta_c^* \left(u^* \right| - C_U - G(x) \right)$$
$$= f(x) \oplus \Delta_c^{**} \left(0 \right| - C_U - G(x) \right)$$
$$\stackrel{\text{Pr. 3.5.2 (v)}}{\leq} f(x) \oplus \Delta_c \left(0 \right| - C_U - G(x) \right) = f(x).$$

Since L(x,0) = f(x), it follows that $\sup_{u^* \in U^*} L_c(x,u^*) = f(x)$.

The preceeding proposition clarifies the relation between the problem (P) and the Lagrangian only for feasible points $x \in S$. It remains the question what happens if x is not feasible. By the analogy to scalar optimization, we expect that $\sup_{u^* \in U^*} L_c(x, u^*) = \emptyset$ in this case. This can be shown under additional assumptions to the constraints and to the choice of $c \in \mathbb{R}^p$. Note that the following proposition is not used in the proof of the duality theorem. In particular, the additional assumption $0^+G(x) \cap -C_U = \{0\}$ in the following proposition is not necessary for duality assertions.

Proposition 4.2.3 If for some $x \in \text{dom } f \cap \text{dom } G$ the condition $0^+G(x) \cap -C_U = \{0\}$ is satisfied, then for all $c \in \mathbb{R}^p \setminus -0^+f(x)$ it holds

$$\sup_{u^* \in U^*} L_c(x, u^*) = \begin{cases} f(x) & \text{if} \quad x \in S \\ \emptyset & \text{else.} \end{cases}$$
4.2. Lagrange duality

Proof. The case $x \in S$ follows from Proposition 4.2.2. Let $x \neq S$, i.e., $G(x) \cap -C_U = \emptyset$. Since $0^+G(x) \cap -C_U = \{0\}$, the separation theorem [1, Corollary 11.4.1] yields the existence of some $\bar{u}^* \in U^*$ such that

$$\alpha(x,\bar{u}^*) = \inf_{u \in G(x)} \langle \bar{u}^*, u \rangle > \sup_{u \in -C_U} \langle \bar{u}^*, u \rangle \ge 0.$$
(4.6)

Assuming that there exists some $u \in -C_U$ such that $\langle \bar{u}^*, u \rangle > 0$, we get $\sup_{u \in -C_U} \langle \bar{u}^*, u \rangle = +\infty$. This contradicts (4.6) because we supposed $G(x) \neq \emptyset$. Hence $\bar{u}^* \in -C_U^\circ$. Of course, it is true that $\sup_{u^* \in U^*} L_c(x, u^*) \geq \sup_{\lambda > 0} L_c(x, \lambda \bar{u}^*)$. Therefore, it remains to show that $\sup_{\lambda > 0} L_c(x, \lambda \bar{u}^*) = \emptyset$. Since $\bar{u}^* \in -C_U^\circ$, an easy calculation shows that $L_c(x, \lambda \bar{u}^*) \subseteq f(x) \oplus [\lambda \alpha(x, \bar{u}^*), +\infty] \{c\}$. Hence it remains to show that

$$\bigcap_{\lambda>0} \left(f(x) \oplus [\lambda \alpha(x, \bar{u}^*), +\infty] \{c\} \right) = \emptyset.$$

Assume the contrary, i.e., there exists some $y \in \mathbb{R}^p$ such that

$$\forall \lambda > 0: \ y \in f(x) \oplus [\lambda \alpha(x, \bar{u}^*), +\infty] \{c\} \, .$$

Hence, for all $n \in \mathbb{N}$ we have $y \in f(x) + [n\alpha(x, \bar{u}^*), +\infty] \{c\} + \frac{1}{n}\mathbb{B}$. Consequently,

$$\forall n \in \mathbb{N}, \ \exists y_n \in \{y\} + \frac{1}{n}\mathbb{B}, \ \exists \mu_n \ge n \cdot \alpha(x, \bar{u}^*) > 0: \quad y_n - \mu_n c \in f(x).$$

It follows

$$\frac{1}{\mu_n} (y_n - \mu_n \cdot c) = \frac{y_n}{\mu_n} - c \stackrel{n \to \infty}{\longrightarrow} -c.$$

By [1, Theorem 8.2] this implies $-c \in 0^+ f(x)$, a contradiction.

The next example shows that the assumption $0^+G(x) \cap -C_U = \{0\}$ cannot be omitted in the preceeding proposition. However, this assumption is fulfilled in many important special cases, such as for vector-valued or compact-valued functions G.

Example 4.2.4 Let f, K and c as in Proposition 4.2.3 and let $0 \in \text{dom } f$. We set $A := \left\{x \in \mathbb{R}^2_+ | x_2 \ge \frac{1}{x_1}\right\}$, $G(x) := \{x\} + A$ and $C_U := \left\{u \in \mathbb{R}^2 | u_1 \ge 0\right\}$. Then we have $C_U^\circ \setminus \{0\} = \left\{u^* \in \mathbb{R}^2 | u_1^* < 0, u_2^* = 0\right\}$. An easy computation shows

$$L_{c}(0, u^{*}) := f(0) + \begin{cases} \mathbb{R}_{+} \cdot \{c\} & \text{if} \quad u^{*} \in -C_{U}^{\circ} \setminus \{0\} \\ \mathbb{R}_{-} \cdot \{c\} & \text{if} \quad u^{*} \in C_{U}^{\circ} \setminus \{0\} \\ 0 & \text{if} \quad u^{*} = 0 \\ \mathbb{R} \cdot \{c\} & \text{else.} \end{cases}$$

It follows that $\sup_{u^* \in U^*} L_c(0, u^*) = f(0) \neq \emptyset$, but $G(0) \cap -C_U = \emptyset$, i.e., $0 \notin S$.

We next define the dual problem. The dual objective function is defined by

$$\phi_c: U^* \to \hat{\mathcal{C}}^*, \qquad \phi_c(u^*) := \inf_{x \in X} L_c(x, u^*).$$

The dual problem associated to (P) is defined by

(D_c)
$$D_c := \sup_{u^* \in U^*} \phi_c(u^*).$$

As in the scalar optimization theory, we introduce the value function by

$$v: U \to \hat{\mathcal{C}}^{\star}, \qquad v(u) := \inf \left\{ f(x) \mid x \in X : (G(x) - \{u\}) \cap -C_U \neq \emptyset \right\}$$
$$= \inf \left\{ f(x) \mid x \in X : u \in G(x) + C_U \right\}.$$

The following proposition collects some properties of the value function.

Proposition 4.2.5 The value function $v: U \to \hat{\mathcal{C}}^*$ has the following properties:

- (i) If f is convex and G is C_U -convex, then v is convex,
- (ii) v(0) = P,
- (iii) $\forall c \in \mathbb{R}^p, \ \forall u^* \in U^*: \quad v_c^*(u^*) = \Box \phi_c(-u^*),$
- (iv) $\forall c \in \mathbb{R}^p : v_c^{**}(0) = D_c.$

Proof. (i) For arbitrary $u_1, u_2 \in U$ and $\lambda \in (0, 1)$ it holds

$$\begin{aligned} \lambda v(u_{1}) \oplus (1-\lambda)v(u_{2}) \\ &= \lambda \quad \inf\{f(x_{1}) \mid x_{1} \in X : \ u_{1} \in G(x_{1}) + C_{U}\} \\ &\oplus (1-\lambda)\inf\{f(x_{2}) \mid x_{2} \in X : \ u_{2} \in G(x_{2}) + C_{U}\} \\ ^{\Pr. \pm 2.4} \inf\{\lambda f(x_{1}) \oplus (1-\lambda)f(x_{2}) \mid x_{1}, x_{2} \in X : \ u_{1} \in G(x_{1}) + C_{U}, u_{2} \in G(x_{2}) + C_{U}\} \\ ^{\text{Convexity}} &\geq \inf\{f(\lambda x_{1} + (1-\lambda)x_{2}) \mid x_{1}, x_{2} \in X : \lambda u_{1} + (1-\lambda)u_{2} \in G(\lambda x_{1} + (1-\lambda)x_{2}) + C_{U}\} \\ &= \inf\{f(x) \mid x \in X : \ \lambda u_{1} + (1-\lambda)u_{2} \in G(x) + C_{U}\} \\ &= v(\lambda u_{1} + (1-\lambda)u_{2}). \end{aligned}$$

(ii) Obvious.

(iii) Let $c \in \mathbb{R}^p$ (where $\{c\} \in \mathcal{C}^*$) and $u^* \in U^*$ be given. Then,

$$\begin{split} v_c^*(u^*) &= & \boxminus \inf_{u \in U} \left\{ \langle -u^*, u \rangle \left\{ c \right\} + v(u) \right\} \\ &= & \boxminus \inf_{u \in U} \left\{ \langle -u^*, u \rangle \left\{ c \right\} + \inf \left\{ f(x) \, \middle| \, x \in X : \, u \in G(x) + C_U \right\} \right\} \\ & \overset{\text{Pr. 1.2.4}}{=} & \boxminus \inf \left\{ \langle -u^*, u \rangle \left\{ c \right\} + f(x) \, \middle| \, x \in X, \, u \in U : \, u \in G(x) + C_U \right\} \\ &= & \boxminus \inf_{x \in X} \left\{ \inf_{u \in G(x) + C_U} \left\{ \langle -u^*, u \rangle \left\{ c \right\} + f(x) \right\} \right\} \\ & \overset{\text{Pr. 1.2.4}}{=} & \boxminus \inf_{x \in X} \left\{ f(x) \oplus \inf_{u \in G(x) + C_U} \left\{ \langle -u^*, u \rangle \left\{ c \right\} \right\} \right\} = & \boxminus \phi_c(-u^*). \end{split}$$

(iv) It holds $v_c^{**}(0) = \sup_{u^* \in U^*} \boxminus v_c^*(u^*) \stackrel{\text{(iii)}}{=} \sup_{u^* \in U^*} \phi_c^*(-u^*) = \sup_{u^* \in U^*} \phi_c^*(u^*) = D_c.$

4.3. Some special cases

Since $\hat{\mathcal{C}}^{\star}$ is order complete, we have

$$\forall c \in \mathbb{R}^p: \qquad \sup_{u^* \in U^*} \inf_{x \in X} L_c(x, u^*) \le \inf_{x \in X} \sup_{u^* \in U^*} L_c(x, u^*), \tag{4.7}$$

even if L_c is replaced by an arbitrary function from $X \times U^*$ into $\hat{\mathcal{C}}^*$. By Proposition 4.2.2 we know that $\inf_{x \in X} \sup_{u^* \in U^*} L_c(x, u^*) \leq P$. Hence, (4.7) yields weak duality between (P) and (D_c), i.e., $D_c \leq P$. Furthermore, it is easy to see that strong duality (i.e., $P = D_c$) implies that (4.7) is satisfied with equality. It follows the main result of this section, a strong duality theorem.

Theorem 4.2.6 Let f be convex and let G be C_U -convex, let

$$G(\operatorname{dom} f) \cap -\operatorname{int} C_U \neq \emptyset, \tag{4.8}$$

and, in dependence on $K := 0^+ P$, let the vector $c \in \mathbb{R}^p$ be chosen as follows:

- (i) $c \in \operatorname{ri} K \cup -\operatorname{ri} K$, if $K \subsetneq \mathbb{R}^p$ is not a linear subspace of \mathbb{R}^p or $K = \mathbb{R}^p$,
- (ii) $c \in \mathbb{R}^p \setminus K$, if $K \subsetneq \mathbb{R}^p$ is a linear subspace of \mathbb{R}^p .

Then, we have strong duality, i.e., $D_c = P$.

Proof. The proof is exactly the same as the (first) proof of Theorem 4.1.2, but using Proposition 4.2.5 instead of Proposition 4.1.1. \Box

4.3 Some special cases

In this section, we indicate some special optimization problems and calculate the corresponding dual problem. We consider the problem of minimizing a convex function $f : \mathbb{R}^n \to \hat{\mathcal{C}}^*$ with respect to a nonempty closed convex set $S \subseteq \mathbb{R}^n$. This problem can be formulated as

$$\inf_{x \in \mathbb{R}^n} \big\{ f(x) + \Delta(x \mid S) \big\},\$$

where $\Delta(\cdot | S) : \mathbb{R}^n \to \hat{\mathcal{C}}^*$ is the set-valued indicator function, defined in Section 3.5. Recall that its conjugate $\Delta_c^*(\cdot | S) : \mathbb{R}^n \to \hat{\mathcal{C}}^\diamond$ was already considered in Section 3.5 and in Section 4.2.

As a special case for the function f, let us consider $f(x) = \{M \cdot x\}$, where M is a real $p \times n$ matrix. An easy computation shows that

$$f_c^*(x^*) = \bigcup_{x \in \mathbb{R}^n} \left\{ \left(M - c \cdot (x^*)^T \right) \cdot x \right\} =: \left(M - c \cdot (x^*)^T \right) \cdot \mathbb{R}^n.$$

In the special case of n = p (recall that $\hat{\mathcal{C}}^* = \hat{\mathcal{C}}^*(\mathbb{R}^p)$) and M := I being the $n \times n$ unit matrix, Corollary 4.1.3 yields the following dual description of a nonempty closed convex set $S \subseteq \mathbb{R}^n$. In contrast to the usual dual description $S = \bigcap_{x^* \in \mathbb{R}^n} \{x \in \mathbb{R}^n | \langle x^*, x \rangle \leq \delta^*(x^*|S)\}$, we have a different parameterization in the following formula, i.e., the same x^* may generate different sets. This parameterization depends on the choice of c. If 0^+S is not a linear subspace of \mathbb{R}^n , for all $c \in \operatorname{ri}(0^+S)$ it holds

$$S = \bigcap_{x^* \in \mathbb{R}^n} \left(\left(I - c \cdot (x^*)^T \right) \cdot \mathbb{R}^n + \left[-\delta^*(x^*, -S), \delta^*(x^*, S) \right] \cdot \{c\} \right).$$
(4.9)

Moreover, if 0^+S is a linear subspace of \mathbb{R}^n , (4.9) is valid for all $c \in \mathbb{R}^n \setminus 0^+S$. Note that in (4.9) the constraint qualification (4.1) is superfluous, see Remark 4.3.1 below.

We next turn to the case of linear inequality constraints. Let A be a real $m \times n$ matrix and $b \in \mathbb{R}^m$ a given vector. We write $u \leq v$ if $v - u \in \mathbb{R}^m_+$. Consider the problem

$$\inf_{x \in S} \{Mx\}, \quad S = \{x \in \mathbb{R}^n | \ Ax \ge b\}.$$
(4.10)

In problem (P) in Section 4.1, we set $g(\cdot) = \Delta(\cdot | \bar{S})$, where $\bar{S} := \{u \in \mathbb{R}^m | u \ge b\}$. The (set-valued) support function of \bar{S} (where $\{c\} \in \mathcal{C}^{\diamond}$) can be expressed by

$$\begin{aligned} \Delta_c^*(u^*|\bar{S}) &= \bigcup_{u \ge b} \left(\langle u^*, u \rangle \{c\} \right) = \bigcup_{u \ge 0} \left(\langle u^*, u + b \rangle \{c\} \right) \\ &= \left\{ c \cdot (u^*)^T \cdot b \right\} + \bigcup_{u \ge 0} \left(\langle u^*, u \rangle \{c\} \right) = \left\{ c \cdot (u^*)^T \cdot b \right\} + \Delta_c^*(u^*|\mathbb{R}^m_+), \end{aligned}$$

where

$$\Delta_c^*(u^*|\mathbb{R}^m_+) = \begin{cases} \mathbb{R}_+ \cdot \{c\} & \text{if} \quad u^* \in \mathbb{R}^m_+ \setminus \{0\} \\ \mathbb{R}_- \cdot \{c\} & \text{if} \quad u^* \in \mathbb{R}^m_- \setminus \{0\} \\ \{0\} & \text{if} \quad u^* = 0 \\ \mathbb{R} \cdot \{c\} & \text{else.} \end{cases}$$

Note that $\Delta_c^*(u^* | \mathbb{R}^m_+) = -\Delta_c^*(-u^* | \mathbb{R}^m_+)$, hence the dual objective function $d_c : \mathbb{R}^m \to \hat{\mathcal{C}}^*$ (see Section 4.1) is given by

$$d_c(u^*) = \boxplus \left(M - c \cdot (A^T \cdot u^*)^T \right) \cdot \mathbb{R}^n \boxplus \left\{ c \cdot (u^*)^T \cdot b \right\} \boxplus \Delta_c^*(u^* | \mathbb{R}^m_+).$$

By Theorem 4.1.2 (Fenchel duality theorem) we obtain the following strong duality assertion. Since S is polyhedral, we have $\inf_{x \in S} \{Mx\} = \bigcup_{x \in S} \{Mx\} =: M \cdot S$. Let $K := 0^+ (M \cdot S)$. If there exists some $x \in \mathbb{R}^n$ such that $Ax \ge b$, then, for all $c \in \operatorname{ri} K$ if K is not a linear subspace of \mathbb{R}^p and for all $c \in \mathbb{R}^p \setminus K$ if K is a linear subspace of \mathbb{R}^p it is true that (we omit the orientation)

$$M \cdot S = \bigcap_{u^* \in \mathbb{R}^m} \left(\left(M - c \cdot (A^T \cdot u^*)^T \right) \cdot \mathbb{R}^n + \left\{ c \cdot (u^*)^T \cdot b \right\} + \Delta_c^* (u^* | \mathbb{R}^m_+) \right).$$
(4.11)

Remark 4.3.1 In Theorem 4.1.2 (Fenchel duality theorem) we suppose the constraint qualification (4.1). In the second proof of this theorem we use this condition in order to obtain the corresponding condition for the scalar problems (4.2). If all these problems are polyhedral, (4.1) can be replaced by dom $g \cap A \operatorname{dom} f \neq \emptyset$, compare e.g. [10, Corollary 5.1.9]. Hence, in (4.9) and (4.11) the constraint qualification reduces to $S \neq \emptyset$.

Chapter 5

Relationship to vector optimization

In this chapter, we discuss the relationship between the duality theory developed in the previous chapter and duality theory in vector optimization and set optimization with point relations. Duality assertions for vector optimization problems have been investigated by many authors such as Gale, Kuhn and Tucker [24], van Slyke and Wets [72], Breckner [12], Zowe [86], Brumelle [14], Gerstewitz (Tammer) [25], Jahn [39], [40], [41], Sawaragi, Nakayama and Tanino [71], Göpfert and Gerth (Tammer) [28], Göpfert and Nehse [30], Tammer [75], Bot and Wanka [11] and many others. References with respect to the extention to set–valued objective maps are already enumerated in the introduction.

Duality for vector optimization problems (and the extension to set optimization problems with point relations) have been developed in a very general setting, so the image space is often a general linear topological space. In contrast to this, up to now, our duality results have been developed for finite dimensional "image spaces" only (more precisely, the image space is based on a finite dimensional space). So our comparison is made in a finite dimensional context.

This chapter is organized as follows. We start with recalling some notions such as that of a supremal and an infimal set and we collect some related auxiliary assertion. The second section is devoted to a comparison of duality assertions of both types of set optimization problems, i.e., a comparison between point relation approach and set relation approach. In the last section, we continue the discussion about the structure of vector optimization problems, started in the introduction. We use the duality results of the previous chapter in order to give an alternative representation of a vector optimization problem.

5.1 Basic concepts

In this section, we recall some basic concepts of vector optimization and prove some auxiliary assertions that will be used in the following two sections. Throughout this section, let $C \subseteq \mathbb{R}^p$ be a closed convex pointed cone with nonempty interior. The cone C has the meaning of the

ordering cone in the following sense:

$$y_1 \le y_2 : \Leftrightarrow y_2 - y_1 \in C$$
 and $y_1 < y_2 : \Leftrightarrow y_2 - y_1 \in \operatorname{int} C$.

The duality assertions considered in Section 5.2 below are mainly based on the concepts of supremal and infimal sets, which seem to be due to Nieuwenhuis [63] and was extended by Tanino [77]. The notion of an infimal point is closely related to the concept of a *weakly minimal point* (or weakly efficient point), for instance, see [63],[40],[77],[57]. Given a set $A \subseteq \mathbb{R}^p$, a point $y \in \mathbb{R}^p$ is said to be a *weakly minimal point* of A if

$$y \in A$$
 and $(\{y\} - \operatorname{int} C) \cap A = \emptyset$.

The set of all weakly minimal points of A is denoted by $\operatorname{wMin}[A, C]$ or simply by wMin A. A point $y \in \mathbb{R}^p$ is said to be an *infimal point* of A if

$$y \notin A + \operatorname{int} C$$
 and $\{y\} + \operatorname{int} C \subseteq A + \operatorname{int} C$.

The set of all infimal points of A is denoted by Inf[A, C] or by Inf A. Likewise, by replacing C by -C, we define *weakly maximal points* and *supremal points* of A as well as the sets wMax[A, C] and Sup[A, C].

In contrast to [77], we avoid the extension of the space \mathbb{R}^p by two imaginary elements $+\infty$ and $-\infty$, because this makes the proofs a little bit easier. As a consequence, it may happen that the infimal set is empty. In particular, we have $\operatorname{Inf}[\emptyset, C] = \emptyset$ (instead of $\operatorname{Inf}[\emptyset, C] = +\infty$) and $\operatorname{Inf}[A, C] = \emptyset$ (instead of $\operatorname{Inf}[A, C] = -\infty$) if $A + C = \mathbb{R}^p$. Otherwise, the infimal set is nonempty, see [63, Theorem I-18].

We next summarize some properties of supremal and infimal sets.

Proposition 5.1.1 ([63],[77]) Let $A \subseteq \mathbb{R}^p$. Then it holds

- (i) $\operatorname{Inf}[A, C] = \operatorname{Inf}[\operatorname{cl} A, C],$
- (ii) $\operatorname{Inf} A = \operatorname{wMin} \operatorname{cl} (A + \operatorname{int} C),$
- (iii) If $A + C \neq Y$, then $A + \operatorname{int} C = \operatorname{Inf} A + \operatorname{int} C$.

Proof. (i) See [63, Theorem I-15] or apply Proposition A.8.
(ii) See [77, Proposition 4.3] or apply Proposition A.8.
(iii) See [77, Proposition 4.4].

Similar to the relation \preccurlyeq_K , discussed in the introduction, we define a relation in $\mathcal{P}(\mathbb{R}^p)$ by

$$A \preccurlyeq_C B \qquad :\Leftrightarrow \qquad B \subseteq \operatorname{cl}(A+C).$$

Of course, this relation is reflexive and transitive, but not antisymmetric. In the following two corollaries we present some simple properties concerning infimal sets.

Corollary 5.1.2 Let $A \subseteq B \subseteq \mathbb{R}^p$ such that $B + C \neq \mathbb{R}^p$. Then, $\inf B \preccurlyeq_C \inf A$.

Proof. Let $y \in \text{Inf } A$. By Proposition 5.1.1 (iii) we conclude that $\{y\} + \text{int } C \subseteq A + \text{int } C \subseteq B + \text{int } C = \text{Inf } B + \text{int } C$. Letting $c \in \text{int } C$, for all $\lambda > 0$ we have $y + \lambda c \in \text{Inf } B + \text{int } C$, hence $y \in \text{cl}(\text{Inf } B + \text{int } C) \subseteq \text{cl}(\text{Inf } B + C)$.

Corollary 5.1.3 Let $A, B \subseteq \mathbb{R}^p$. Then,

$$(\operatorname{Inf} B \preccurlyeq_C \operatorname{Inf} A, \operatorname{Inf} A \preccurlyeq_C \operatorname{Inf} B) \Rightarrow \operatorname{Inf} A = \operatorname{Inf} B.$$

Proof. We can assume that $\inf A$ and $\inf B$ are nonempty, hence $A + C \neq Y$ and $B + C \neq Y$. Using Proposition A.8, we obtain $\inf A + \operatorname{int} C = \operatorname{Inf} B + \operatorname{int} C$. Proposition 5.1.1 (iii) yields $A + \operatorname{int} C = B + \operatorname{int} C$. Hence, the result follows from the definition of the infimal set. \Box

We close this section with two results on the existence of weakly minimal elements of closed convex subsets of \mathbb{R}^p .

Proposition 5.1.4 Let $A \subseteq \mathbb{R}^p$ be a nonempty closed convex set with $\mathbb{R}^p \neq 0^+ A \supseteq C$. Then, $\forall c \in \text{int } C : \qquad A = \text{wMin}[A, C] + \mathbb{R}_+ \cdot \{c\}.$

Proof. " \supseteq ". wMin[A, C] + $\mathbb{R}_+ \cdot \{c\} \subseteq A + C \subseteq A + 0^+ A = A$.

" \subseteq ". For given $y \in A$, define $\alpha := \inf \{ \gamma \in \mathbb{R} | y + \gamma c \in A \}$. If $\alpha = -\infty$, we conclude that $-c \in 0^+ A$. Hence, $0 = c - c \in \operatorname{int} C + 0^+ A \subseteq \operatorname{int} 0^+ A$. This contradicts $0^+ A \neq \mathbb{R}^p$. Hence we have $-\infty < \alpha \leq 0$. The closedness of A implies that $y' := y + \alpha c \in A$. We next show that y' even belongs to wMin[A, C]. Assuming the contrary yields some $y'' \in (\{y'\} - \operatorname{int} C) \cap A$. We have y'' = y' - c' for some $c' \in \operatorname{int} C$. Furthermore, there exists some $\delta > 0$ such that $c'' := c' - \delta c \in \operatorname{int} C$. Hence we obtain $y''' := y'' + c'' \in A + C \subseteq A + 0^+ A = A$. On the other hand we have $y''' = y' - \delta c = y + (\alpha - \delta)c \in A$ where $(\alpha - \delta) < \alpha$. This contradicts the minimality of α . Consequently, we obtain $y' \in \operatorname{wMin}[A, C]$ and hence $y \in \operatorname{wMin}[A, C] + \mathbb{R}_+ \cdot \{c\}$.

Proposition 5.1.5 Let $A \subseteq \mathbb{R}^p$ be a nonempty closed convex set with $0^+A \supseteq C$. Then, for all $y \in \mathbb{R}^p \setminus A$ and all $c \in \operatorname{int} C$ there exists some $\alpha > 0$ such that $y + \alpha c \in \operatorname{wMin}[A, C]$.

Proof. Let $\bar{y} \in \mathbb{R}^p \setminus A$ and $\bar{c} \in \text{int } C$ be given. Define the set $B := \{\bar{y}\} + \mathbb{R}_+ \{\bar{c}\}$. Assume that $A \cap B = \emptyset$. Then, the separation theorem [1, Theorem 11.3] implies the existence of some $y^* \in \mathbb{R}^p \setminus \{0\}$ such that

$$-\infty < \sup_{a \in A} \langle y^*, a \rangle \le \inf_{b \in B} \langle y^*, b \rangle < \infty.$$
(5.1)

By Lemma 1.3.1 (i) we know that $y^* \in (0^+A)^\circ$. Since $0^+A \supseteq C$, we obtain $(0^+A)^\circ \subseteq C^\circ$ and hence $y^* \in C^\circ \setminus \{0\}$. Proposition 2.3.2 yields that $\langle y^*, \bar{c} \rangle < 0$. It follows that $\inf_{b \in B} \langle y^*, b \rangle = -\infty$, which contradicts (5.1). Hence, we have $A \cap B \neq \emptyset$. Consequently, there exists some $\bar{\alpha} > 0$ such that $\bar{y} + \bar{\alpha}\bar{c} \in A$. From Proposition 5.1.4 we deduce that there exists $\hat{y} \in \text{wMin}[A, C]$ and $\hat{\alpha} \ge 0$ such that $\bar{y} + \bar{\alpha}\bar{c} = \hat{y} + \hat{\alpha}\bar{c}$. It remains to show that $\alpha := \bar{\alpha} - \hat{\alpha} > 0$. Assuming the contrary, we obtain $\bar{y} \in \text{wMin}[A, C] + C \subseteq A + 0^+A = A$, a contradiction. \Box

5.2 Point relation vs. set relation approach

The aim of this section is to show that a strong duality assertion with respect to set relation approach follows from a strong duality assertion with respect to point relation approach and vice versa. This is expressed by the "equivalence theorem", which states that, without the typical constraint qualification, both types of strong duality assertions are equivalent. We begin with the definition of the dual pair of set optimization problems with point relations. Of course, as a special case we can consider vector optimization problems instead.

For given set-valued maps $F : \mathbb{R}^n \Rightarrow \mathbb{R}^p$, $G : \mathbb{R}^n \Rightarrow \mathbb{R}^m$, a closed convex pointed cone $C \subseteq \mathbb{R}^p$ with nonempty interior and a nonempty closed convex cone $C_U \subseteq \mathbb{R}^m$, we define a primal set optimization problem by (the superscript "PR" stands for "point relation")

$$(\mathbf{P}^{\mathrm{PR}}) \qquad \bar{P} := \mathrm{Inf}\left[\bigcup_{x \in S} F(x), C\right], \qquad S := \left\{x \in \mathbb{R}^n \mid G(x) \cap -C_U \neq \emptyset\right\}.$$

For given $c \in \mathbb{R}^p$, consider the Lagrangian $\overline{L}_c : \mathbb{R}^n \times \mathbb{R}^m \implies \mathbb{R}^p$, defined similar to the Lagrangian $L_c : \mathbb{R}^n \times \mathbb{R}^m \to \hat{\mathcal{C}}^*$ in Section 4.2, by

$$\bar{L}_c(x, u^*) := F(x) + C + \bigcup_{u \in G(x) + C_U} \left\{ \langle u^*, u \rangle \left\{ c \right\} \right\}.$$

The dual objective function $\Phi_c : \mathbb{R}^m \rightrightarrows \mathbb{R}^p$ is defined by

$$\Phi_c(u^*) := \inf \bigcup_{x \in \mathbb{R}^n} \bar{L}_c(x, u^*),$$

and the dual problem associated to (P^{PR}) is as follows:

$$(\mathbf{D}_{c}^{\mathrm{PR}}) \qquad \qquad \bar{D}_{c} := \mathrm{Sup} \bigcup_{u^{*} \in \mathbb{R}^{m}} \Phi_{c}(u^{*}).$$

Let us first prove some auxiliary assertions.

Proposition 5.2.1 Let X be an arbitrary set and $H : X \rightrightarrows \mathbb{R}^p$ be a set-valued map. Then it holds

(i)
$$\left(\bigcup_{x\in X} \operatorname{wMin} H(x) - \operatorname{int} C\right) \cap \left(\bigcap_{x\in X} H(x)\right) = \emptyset,$$

If H has nonempty closed convex values and $0^+H(x) \supseteq C$ for all $x \in X$, then it holds

(ii)
$$\left(\bigcup_{x\in X} \operatorname{wMin} H(x) - \operatorname{int} C\right) \cup \left(\bigcap_{x\in X} H(x)\right) = \mathbb{R}^p,$$

(iii) $\sup \bigcup_{x \in X} \inf H(x) = \operatorname{wMin} \bigcap_{x \in X} H(x).$

5.2. Point relation vs. set relation approach

Proof. (i) Assume there exists some $y \in (\bigcup_{x \in X} \operatorname{wMin} H(x) - \operatorname{int} C) \cap (\bigcap_{x \in X} H(x))$. Hence there is some $\bar{x} \in X$ such that $y \in (\operatorname{wMin} H(\bar{x}) - \operatorname{int} C) \cap H(\bar{x})$. Consequently, there exists some $\bar{y} \in \operatorname{wMin} H(\bar{x})$ such that $y \in \{\bar{y}\} - \operatorname{int} C$. Since $\bar{y} \in \operatorname{wMin} H(\bar{x})$ we have $(\{\bar{y}\} - \operatorname{int} C) \cap H(\bar{x}) = \emptyset$ and hence $y \notin H(\bar{x})$, a contradiction.

(ii) Let $y \in \mathbb{R}^p \setminus \bigcap_{x \in X} H(x)$. Then there exists some $\bar{x} \in X$ such that $y \notin H(\bar{x})$. Let $c \in \operatorname{int} C$ be given. By Proposition 5.1.5 there exists $\alpha > 0$ such that $y + \alpha c \in \operatorname{wMin}[H(\bar{x}), C]$. Hence, $y \in \bigcup_{x \in X} \operatorname{wMin} H(x) - \operatorname{int} C.$

(iii) By Proposition 5.1.1 (ii) we have $\inf H(x) = \operatorname{wMin} \operatorname{cl} (H(x) + \operatorname{int} C)$. By virtue of Proposition A.3 (iii) and A.10 and since $C \subseteq 0^+ H(x)$ it follows $\inf H(x) = \operatorname{wMin} H(x)$. Let $y \in \operatorname{Sup} \bigcup_{x \in X} \operatorname{Inf} H(x) = \operatorname{Sup} \bigcup_{x \in X} \operatorname{wMin} H(x)$. By the definition of the supremal set this is equivalent to

$$y \notin \bigcup_{x \in X} \operatorname{wMin} H(x) - \operatorname{int} C$$
, $\{y\} - \operatorname{int} C \subseteq \bigcup_{x \in X} \operatorname{wMin} H(x) - \operatorname{int} C$

By (i) and (ii), this is equivalent to

$$y \in \bigcap_{x \in X} H(x) , \qquad (\{y\} - \operatorname{int} C) \cap \left(\bigcap_{x \in X} H(x)\right) = \emptyset,$$

i.e., $y \in \operatorname{wMin} \bigcap_{x \in X} H(x).$

We next investigate convex problems. We suppose that the objective map F is C-convex (see Definition 4.2.1) and, as in Section 4.2, the map G is supposed to be C_U -convex. Of course, F is C-convex if and only if $f(\cdot) := \operatorname{cl}(F(\cdot) + C)$ is convex function $f: \mathbb{R}^n \to \hat{\mathcal{C}}^*$ (see Corollary 3.4.7). Furthermore, if $G : \mathbb{R}^n \rightrightarrows \mathbb{R}^m$ is C_U -convex, we deduce that for all $u^* \in \mathbb{R}^m$ the map

$$g_{u^*}(\,\cdot\,) := \operatorname{cl} \bigcup_{u \in G(\,\cdot\,) + C_U} (\langle u^*, u \rangle \, \{c\})$$

can be interpreted as a convex function $g_{u^*}: \mathbb{R}^n \to \hat{\mathcal{C}}^*$. Consequently, we have the following relationship between the objective function Φ_c of problem (D_c^{PR}) and the objective function ϕ_c of problem (D_c), defined in Section 4.2.

Proposition 5.2.2 If F is C-convex and G is C_U -convex, we have

$$\forall u^* \in \mathbb{R}^m : \qquad \Phi_c(u^*) = \inf \phi_c(u^*).$$

Proof. For arbitrary $u^* \in \mathbb{R}^m$ it holds

$$\Phi_{c}(u^{*}) = \operatorname{Inf} \bigcup_{x \in \mathbb{R}^{n}} \overline{L}(x, u^{*}) \stackrel{\operatorname{Pr. 5.1.1}(i)}{=} \operatorname{Inf} \operatorname{cl} \bigcup_{x \in \mathbb{R}^{n}} \overline{L}(x, u^{*})$$

$$\overset{\operatorname{Pr. A.2 (iii), A.4 (vii)}}{=} \operatorname{Inf} \operatorname{cl} \bigcup_{x \in \mathbb{R}^{n}} \left(f(x) \oplus g_{u^{*}}(x) \right)$$

$$\overset{f, g_{u^{*}} \text{ convex}}{=} \operatorname{Inf} \inf_{x \in \mathbb{R}^{n}} L(x, u^{*}) = \operatorname{Inf} \phi_{c}(u^{*}).$$

We now compare strong duality between the problems (P^{PR}) and (D_c^{PR}) with strong duality between the problems (P) and (D_c) , defined in Section 4.2.

Theorem 5.2.3 (Equivalence theorem) Let F be C-convex and let G be C_U -convex. If $S \neq \emptyset$, then for all $c \in \mathbb{R}^p$ the following strong duality assertions are equivalent:

- (i) Inf $\bigcup_{x \in S} F(x) = \sup \bigcup_{u^* \in \mathbb{R}^m} \Phi_c(u^*),$
- (ii) $\inf_{x \in S} f(x) = \sup_{u^* \in \mathbb{R}^m} \phi_c(u^*).$

Proof. The *C*-convexity of *F* and the C_U -convexity of *G* imply that the set $\bigcup_{x \in S} f(x)$ is convex, hence

$$\operatorname{Inf} \bigcup_{x \in S} F(x) \xrightarrow{\operatorname{Pr. 5.1.1 (ii)}} \operatorname{wMin cl} \left(\bigcup_{x \in S} F(x) + \operatorname{int} C \right)$$
$$\xrightarrow{\operatorname{Pr. A.1 (iii), A.2 (iii), A.4 (vii)}} \operatorname{wMin cl} \bigcup_{x \in S} f(x) = \operatorname{wMin \inf_{x \in S} f(x)}$$

Moreover, taking into account that $0^+\phi_c(u^*) \supseteq C$ for all $u^* \in \mathbb{R}^m$, we have

$$\sup \bigcup_{u^* \in \mathbb{R}^m} \Phi_c(u^*) \stackrel{\text{Pr. 5.2.2}}{=} \sup \bigcup_{u^* \in \mathbb{R}^m} \inf \phi_c(u^*) \stackrel{\text{Pr. 5.2.1 (iii)}}{=} \text{wMin} \sup_{u^* \in \mathbb{R}^m} \phi_c(u^*).$$

We directly conclude that (ii) implies (i). In order to show the opposite inclusion, let (i) be satisfied. Note that $S \neq \emptyset$ implies that $0^+ \inf_{x \in S} f(x) \supseteq C$ and by the weak duality inequality it follows that $0^+ \sup_{u^* \in \mathbb{R}^m} \phi_c(u^*) \supseteq C$. If $\inf_{x \in S} f(x) \neq \mathbb{R}^p$, Proposition 5.1.4 yields $(c \in \operatorname{int} C)$

$$\inf_{x \in S} f(x) = \operatorname{wMin} \inf_{x \in S} f(x) + \mathbb{R}^+ \cdot \{c\} = \operatorname{wMin} \sup_{u^* \in \mathbb{R}^m} \phi_c(u^*) + \mathbb{R}^+ \cdot \{c\} = \sup_{u^* \in \mathbb{R}^m} \phi_c(u^*),$$

otherwise (ii) follows from the weak duality inequality of Section 4.2.

Finally, we prove a duality theorem for the problems (P^{PR}) and (D_c^{PR}) by applying the duality theorem of Section 4.2, which is based on set relation approach.

Theorem 5.2.4 (Duality Theorem) For the problems (P^{PR}) and (D^{PR}_c), for all $c \in \mathbb{R}^p$ it holds weak duality, i.e.,

$$(\bar{D}_c - \operatorname{int} C) \cap \bar{P} = \emptyset.$$

If, additionally, F is C-convex and G is C_U -convex and the constraint qualification

$$G(\operatorname{dom} F) \cap -\operatorname{int} C_U \neq \emptyset \tag{5.2}$$

is satisfied, for all $c \in \text{int } C$, we have strong duality between $(\mathbf{P}^{\mathrm{PR}})$ and $(\mathbf{D}_{c}^{\mathrm{PR}})$, i.e., $\bar{P} = \bar{D}_{c}$.

5.3. On the structure of vector optimization problems

Proof. Weak duality. It holds

$$\begin{split} \bar{D}_{c} - \operatorname{int} C &= \operatorname{Sup} \bigcup_{u^{*} \in \mathbb{R}^{m}} \Phi_{c}(u^{*}) - \operatorname{int} C &\subseteq \bigcup_{u^{*} \in \mathbb{R}^{m}} \Phi_{c}(u^{*}) - \operatorname{int} C \\ &= \bigcup_{u^{*} \in \mathbb{R}^{m}} \operatorname{Inf} \bigcup_{x \in \mathbb{R}^{n}} \bar{L}_{c}(x, u^{*}) - \operatorname{int} C \\ \\ \overset{\operatorname{Pr. 5.1.1 (ii)}}{=} & \bigcup_{u^{*} \in \mathbb{R}^{m}} \operatorname{wMin cl} \left(\bigcup_{x \in \mathbb{R}^{n}} \bar{L}_{c}(x, u^{*}) + \operatorname{int} C \right) - \operatorname{int} C \\ \\ \overset{\operatorname{Pr. A.1 (iii), A.2 (iii), A.4 (vii)}}{=} & \bigcup_{u^{*} \in \mathbb{R}^{m}} \operatorname{wMin cl} \bigcup_{x \in \mathbb{R}^{n}} \bar{L}_{c}(x, u^{*}) - \operatorname{int} C \\ \end{array} \end{split}$$

On the other hand, we have

$$\bar{P} = \operatorname{Inf} \bigcup_{x \in S} F(x) \stackrel{\text{Pr. 5.1.1 (ii)}}{=} \operatorname{wMin} \operatorname{cl} \left(\bigcup_{x \in S} F(x) + \operatorname{int} C \right) \subseteq \operatorname{cl} \bigcup_{x \in S} \left(F(x) + C \right).$$

For $x \in S$ and arbitrary $u^* \in \mathbb{R}^m$, it is easy to check that $F(x) + C \subseteq \overline{L}(x, u^*)$ and hence $F(x) + C \subseteq \bigcap_{u^* \in \mathbb{R}^m} \overline{L}(x, u^*)$. Using a (very easy to prove) weak duality assertion analogous to that in Section 4.2, but for $\hat{\mathcal{F}}$ -valued functions, we obtain

$$\bar{P} \subseteq \operatorname{cl} \bigcup_{x \in S} \bigcap_{u^* \in \mathbb{R}^m} \bar{L}_c(x, u^*) \subseteq \operatorname{cl} \bigcup_{x \in \mathbb{R}^n} \bigcap_{u^* \in \mathbb{R}^m} \bar{L}_c(x, u^*) \subseteq \bigcap_{u^* \in \mathbb{R}^m} \operatorname{cl} \bigcup_{x \in \mathbb{R}^n} \bar{L}_c(x, u^*) =: A_2$$

Proposition 5.2.1 (i) yields that A_1 and A_2 are disjoint. Hence $(\overline{D}_c - \operatorname{int} C) \cap \overline{P} = \emptyset$.

Strong duality. It is easy to verify that the assumptions of Theorem 4.2.6 are satisfied. For instance, we have $K := 0^+ \inf_{x \in S} f(x) \supseteq C$. Since $c \in \operatorname{int} C$ we deduce that either $K = \mathbb{R}^p$ or $K \subsetneq \mathbb{R}^p$ is not a linear subspace of \mathbb{R}^p and $c \in \operatorname{ri} K$. Theorem 4.2.6 yields strong duality between (P) and (D_c). Hence the result follows from Theorem 5.2.3.

5.3 On the structure of vector optimization problems

In this section, we discuss the structure of a vector optimization problems and set optimization problems with point relations. We already started this discussion in the introduction.

Let $C \subseteq \mathbb{R}^p$ be a closed convex pointed cone and $A \subseteq \mathbb{R}^p$. Recall that an element $y \in A$ is said to be *minimal* (or *efficient*) if $y \in A$ and $(\{y\} - C \setminus \{0\}) \cap A = \emptyset$. The set of all minimal elements of A is denoted by Min[A, C].

Let $F : \mathbb{R}^n \Rightarrow \mathbb{R}^p$ set-valued objective map and let $S \subseteq \mathbb{R}^n$ a set which describes the constraints. Then we are interested to know the set

$$\operatorname{Min}\left[\bigcup_{x\in S}F(x),C\right],\tag{5.3}$$

We observe that problem (5.3) consists of the following two components:

- (i) $A := \bigcup_{x \in S} F(x)$ (an optimization problem based on set inclusion),
- (ii) Min[A, C] (the determination of the set of minimal points).

Note that the union in (i) can be interpreted as the infimum in the ordered conlinear space $(\hat{\mathcal{P}}(\mathbb{R}^p), \supseteq)$ of all subsets of \mathbb{R}^p .

We next investigate convex problems. Recall that a set-valued map $F : \mathbb{R}^n \rightrightarrows \mathbb{R}^p$ is said to be *C*-closed (e.g. [57]) if gr ($F(\cdot) + C$) is closed. It is easy to verify that F is *C*-convex (see Definition 4.2.1) and *C*-closed if and only if $f(\cdot) := \operatorname{cl}(F(\cdot) + C)$ can be interpreted as a lower semi-continuous convex function $f : \mathbb{R}^n \to \hat{C}^*$ (see Corollary 3.4.7). Moreover, the following assertion holds true.

Proposition 5.3.1 Let F be C-convex and C-closed and let $S \subseteq \mathbb{R}^n$ be convex and compact. Then,

$$\operatorname{Min}\left[\bigcup_{x\in S}F(x),C\right] = \operatorname{Min}\left[\inf_{x\in S}f(x),C\right].$$

Proof. We show that the set $P := \bigcup_{x \in S} F(x) + C$ is convex and closed. Indeed, let $y_i \in P$, (i = 1, 2). It follows $y_i \in F(x_i) + C$ for some $x_i \in S$, (i = 1, 2). Hence $\lambda y_1 + (1 - \lambda)y_2 \in \lambda(F(x_1) + C) + (1 - \lambda)(F(x_2) + C) \subseteq F(\lambda x_1 + (1 - \lambda)x_2) + C \subseteq P$. To show the closedness, let $\{y_n\}_{n \in \mathbb{N}} \subseteq P$ with $y_n \to \overline{y}$ be given. Then there is a sequence $\{x_n\}_{n \in \mathbb{N}} \subseteq S$ such that $(x_n, y_n) \in \operatorname{gr} (F(\cdot) + C)$. Since S is compact we can choose a convergent subsequence $x_{n_k} \to \overline{x} \in S$. Since $\operatorname{gr} (F(\cdot) + C)$ is closed, we deduce that $(\overline{x}, \overline{y}) \in \operatorname{gr} (F(\cdot) + C)$, i.e., $\overline{y} \in F(\overline{x}) + C \subseteq P$. Hence we have $P = \operatorname{cl} \operatorname{conv} \bigcup_{x \in S} (F(x) + C) = \operatorname{cl} \operatorname{conv} \bigcup_{x \in S} \operatorname{cl} (F(x) + C) = \operatorname{inf}_{x \in S} f(x)$. As an easy consequence of the definition of minimal elements we always have $\operatorname{Min}[A, C] = \operatorname{Min}[A + C, C]$ (for this, C has to be pointed). It follows that $\operatorname{Min}[P, C] = \operatorname{Min}[\bigcup_{x \in S} F(x) + C, C] = \operatorname{Min}[\bigcup_{x \in S} F(x), C]$.

As in Section 4.2, let $G : \mathbb{R}^n \Rightarrow \mathbb{R}^m$ be a set-valued map, $C_U \subseteq \mathbb{R}^m$ a nonempty closed convex cone and $S := \{x \in \mathbb{R}^n | G(x) \cap -C_U \neq \emptyset\}$. Using the strong duality assertion of Theorem 4.2.6 we obtain an equivalent characterization of the problem (5.3). For simplicity we suppose that C has a nonempty interior.

Corollary 5.3.2 Let F be C-convex and C-closed, let G be C_U -convex, S compact, $c \in \text{int } C$ and $G(\text{dom } F) \cap -\text{int } C_U \neq \emptyset$. Then,

$$\operatorname{Min}\left[\bigcup_{x\in S} F(x), C\right] = \operatorname{Min}\left[\bigcap_{u^*\in\mathbb{R}^m} \phi_c(u^*), C\right].$$

where $\phi_c : \mathbb{R}^m \to \hat{\mathcal{C}}^*$ is the dual objective function of problem (P) for $f(\cdot) := \operatorname{cl}(F(\cdot) + C)$, defined in Section 4.2.

Proof. By Proposition 5.3.1 we know that $\operatorname{Min}\left[\bigcup_{x\in S} F(x), C\right] = \operatorname{Min}\left[\inf_{x\in S} f(x), C\right]$, where the convexity of S is an easy consequence of the C_U -convexity of G. In order to get

 $\inf_{x \in S} f(x) = \sup_{u^* \in \mathbb{R}^m} \phi_c(u^*)$ we apply Theorem 4.2.6. The assumptions of Theorem 4.2.6 are easy to verify, for instance, we have $K := 0^+ \inf_{x \in S} f(x) \supseteq C$. Since $c \in \operatorname{int} C$ we deduce that either $K = \mathbb{R}^p$ or $K \subsetneq \mathbb{R}^p$ is not a linear subspace of \mathbb{R}^p and $c \in \operatorname{ri} K$. \Box

We close this work with an illustration how duality of optimization problems based on set inclusion can be used to sandwich the infimal set of a vector optimization problem. In scalar optimization we can use approximate primal and dual problems in order to get upper and lower bounds for the infimum of the original problem. For instance, we can take the infimum (supremum) over finitely many primal (dual) feasible points in order to obtain such bounds. Let us proceed analogously. Instead of the problems (P) and (D_c), defined in Section 4.2, we consider the following approximate problems

$$\widetilde{P} := \inf_{x \in \widetilde{S}} f(x), \quad \widetilde{S} \subseteq S; \qquad \qquad \widetilde{D}_c := \sup_{u^* \in \widetilde{S}_D} \phi_c(u^*), \quad \widetilde{S}_D \subseteq \mathbb{R}^m$$

By the weak duality we always have $\widetilde{D}_c \leq D_c \leq P \leq \widetilde{P}$. If $\widetilde{D}_c + C \neq \mathbb{R}^p$, Corollary 5.1.2 yields that

$$\operatorname{Inf} D_c \preccurlyeq_C \operatorname{Inf} D_c \preccurlyeq_C \operatorname{Inf} P \preccurlyeq_C \operatorname{Inf} P.$$

Of course, if F is C-convex we have

$$\operatorname{Inf} P = \operatorname{Inf} \inf_{x \in S} f(x) = \operatorname{Inf} \left(\bigcup_{x \in S} F(x) + C \right) = \operatorname{Inf} \bigcup_{x \in S} F(x) =: \operatorname{Inf} F(S).$$

Hence the set $\inf F(S)$ is sandwiched in the following sense:

$$\operatorname{Inf} D_c \preccurlyeq_C \operatorname{Inf} F(S) \preccurlyeq_C \operatorname{Inf} P.$$



By Corollary 5.1.3, strong duality between (P) and (D_c) implies that

$$\operatorname{Inf} D_c = \operatorname{Inf} F(S) = \operatorname{Inf} P.$$

Chapter 5. Relationship to vector optimization

Appendix

A Some calculus rules of sets

For the convenience of the reader we collect some simple calculus rules for sets. Of course, these rules are involved in many books (but unfortunately not all in one). Therefore, we present the most assertions with proof. Furthermore, in case of inclusions, we show that the opposite inclusion is not true, in general.

Proposition A.1 Let X be a linear space, let $A_i, A, B \subseteq X$ and let I be an arbitrary index set. Then,

- (i) conv $\bigcap_{i \in I} A_i \subseteq \bigcap_{i \in I} \operatorname{conv} A_i$,
- (ii) conv $\bigcup_{i \in I} A_i \supseteq \bigcup_{i \in I} \operatorname{conv} A_i$,

(iii)
$$\bigcup_{i \in I} (A_i + B) = \left(\bigcup_{i \in I} A_i\right) + B,$$

(iv)
$$\bigcap_{i \in I} (A_i + B) \supseteq \left(\bigcap_{i \in I} A_i\right) + B$$
,

(v)
$$\operatorname{conv}(A+B) = \operatorname{conv} A + \operatorname{conv} B.$$

Proof. The proof of (i) - (iv) is immediate, (v) can be found in [60].

In general, (i), (ii) and (iv) does not hold with equality as the following examples show: (i) $A_1 = \{0, 2\}, A_2 = \{1\}.$ (ii) $A_1 = \{0, 1\}, A_2 = \{2, 3\}.$ (iv) $A_1 = \{0\}, A_2 = \{1\}, B = [-1, 1].$

Proposition A.2 Let X be a topological space, let $A_i, A, B \subseteq X$ and let I be an arbitrary index set. Then,

- (i) cl $\bigcap_{i \in I} A_i \subseteq \bigcap_{i \in I} cl A_i$,
- (ii) cl $\bigcup_{i \in I} A_i \supseteq \bigcup_{i \in I} cl A_i$,

- (iii) cl $\bigcup_{i \in I} A_i = cl \bigcup_{i \in I} cl A_i$,
- (iv) $\operatorname{cl}(A \cup B) = \operatorname{cl} A \cup \operatorname{cl} B$.

Proof. (i) cl $\bigcap_{i \in I} A_i \subseteq$ cl $\bigcap_{i \in I}$ cl A_i . Since $\bigcap_{i \in I}$ cl A_i is closed, cl $\bigcap_{i \in I}$ cl $A_i = \bigcap_{i \in I}$ cl A_i . (ii) For all $i \in I$ we have cl $\bigcup_{i \in I} A_i \supseteq$ cl A_i . Hence cl $\bigcup_{i \in I} A_i \supseteq \bigcup_{i \in I}$ cl A_i .

- (iii) From (ii).
- (iv) Since $\operatorname{cl} A \cup \operatorname{cl} B$ is closed, this follows from (iii).

To see that, in general, (i) and (ii) does not hold with equality consider the following examples: (i) $A_1 = (0, 1), A_2 = (1, 2).$ (ii) $I = \mathbb{N}, A_i = \{1/i\}.$

Proposition A.3 Let X be a linear topological space and let $A \subseteq X$ be convex. Then,

- (i) $\operatorname{cl} A$ is convex,
- (ii) int A is convex,
- (iii) int $A \neq \emptyset$ implies that $\operatorname{cl} A = \operatorname{cl}(\operatorname{int} A)$ and $\operatorname{int}(\operatorname{cl} A) = \operatorname{int} A$.

Proof. For instance, see [46].

Assertion (iii) of the previous proposition, in general, does not hold for nonconvex sets. For instance, consider $A = [0, 1] \cup \{2\}$ and $A = \mathbb{R} \setminus \{0\} \subseteq \mathbb{R}$, respectively.

Proposition A.4 Let X be a linear topological space, let $A_i, A, B \subseteq X$ and let I be an arbitrary index set. Then,

- (i) $\operatorname{cl}\operatorname{conv} A \supseteq \operatorname{conv} \operatorname{cl} A$,
- (ii) $\operatorname{cl}\operatorname{conv} A = \operatorname{cl}\operatorname{conv}\operatorname{cl} A$,
- (iii) $\operatorname{cl}\operatorname{conv}\bigcap_{i\in I}A_i\subseteq\bigcap_{i\in I}\operatorname{cl}\operatorname{conv}A_i$,
- (iv) $\operatorname{cl}\operatorname{conv}\bigcup_{i\in I}A_i\supseteq\bigcup_{i\in I}\operatorname{cl}\operatorname{conv}A_i$,
- (v) cl conv $\bigcup_{i \in I} A_i = cl conv \bigcup_{i \in I} cl conv A_i$,
- (vi) $\operatorname{cl}(A+B) \supseteq \operatorname{cl} A + \operatorname{cl} B$,
- (vii) $\operatorname{cl}(A+B) = \operatorname{cl}(\operatorname{cl} A + \operatorname{cl} B),$
- (viii) $\operatorname{cl}\operatorname{conv}(A+B) \supseteq \operatorname{cl}\operatorname{conv} A + \operatorname{cl}\operatorname{conv} B$,
- (ix) $\operatorname{cl}\operatorname{conv}(A+B) = \operatorname{cl}(\operatorname{cl}\operatorname{conv} A + \operatorname{cl}\operatorname{conv} B).$

A. Some calculus rules of sets

Proof. (i) $\operatorname{cl} \operatorname{conv} A \supseteq \operatorname{cl} A$. Since $\operatorname{cl} \operatorname{conv} A$ is convex , we have $\operatorname{cl} \operatorname{conv} A \supseteq \operatorname{conv} \operatorname{cl} A$. (ii) Follows from (i).

(iii) Since $\bigcap_{i \in I} A_i \subseteq \bigcap_{i \in I} \operatorname{cl} \operatorname{conv} A_i$ and $\bigcap_{i \in I} \operatorname{cl} \operatorname{conv} A_i$ is closed and convex.

(iv) For all $i \in I$ it holds $\operatorname{cl} \operatorname{conv} \bigcup_{i \in I} A_i \supseteq \operatorname{cl} \operatorname{conv} A_i$. Hence (iv) is true.

(v) From (iv).

(vi) Let $x \in \operatorname{cl} A + \operatorname{cl} B$, i.e., $x = \overline{a} + \overline{b}$ with $\overline{a} \in \operatorname{cl} A$ and $\overline{b} \in \operatorname{cl} B$. Let \mathcal{B} be a neighborhood base of the origin of X, formed by balanced absorbing sets (for instance, see [31]). Then, for $A \subseteq X$ it holds $\operatorname{cl} A = \bigcap \{A + V | V \in \mathcal{B}\}$. Let $V \in \mathcal{B}$ be given. By the definition of a topological linear space there exists $W \in \mathcal{B}$ such that $W + W \subseteq V$. For such $W \in \mathcal{B}$ there exists $a \in A$ and $b \in B$ such that $\overline{a} - a \in W$ and $\overline{b} - b \in W$. Hence $(\overline{a} + \overline{b}) - (a + b) \in V$. Let us sum up these facts: For all $V \in \mathcal{B}$ there exists $y := a + b \in A + B$ such that $x - y \in V$, i.e., $x \in \operatorname{cl} (A + B)$. Consequently, we have $\operatorname{cl} A + \operatorname{cl} B \subseteq \operatorname{cl} (A + B)$. (vii) From (vi).

(viii) From (vi) and Proposition A.1 (v).

(ix) From (viii).

In general, (i), (iii), (iv), (vi) and (viii) do not hold with equality: (i) $A = \{x \in \mathbb{R}^2 | x_2 \ge 1/x_1, x_1 > 0\} \cup \{0, 0\}.$ (iii) $A_1 = \{0, 2\}, A_2 = \{1, 3\}.$ (iv) $A_1 = \{0\}, A_2 = \{1\}.$ (vi) and (viii). $A = \{x \in \mathbb{R}^2 | x_2 \ge 1/x_1, x_1 > 0\}, B = \{x \in \mathbb{R}^2 | x_2 = 0, x_1 \le 0\}.$

Proposition A.5 Let X, Y be arbitrary sets, let $A_i \subseteq X$, let I be an arbitrary index set and $T: X \to Y$ a function. Then,

- (i) $T(\bigcup_{i\in I} A_i) = \bigcup_{i\in I} T(A_i),$
- (ii) $T(\bigcap_{i\in I} A_i) \subseteq \bigcap_{i\in I} T(A_i).$

Proof. (i) $T(\bigcup_{i \in I} A_i) = \bigcup \{T(x) | x \in \bigcup_{i \in I} A_i\} = \bigcup_{i \in I} \bigcup_{x \in A_i} T(x) = \bigcup_{i \in I} T(A_i).$ (ii) $y \in T(\bigcap_{i \in I} A_i) \Rightarrow \forall i \in I : y \in T(A_i) \Rightarrow \bigcap_{i \in I} T(A_i).$

To see that, in general, (ii) does not hold with equality, let $T : \mathbb{R} \to \mathbb{R}$, T(x) = ||x||, $A_1 = \mathbb{R}_+$, $A_2 = \mathbb{R}_-$.

Proposition A.6 Let X, Y be linear spaces, let $A \subseteq X$ and let $T : X \to Y$ be a linear operator. Then, conv T(A) = T(conv A).

Proof. Follows from the definition of the convex hull (by convex combinations). \Box

Proposition A.7 Let X, Y be normed spaces, let $A \subseteq X$ and let $T : X \to Y$ be a linear continuous operator. Then,

- (i) $T(\operatorname{cl} A) \subseteq \operatorname{cl} T(A)$,
- (ii) $T(\operatorname{cl}\operatorname{conv} A) \subseteq \operatorname{cl}\operatorname{conv} T(A)$,
- (iii) $\operatorname{cl} T(\operatorname{cl} \operatorname{conv} A) = \operatorname{cl} \operatorname{conv} T(A)$.

Proof. (i) Let $y \in T(c|A)$. Then there is some $x \in c|A$ such that y = T(x). There exists a sequence $\{a_n\}_{n\in\mathbb{N}}\subseteq A$ with $a_n\to x$. Hence $T(a_n)\to y$, i.e., $y\in \operatorname{cl} T(A)$. (ii) Follows from (i) and Proposition A.6.

(iii) Follows from (ii) and Proposition A.6.

In order to see that, in general, (i) and (ii) do not hold with equality consider the set A = $\{x \in \mathbb{R}^2 | x_2 \ge 1/x_1, x_1 > 0\}$ and the operator $T : \mathbb{R}^2 \to \mathbb{R}, T(x) = x_1$.

Proposition A.8 Let X be a linear topological space, $A, B \subseteq X$ and $int B \neq 0$. Then, $A + \operatorname{int} B = \operatorname{cl} A + \operatorname{int} B.$

Proof. Of course, $A + \operatorname{int} B \subseteq \operatorname{cl} A + \operatorname{int} B$. To show the opposite inclusion let $y \in \operatorname{cl} A + \operatorname{int} B$. We have $y - a \in \text{int } B$ for some $a \in \text{cl } A$. For all neighborhoods N of 0 there exists some $\bar{a} \in A$ such that $-\bar{a} \in \{-a\} + N$. Since int B is nonempty and open, there exists some neighborhood N of 0 such that $y - \overline{a} \in \{y - a\} + N \subseteq \operatorname{int} B$.

Proposition A.9 Let $A, B \subseteq \mathbb{R}^p$ be nonempty closed convex sets. Then, $A \subseteq B$ implies $0^+ A \subseteq 0^+ B.$

Proof. [1, Corollary 8.3.3] yields
$$0^+A = 0^+(A \cap B) = 0^+A \cap 0^+B \subseteq 0^+B$$
.

The previous assertion is not true for nonclosed sets, in general. For instance, setting A := $\{x \in \mathbb{R}^2 | x_1 > 0, x_2 > 0\}$ and $B := A \cup \{0, 0\}$, we obtain $0^+ A = \mathbb{R}^2_+$ and $0^+ B = B$.

Proposition A.10 Let $A, B \subseteq \mathbb{R}^p$ be convex and let int $B \neq \emptyset$. Then it holds

$$A + \operatorname{int} B = \operatorname{int} (A + B).$$

Proof. Of course, $A + \operatorname{int} B \subseteq A + B$. Since $A + \operatorname{int} B$ is open, it follows $A + \operatorname{int} B \subseteq \operatorname{int} (A + B)$. Without loss of generality we can assume $A \neq \emptyset$. Hence we have int $(A + B) \neq \emptyset$. The definition of the relative interior of a convex set implies int $(A+B) = \operatorname{ri}(A+B)$, int $B = \operatorname{ri} B$ and ri $A \subseteq A$. Finally, from [1, Corollary 6.6.2] it follows int (A+B) = ri(A+B) = ri(A+ri)B = ri(A+ri)B $\operatorname{ri} A + \operatorname{int} B \subseteq A + \operatorname{int} B.$

In the previous assertion the assumptions to A and B cannot be omitted as the following examples show.

(i) $A, B \subseteq \mathbb{R}^2, A = \mathbb{B}, B = \{0\}$ (int $B = \emptyset$). Then $\emptyset = A + \operatorname{int} B \neq \operatorname{int} (A + B) = \operatorname{int} \mathbb{B}$.

(ii) $A, B \subseteq \mathbb{R}^2$, $A = \operatorname{conv} \{(0,1)^T, (0,-1)^T\}$, $B = \mathbb{B} \cup \operatorname{conv} \{(1,0)^T, (2,0)^T\}$ (*B* is not convex). The point $(0,3/2)^T$ belongs to int (A+B), but it does not belong to $A + \operatorname{int} B$.

(iii) $A, B \subseteq \mathbb{R}^2$, $A = \operatorname{conv} \left\{ (-1, 1)^T, (-1, -1)^T \right\} \cup \operatorname{conv} \left\{ (1, 1)^T, (1, -1)^T \right\}$, $B = \mathbb{B}$ (A is not convex). The point $(0, 0)^T$ belongs to int (A + B), but it does not belong to $A + \operatorname{int} B$.

B Partially ordered sets

In this section, we recall some basic ideas with respect to partially ordered sets, which can be found, for instance, in [6] or [84].

Let (Y, \leq) be a partially ordered set, i.e., Y is equipped with a reflexive, transitive and antisymmetric relation.

If V is a subset of Y and the point $y_0 \in Y$ satisfies $v \leq y_0$ for all $v \in V$, then y_0 is called an *upper bound* of V. The subset V is now said to be *bounded above*.

If $y_0 \in Y$ is an upper bound of V such that $y_0 \leq \overline{y}$ for any other upper bound $\overline{y} \in Y$ of V, then y_0 is called *least upper bound* or *supremum* of V and is denoted by $\sup V$. If V has a supremum, then it is uniquely defined. This is an easy consequence of the antisymmetry of the relation \leq .

The definitions of bounded below and lower bound are analogous. The greatest lower bound or infimum is defined analogously and is denoted by V.

A partially ordered set Y is said to be order complete (or a complete lattice) if every subset of Y has a supremum and an infimum. If Y is order complete and $V = \emptyset$, then $\sup V = \inf Y$ and $\inf V = \sup Y$. The set Y is called *Dedekind complete* if every nonempty subset of Y which is bounded above (bounded below) has a supremum (infimum). Note that for Dedekind completeness an one-sided condition is already sufficient, this means Y is Dedekind complete if and only if every nonempty subset of Y which is bounded above has a supremum [84, Theorem 1.4]. An element $\bar{y} \in Y$ is called the *largest* element of (Y, \leq) if $y \leq \bar{y}$ for all $y \in Y$. The *smallest* element is defined analogously. If (Y, \leq) has a largest (smallest) element, then it is uniquely defined.

Let (Y, \leq) , (Y^*, \leq^*) be Dedekind complete partially ordered sets with $Y^* \subseteq Y$ and such that \leq and \leq^* coincide on Y^* . We denote the supremum (infimum) of a subset $V \subseteq Y$ with respect to (Y, \leq) by $\sup V$ (inf V) and the supremum (infimum) of a subset $V \subseteq Y^*$ with respect to (Y^*, \leq^*) by $\sup^* V$ (inf * V).

Proposition B.1 If $\emptyset \neq V \subseteq Y^*$ is bounded above, then $\sup V \leq \sup^* V$. Under the additional assumption $\sup V \in Y^*$ we even have equality.

Proof. By hypothesis, the set $V \subseteq Y^*$ is bounded above (with respect to (Y^*, \leq^*)). Hence $V \subseteq Y$ is bounded above (with respect to (Y, \leq)). Since V is nonempty and (Y, \leq) and (Y^*, \leq^*)) are Dedekind complete, sup V and sup^{*} V exist. Of course, sup^{*} V $\in Y^*$ is an upper bound of V with respect to (Y, \leq) . By the definition of the supremum we get $\sup V \leq \sup^* V$. The second assertion follows from the definition.

C Ordered conlinear spaces

The concept of an *ordered conlinear space* provides the theoretical background of this work and seems to be the natural framework for convexity rather than linear spaces. A systematic study of this concept can be found in Hamel [33]. Similar concepts like semi-linear spaces and almost linear spaces are also discussed there. Here we just give some definitions.

Definition C.1 ([33]) A set Y equipped with an addition $+ : Y \times Y \to Y$ is said to be a (real) conlinear space (Y, +) if the following axioms are satisfied:

(C1) (Y, +) is a commutative monoid with neutral element θ , i.e.,

- (i) $\forall y_1, y_2, y_3 \in Y : y_1 + (y_2 + y_3) = (y_1 + y_2) + y_3,$
- (ii) $\exists \theta \in Y, \forall y \in Y : y + \theta = \theta + y = y$,
- (iii) $\forall y_1, y_2 \in Y : y_1 + y_2 = y_2 + y_1.$

(C2) There is mapping from $\mathbb{R}_+ \times Y$ into Y, assigning $\alpha \ge 0$ and $y \in Y$ the product $\alpha y := \alpha \cdot y \in Y$ such that the following conditions are satisfied:

- (i) $\forall y \in Y, \forall \alpha, \beta \ge 0 : \alpha \cdot (\beta \cdot y) = (\alpha \beta) \cdot y$,
- (ii) $\forall y \in Y : 1 \cdot y = y$,
- (iii) $\forall y \in Y : 0 \cdot y = \theta$.
- (iv) $\forall \alpha \ge 0, \forall y_1, y_2 \in Y : \alpha \cdot (y_1 + y_2) = (\alpha \cdot y_1) + (\alpha \cdot y_2).$

Definition C.2 ([33]) Let (Y, +) be a conlinear space and \leq a partial ordering on Y satisfying the following conditions:

- (O1) $(y_1, y_2, y \in Y, y_1 \le y_2) \Rightarrow y_1 + y \le y_2 + y,$
- (O2) $(y_1, y_2 \in Y, y_1 \le y_2, \alpha \ge 0) \Rightarrow \alpha y_1 \le \alpha y_2.$

Then, $(Y, +, \leq)$ is called an ordered conlinear space.

Definition C.3 ([33]) Let (Y, +) be a conlinear space. A subset $V \subseteq Y$ is said to be convex if $v_1, v_2 \in V, \lambda \in [0, 1]$ implies that $\lambda v_1 + (1 - \lambda)v_2 \in V$.

In [33], a cone in conlinear space Y is defined to be an element $y \in Y$ satisfying $\alpha y = y$ for all $\alpha > 0$. In contrast to this, we define a cone in Y as a subset of Y, because this definition is closer related to the usual definition of a cone in linear spaces.

Definition C.4 Let (Y, +) be a conlinear space. A subset $V \subseteq Y$ is said to be cone if $v \in V, \alpha > 0$ implies that $\alpha v \in V$.

Appendix

Index of notation

\mathbb{N}	positive integers
\mathbb{R}^{p}	<i>p</i> –dimensional Euclidean space
∥ ∙∥	Euclidean norm in \mathbb{R}^p
\mathbb{R}^p_+	non–negative orthant of \mathbb{R}^p
\mathbb{R}_+	equals \mathbb{R}^1_+
[a,b]	closed interval in \mathbb{R}
(a,b)	open interval in \mathbb{R}
$\operatorname{cl} A$	closure of the set A
$\operatorname{int} A$	(topological) interior of the set A
$\operatorname{ri} A$	relative interior of the set A
$\operatorname{rb} A$	relative boundary of the set A
$\operatorname{conv} A$	convex hull of the set A
$\ln A \; (\text{aff } A)$	linear (affine) hull of the set A
K°	polar cone of a nonempty cone $K \subseteq \mathbb{R}^p$
L^{\perp}	orthogonal space of a linear subspace $L \subseteq \mathbb{R}^p$ $(L^{\perp} = L^{\circ})$
0^+A	recession cone of a nonempty convex set $A \subseteq \mathbb{R}^p$
$\hat{\mathcal{F}}\left(\mathcal{F} ight)$	set of (nonempty) closed subsets of \mathbb{R}^p
$\hat{\mathcal{C}}~(\mathcal{C})$	set of (nonempty) closed convex subsets of \mathbb{R}^p
\mathcal{C}_K	set of all $A \in \mathcal{C}$ with $0^+A = K$
$\hat{\mathcal{C}}_K$	$\mathcal{C}_K \cup \{ \emptyset \}$
$\hat{\mathcal{C}}^{\star}, \mathcal{C}^{\star}, \hat{\mathcal{C}}^{\star}_{K}, \mathcal{C}^{\star}_{K}$	$\hat{\mathcal{C}}, \mathcal{C}, \hat{\mathcal{C}}_K, \mathcal{C}_K$, but with supremum oriented members
$\hat{\mathcal{C}}^\diamond, \mathcal{C}^\diamond, \hat{\mathcal{C}}^\diamond_K, \mathcal{C}^\diamond_K$	$\hat{\mathcal{C}}, \mathcal{C}, \hat{\mathcal{C}}_K, \mathcal{C}_K$, but with infimum oriented members
Γ_K	set of positively homogeneous real–valued functions defined on ri K°
Γ_K^{\star}	set of convave members of Γ_K
Γ_K^\diamond	set of convex members of Γ_K
$\hat{\Gamma}_{K}^{\star}$	Γ_K^{\star} extended by a largest element $+\infty_K$
$\hat{\Gamma}_{K}^{\diamond}$	Γ_K^\diamond extended by a smallest element $-\infty_K$
$\mathcal{N}(x)$	set of all neighborhoods of x
$\mathcal{N}^{\#}_{\infty}$	set of infinite subsets of \mathbb{N}
\mathcal{N}_{∞}	set of cofinal subsets of \mathbb{N}
inf (INF)	(alternative concept of) infimum
$\sup(SUP)$	(alternative concept of) supremum
$\liminf (\text{LIM INF})$	(alternative concept of) lower limit
$\limsup (\text{LIM SUP})$	(alternative concept of) upper limit
. , , , , , , , , , , , , , , , , , , ,	

lim (LIM)	(alternative concept of) limit
$\operatorname{Inf}[A, C]$	infimal set of A (with respect to the ordering cone C)
$\operatorname{Sup}[A, C]$	supremal set of A (with respect to the ordering cone C)
$\operatorname{Min}[A, C]$	set of minimal $(=$ minimal efficient) elements of A
$\operatorname{Max}[A, C]$	set of maximal $(=$ maximal efficient) elements of A
$\operatorname{wMin}[A, C]$	set of weakly minimal (= minimal weakly efficient) elements of A
$\operatorname{wMax}[A, C]$	set of weakly maximal (= maximal weakly efficient) elements of A
$\mathrm{Inf}A\;(\mathrm{Sup}A)$	= Inf[A, C] (= Sup[A, C])
$\operatorname{Min} A (\operatorname{Max} A)$	$= \operatorname{Min}[A, C] \ (= \operatorname{Max}[A, C])$
$\operatorname{wMin} A (\operatorname{wMax} A)$	$= \operatorname{wMin}[A, C] (= \operatorname{wMax}[A, C])$
\oplus	closed Minkowski addition
\blacksquare	(Minkowski addition with) change of orientation
	equals " \square -"
$\operatorname{epi} f(\operatorname{hyp} f)$	epigraph (hypograph) of a function f
$\operatorname{gr} f$	graph of a map f
$\operatorname{dom} f$	(effective) domain of a map f
$\operatorname{rg} f$	range of a map f
$\operatorname{isc} f$	inner semi–continuous hull of f
$\operatorname{osc} f$	outer semi–continuous hull of f
$\operatorname{lsc} f$	lower semi–continuous hull of f
$\operatorname{usc} f$	upper semi–continuous hull of f
$\delta(\cdot A)$	(convex) indicator function of $A \subseteq \mathbb{R}^p$
$\delta^*(\cdot A)$	support function of $A \subseteq \mathbb{R}^p$
$\Delta(\cdot A)$	set–valued indicator function of $A \subseteq \mathbb{R}^p$
$\Delta_c^*(\cdot A)$	set–valued support function of $A \subseteq \mathbb{R}^p$ with respect to $c \in \mathbb{R}^p$
$f^*(\cdot)$	the conjugate of a real-valued function f
$f_c^*(\cdot)$	the conjugate of a $\hat{\mathcal{C}}$ -valued function f

Bibliography

- [1] Rockafellar, R. T.: Convex Analysis, Princeton University Press, Princeton N. J., 1970
- [2] Aubin, J.-P.; Frankowska, H.: Set-valued Analysis, Systems and Control: Foundations and Applications, 2. Boston etc.: Birkhäuser, 1990
- [3] Baier, R.; Farkhi, E. M.: Differences of convex compact sets in the space of directed sets.
 I: The space of directed sets, Set-Valued Anal. 9, No. 3, (2001), 217–245
- [4] Beer, G.: Topologies on Closed and Closed Convex Sets, Mathematics and its Applications, 268, Kluwer Academic Publishers, Dordrecht, 1993
- [5] Berge, C.: Espaces topologiques Fonctions multivoques, Dunod, Paris, 1959
- [6] Birkhoff, G.: Lattice Theory, (3rd ed.) Am. Math. Soc. Coll. Publ. 25. Providence, 1976
- [7] Borwein, J.: Multivalued convexity and optimization: A unified approach to inequality and equality constraints, *Math. Program.* **13**, (1977), 183–199
- [8] Borwein, J. M.: A Lagrange multiplier theorem and a sandwich theorem for convex relations, *Math. Scand.* 48, (1981), 189–204
- [9] Borwein, J. M.: Adjoint process duality, Math. Oper. Res. 8, (1983), 403–434
- [10] Borwein, J.; Lewis, A.: Convex Analysis and Nonlinear Optimization, Springer, New York, 2000
- Boţ, R. I.; Wanka, G.: An analysis of some dual problems in multiobjective optimization.
 I, II, Optimization 53, No.3, (2004), 281–300, 301–324
- [12] Breckner, W. W.: Dualität bei Optimierungsaufgaben in halbgeordneten topologischen Vektorräumen. I., Math., Rev. Anal. Numér. Théor. Approximation 1, (1972), 5–35
- [13] Brink, C.: Power structures, Algebra Universalis 30, (1993), 177–216
- [14] Brumelle, S.: Duality for multiple objective convex programs, Math. Oper. Res. 6, (1981), 159–172
- [15] Brunn, H.: Über Curven ohne Wendepunkte, München, 1889

- [16] Chen, G. Y.; Jahn, J.: Optimality conditions for set-valued optimization problems, Math. Methods Oper. Res. 48, No.2, (1998), 187–200
- [17] Choquet, G.: Outils topologiques et métriques de l'analyse mathématique, Centre de documentation universitaire et SEDES, Paris, 1969
- [18] Clarke, F. H.: Generalized gradients and applications, Trans. Am. Math. Soc. 205, (1975), 247–262
- [19] Corley, H. W.: Duality theory for maximizations with respect to cones, J. Math. Anal. Appl. 84, (1981), 560–568
- [20] Corley, H. W.: Existence and Lagrangian duality for maximization of set-valued functions, J. Optimization Theory Appl. 54, (1987), 489–501
- [21] Corley, H. W.: Optimality conditions for maximizations of set-valued functions, J. Optimization Theory Appl. 58, No.1, (1988), 1–10
- [22] Crespi, G. P.; Ginchev, I.; Rocca, M.: First-order optimality conditions in set-valued optimization, *Math. Methods Oper. Res.*, accepted for publication
- [23] Deutsch, F.; Singer, I.: On single-valuedness of convex set-valued maps, Set-Valued Anal. 1, (1993), 97–103
- [24] Gale, D.; Kuhn, H. W.; Tucker, A. W.: Linear programming and the theory of games, Activity Analysis of Production and Allocation, Chap. XIX, (1951), 317–329
- [25] Gerstewitz (Tammer), Chr.: Lagrange- und Fencheldualität in der Vektoroptimierung, Wiss. Z. Tech. Hochsch. Ilmenau 28, No.5, (1982), 61–75
- [26] Getán, J.; Martínez-Legaz, J.-E.; Singer, I.: (*, s)-dualities, J. Math. Sci. 115, No.4, (2003), 2506–2541
- [27] Godini, G.: A framework for best simultaneous approximation: Normed almost linear spaces, J. Approximation Theory 43, (1985), 338–358
- [28] Göpfert, A.; Gerth (Tammer), Chr.: Über die Skalarisierung und Dualisierung von Vektoroptimierungsproblemen, Z. Anal. Anwend. 5, (1986), 377–384
- [29] Göpfert, A.: Mathematische Optimierung in allgemeinen Vektorräumen, Mathematisch-Naturwissenschaftliche Bibliothek, Band 58, BSB B. G. Teubner Verlagsgesellschaft, Leipzig, 1973
- [30] Göpfert, A.; Nehse, R.: Vektoroptimierung. Theorie, Verfahren und Anwendungen, Teubner, Leipzig, 1990
- [31] Göpfert, A.; Riahi, H.; Tammer, Chr.; Zălinescu, C.: Variational Methods in Partially Ordered Spaces, CMS Books in Mathematics 17. Springer-Verlag, 2003

- [32] Götz, A.; Jahn, J.: The Lagrange multiplier rule in set-valued optimization, SIAM J. Optim. 10, No.2, (1999), 331–344
- [33] Hamel, A.: Variational Principles on Metric and Uniform Spaces, Habilitation thesis, Martin-Luther-Universität Halle-Wittenberg, 2005
- [34] Hamel, A.; Heyde, F.; Löhne, A.; Tammer, Chr.; Winkler, K.: Closing the duality gap in linear vector optimization, *Journal of Convex Analysis* 11, No. 1, (2004), 163–178
- [35] Hamel, A.; Löhne, A.: Generalized minimal element theorems and Ekeland's principle for set relations, J. Nonl. Conv. Anal., accepted for publication
- [36] Hiriart-Urruty, J.-B.; Lemaréchal, C.: Convex Analysis and Minimization Algorithms. Part 1: Fundamentals, Grundlehren der Mathematischen Wissenschaften, 305: Springer-Verlag, Berlin, 1993
- [37] Holmes, R. B.: Geometric Functional Analysis and its Applications, Graduate Texts in Mathematics 24, Springer, New York, 1975
- [38] Ioffe, A. D.: On foundations of convex analysis, Collective phenomena, 3rd int. Conf., Moscow 1978, Ann. New York Acad. Sci. 337, (1980), 103–117
- [39] Jahn, J.: Duality in vector optimization, Math. Program. 25, (1983), 343–353
- [40] Jahn, J.: Mathematical Vector Optimization in Partially Ordered Linear Spaces, Verlag Peter Lang, Frankfurt am Main–Bern–New York, 1986
- [41] Jahn, J.: Grundlagen der Mengenoptimierung, Habenicht, W.; Scheubrein, B.; Scheubrein, R. (eds.): Multi-Criteria- und Fuzzy-Systeme in Theorie und Praxis, Deutscher Universitäts-Verlag, 2003
- [42] Jahn, J.: Vector Optimization. Theory, Applications, and Extensions, Springer-Verlag, Berlin, 2004
- [43] Jahn, J.; Rauh, R.: Contingent epiderivatives and set-valued optimization, Math. Methods Oper. Res. 46, No.2, (1997), 193–211
- [44] Kawasaki, H.: A duality theorem in multiobjective nonlinear programming, Math. Oper. Res. 7, (1982), 95–110
- [45] Kelley, J. L.: *General Topology*, Springer, New York, 1955
- [46] Kelley, J. L.; Namioka, I.: Linear Topological Spaces, Springer, Berlin, 1963
- [47] Kuratowski, C.: Topologie. Vol. I., Panstwowe Wydawnictwo Naukowe, Warszawa, 1958
- [48] Kuratowski, K.; Mostowski, A.: Set Theory, Polish Scientific Publishers, Warszawa, 1968
- [49] Kuroiwa, D.: Lagrange duality of set-valued optimization with natural criteria, RIMS Kokyuroku 1068, (1998), 164–170

- [50] Kuroiwa, D.: The natural criteria in set-valued optimization, RIMS Kokyuroku 1031, (1998), 85–90
- [51] Kuroiwa, D.: Some duality theorems of set-valued optimization, *RIMS Kokyuroku* 1079, (1999), 15–19
- [52] Kuroiwa, D.: Some duality theorems of set-valued optimization with natural criteria, in: Takahashi, Wataru (ed.) et al., *Nonlinear analysis and convex analysis*, Proceedings of the 1st international conference (NACA98), Niigata, Japan, July 28–31, 1998. Singapore: World Scientific. 221–228, 1999
- [53] Kuroiwa, D.: On set-valued optimization, Nonlinear Analysis 47, (2001), 1395–1400
- [54] Löhne, A.: On convex functions with values in conlinear spaces, *Journal of Nonlinear* and Convex Analysis, accepted for publication
- [55] Löhne, A.: On conjugate duality in optimization with set relations, in: Geldermann, J.; Treitz, M. (eds.): Entscheidungstheorie und -praxis in industrieller Produktion und Umweltforschung, Shaker, Aachen, 2004
- [56] Löhne, A.: Optimization with set relations: Conjugate Duality, *Optimization*, accepted for publication
- [57] Luc, D. T.: Theory of Vector Optimization, Lecture Notes in Economics and Mathematical Sciences, 319, Springer–Verlag, Berlin, 1988
- [58] Luc, D. T.: Contingent derivatives of set-valued maps and applications to vector optimization, Math. Program., Ser. A 50, No.1, (1991), 99–111
- [59] Luc, D. T.; Jahn, J.: Axiomatic approach to duality in optimization, Numer. Funct. Anal. Optimization 13, No.3–4, (1992), 305–326
- [60] Marti, J. T.: Konvexe Analysis, Birkhäuser Verlag, Basel and Stuttgart, 1977
- [61] Martínez-Legaz, J.-E.; Rubinov, A. M.; Singer, I.: Downward sets and their separation and approximation properties, J. Glob. Optim. 23, No.2, (2002), 111–137
- [62] Martínez-Legaz J.-E.; Singer, I.: *-dualities, Optimization 30, No.4, (1994), 295–315
- [63] Nieuwenhuis, J. W.: Supremal points and generalized duality, Math. Operationsforsch. Stat., Ser. Optimization 11, (1980), 41–59
- [64] Nishnianidze, Z. G.: Fixed points of monotonic multiple-valued operators (in Russian), Bull. Georgian Acad. 114, (1984), 489–491
- [65] Ohlendorf, E.; Tammer, Chr.: Multicriteria fractional programming an approach by means of conjugate functions. OR Spektrum 16, No.4, (1994), 249–254
- [66] Postolică, V.: Vectorial optimization programs with multifunctions and duality, Ann. Sci. Math. Québek 10, (1986), 85–102

- [67] Radström, H.: An embedding theorem for spaces of convex sets, Proc. Am. Math. Soc. 3, (1952), 165–169
- [68] Rockafellar, R. T.; Wets, R. J.-B.: Variational Analysis, Springer, Berlin, 1998
- [69] Salinetti, G.; Wets, R. J.-B.: On the relations between two types of convergence for convex functions, J. Math. Anal. Appl. 60, (1977), 211–226
- [70] Salinetti, G.; Wets, R. J.-B.: On the convergence of sequences of convex sets in finite dimensions, SIAM Rev. 21, (1979), 18–33
- [71] Sawaragi, Y.; Nakayama, H.; Tanino, T.: Theory of multiobjective optimization, Mathematics in Science and Engineering, Vol. 176. Academic Press Inc., Orlando, 1985
- [72] van Slyke, R. M.; Wets, R. J.-B.: A duality theory for abstract mathematical programs with applications to optimal control theory, J. Math. Anal. Appl. 22, (1968), 679–706
- [73] Song, W.: A generalization of Fenchel duality in set-valued vector optimization, Math. Methods Oper. Res., 48, (1998), 259–272
- [74] Sonntag, Y.; Zălinescu, C.: Scalar convergence of convex sets, J. Math. Anal. Appl. 164, No.1, (1992), 219–241
- [75] Tammer, Chr.: Lagrange–Dualität in der Vektoroptimierung. Wiss. Z. Tech. Hochsch. Ilmenau 37, No.3, (1991), 71–88
- [76] Tammer, Chr.; Tammer K.: Duality results for convex vector optimization problems with linear restrictions, in: Kall, Peter (ed.): System modeling and optimization, Proceedings of the 15th IFIP conference, Zurich, Switzerland, September 2–6, 1991, Lect. Notes Control Inf. Sci. 180, 55–64, Springer, Berlin, 1992
- [77] Tanino, T.: On the supremum of a set in a multi-dimensional space, J. Math. Anal. Appl. 130, (1988), 386–397
- [78] Tanino, T.: Conjugate duality in vector optimization, J. Math. Anal. Appl. 167, No.1, (1992), 84–97
- [79] Tanino, T.; Sawaragi, Y.: Duality theory in multiobjective programming, J. Optimization Theory Appl. 27, (1979), 509–529
- [80] Truong, X. D. H.: Ekeland's variational principle for set-valued maps studied with the set optimization approach, Preprint No. 289, Institute of Applied Mathematics, University of Erlangen–Nürnberg, (2001)
- [81] Tuy, H.: Convex analysis and global optimization, Kluwer Academic Publishers, Dordrecht, 1998
- [82] Wijsman, R.: Convergence of sequences of convex sets, cones and functions, II, Trans. Amer. Math. Soc. 123, (1966), 32–45

- [83] Young, R. C.: The algebra of many-valued quantities, Math. Ann. 104, (1931), 936-946
- [84] Zaanen, A. C.: Introduction to Operator Theory in Riesz Spaces, Springer-Verlag, Berlin, 1997
- [85] Zălinescu, C.: Convex Analysis in General Vector Spaces, World Scientific, Singapore, 2002
- [86] Zowe, J.: A duality theorem for a convex programming problem in order complete vector lattices, J. Math. Anal. Appl. 50, (1975), 273–287